INDEPENDENCE AND COUNTING PROBLEMS IN COMBINATORICS AND NUMBER THEORY

by

ROBERT ARTHUR HANCOCK

A thesis submitted to The University of Birmingham for the degree of DOCTOR OF PHILOSOPHY

> School of Mathematics The University of Birmingham January 2019

UNIVERSITY^{OF} BIRMINGHAM

University of Birmingham Research Archive

e-theses repository

This unpublished thesis/dissertation is copyright of the author and/or third parties. The intellectual property rights of the author or third parties in respect of this work are as defined by The Copyright Designs and Patents Act 1988 or as modified by any successor legislation.

Any use made of information contained in this thesis/dissertation must be in accordance with that legislation and must be properly acknowledged. Further distribution or reproduction in any format is prohibited without the permission of the copyright holder.

Abstract

The method of hypergraph containers has become a very important tool for dealing with problems which can be phrased in the language of independent sets in hypergraphs. This method has applications to numerous problems in combinatorics and other areas. In this thesis we consider examples of such problems; in particular problems concerning sets avoiding solutions to a given system of linear equations \mathcal{L} (known as \mathcal{L} -free sets) or graphs avoiding copies of a given graph H (H-free graphs).

First we attack a number of questions relating to \mathcal{L} -free sets. For example, we give various bounds on the number of maximal \mathcal{L} -free subsets of [n] for three-variable homogeneous linear equations \mathcal{L} .

We then use containers to prove results corresponding to problems concerning tuples of disjoint independent sets in hypergraphs. In particular we generalise the random Ramsey theorem of Rödl and Ruciński by providing a resilience analogue. We obtain similar results for \mathcal{L} -free sets.

Finally we consider the Maker-Breaker game where Maker's aim is to obtain a solution to a given system of linear equations \mathcal{L} amongst a random set of integers. We determine the threshold probability for this game for a large class of systems \mathcal{L} .

Acknowledgements

First I would like to thank Andrew Treglown who has been a fantastic supervisor; I am indebted to your collaboration, support and guidance, which not only has been invaluable throughout my PhD, but which I recognise already will serve me very well for my future pursuits. I'd also like to thank those who have collaborated with me during my PhD, Nicholas Day, Victor Falgas-Ravry and Katherine Staden. I'm especially grateful to Victor for hosting my visit to Umeå during the last semester of my PhD.

Thanks also to the vibrant combinatorics groups both at Birmingham and Umeå for many interesting mathematical discussions and seminars. I also greatly appreciate all the maths teachers who have influenced me throughout my life, from those at Wallace High School, to those who have lectured me at St Andrews and Birmingham. From St Andrews I am particularly indebted to James Mitchell and Collin Bleak, who first introduced me to maths research.

Thanks to Janette Lowe for helping with the administrative requirements of my PhD, and thanks to EPSRC for the funding of this PhD.

I'd like to thank my fellow maths PhD students at Birmingham for their great company, especially those from office 315. Thanks to all of the footballers in Birmingham from maths and beyond for many entertaining games. Finally thanks to my family for their support, in particular my partner Ceridwen for six years of cheering me up, proofreading and supporting me no matter what.

CONTENTS

1	Intr	Introduction					
	1.1	Indepe	endent sets in hypergraphs	1			
		1.1.1	Containers	3			
		1.1.2	Removal lemmas	4			
	1.2 Our work						
		1.2.1	Solution-free sets of integers	5			
		1.2.2	Ramsey properties of graphs and the integers	6			
		1.2.3	The Maker-Breaker Rado game on a random set of integers	7			
	1.3	on and preliminaries	8				
		1.3.1	Solutions to systems of linear equations	8			
		1.3.2	Notation	11			
		1.3.3	Probabilistic tools	12			
2	Con	ontainer and removal lemmas					
3	Solution-free sets of integers						
-	3.1	Introd	uction	16			
		3.1.1	The size of the largest solution-free set	17			
		3.1.2	The number of solution-free sets	18			
		3.1.3	The number of maximal solution-free sets	19			
	3.2	Contai	iners and independent sets in graphs	22			
		3.2.1	Container and removal lemmas	22			
		3.2.2	Independent sets in graphs	22			
		3.2.3	Link graphs and maximal independent sets	23			
	3.3	The si	ze of the largest solution-free set	26			
	3.4	The number of solution-free sets		35			
	3.5	5 The number of maximal solution-free sets		39			
		3.5.1	A general upper bound	39			
		3.5.2	Upper bounds for $px + qy = z$	42			
		3.5.3	Lower bounds for $px + qy = z$	45			
	3.6	Conclu	uding remarks	48			

4	More on solution-free sets of integers 49							
	4.1	Introd	luction	. 49				
		4.1.1	The size of the largest solution-free set	. 50				
		4.1.2	The number of solution-free sets	. 52				
		4.1.3	The number of maximal solution-free sets	. 53				
	4.2	Link ł	hypergraphs and the main lemmas	. 54				
		4.2.1	Link hypergraphs	. 54				
		4.2.2	The main lemmas	. 55				
	4.3	The s	ize of the largest solution-free set	. 59				
	4.4	The number of solution-free sets						
	4.5	The number of maximal solution-free sets						
	4.6	Best upper bounds on $f_{\max}(n, \mathcal{L}^w)$						
	4.7	Concl	uding Remarks	. 75				
5	Ramsey properties of graphs and the integers 76							
	5.1	Introd	luction	. 76				
		5.1.1	Resilience in hypergraphs and the integers	. 78				
		5.1.2	Enumeration questions for Ramsey problems	. 91				
	5.2	2 Container results for disjoint independent sets						
	5.3	cations of the container method to r -tuples of solution-free sets \ldots	. 96					
		5.3.1	Matrices which satisfy $(*)$. 98				
		5.3.2	Useful matrix lemmas	. 99				
		5.3.3	A container theorem for tuples of \mathcal{L}^d -free sets $\ldots \ldots \ldots \ldots$. 102				
		5.3.4	The number of $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -free subsets of $[n]$. 107				
		5.3.5	The resilience of being $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -Rado	. 108				
		5.3.6	The size of the largest (\mathcal{L}^s, r) -free set $\ldots \ldots \ldots \ldots \ldots$. 110				
		5.3.7	Two-coloured analogue of the Cameron–Erdős conjecture	. 113				
	5.4	Applie	cations of the container method to graph Ramsey theory	. 113				
		5.4.1	Definitions and notation	. 114				
		5.4.2	The maximum density of a hypergraph which is not Ramsey	. 115				
		5.4.3	The special case of graphs: maximum size and typical structure .	. 119				
		5.4.4	A container theorem for Ramsey hypergraphs	. 123				
		5.4.5	The number of hypergraphs which are not Ramsey	. 128				
		5.4.6	The resilience of being (H_1, \ldots, H_r) -Ramsey	. 130				
6	The	The Maker-Breaker Rado game on a random set of integers 1						
	6.1	Introduction						
	6.2	Proof of Maker's win in Theorem 6.3						
	6.3	Proof of Breaker's win in Theorem 6.3						
	6.4	3.4 Concluding remarks and Proof of Theorem 6.2						
		6.4.1	Improvements on Breaker's strategy	. 177				
		6.4.2	Matrices which do not satisfy $(*)$. 178				
		6.4.3	Proof of Theorem 6.2	. 180				

List of References

182

Chapter 1 Introduction

1.1 Independent sets in hypergraphs

Let \mathcal{X} be an algebraic object, such as a group, a field, or a subset of one of these such as the set of positive integers $[n] := \{1, \ldots, n\}$. Many famous questions in additive combinatorics concern the combinatorial properties of a subset $X \subseteq \mathcal{X}$ which satisfy (or fail to satisfy) some given algebraic structure. Two key examples of this are finding the size of an extremal set, and counting the number of such sets.

Consider the example of sum-free sets, that is, a set which contains no solutions to x + y = z. It is easy to see that the largest sum-free set $X \subseteq [n]$ has size $\lceil n/2 \rceil$. (One example of an extremal set here is the set of odd numbers, which is sum-free since two odd numbers can never sum to another odd number.) Indeed, a classical question is to ask about the size of an extremal set avoiding solutions in [n] to a given system of linear equations; the most famous of these concern sets not containing arithmetic progressions. Note arithmetic progressions can be encoded as solutions to a system of linear equations.

Similarly to above, two famous questions in extremal graph theory concern finding the size of an extremal graph, or counting the number of graphs, which have (or do not have) a given property \mathcal{P} . For example, Mantel's theorem [80] says that all triangle-free

graphs on n vertices have at most $\lfloor n^2/4 \rfloor$ edges, where the extremal example is a complete bipartite graph with two vertex classes of equal size. Generally, the size of an extremal H-free graph is given asymptotically by the Erdős-Stone-Simonovits theorem [37], while a result of Erdős, Frankl and Rödl [34] determines asymptotically the number of H-free graphs on n vertices.

Many combinatorial problems, such as those described so far, can be phrased in terms of independent sets in hypergraphs. Consider the examples already considered; let \mathcal{H} be a hypergraph with vertex set $V(\mathcal{H}) := [n]$ and a hyperedge for each solution to x + y = zin [n] (note that there are edges of size two and three). Then the independent sets in \mathcal{H} (subsets of $V(\mathcal{H})$ which do not contain any edge of \mathcal{H}) correspond to the sum-free subsets of [n]. Alternatively, let \mathcal{H} be a hypergraph with vertex set $V(\mathcal{H}) := E(K_n)$ (the set of all edges of the complete graph K_n) and a hyperedge for each triple of edges in K_n giving rise to a triangle. Here, the independent sets in \mathcal{H} correspond to triangle-free subgraphs of K_n .

The two key questions mentioned in each topic can now be formulated as follows:

- (i) What is the size $\alpha(\mathcal{H})$ of the largest independent set in \mathcal{H} ?
- (ii) What is the total number $i(\mathcal{H})$ of independent sets in \mathcal{H} ?

Observe that since any subset of an independent set is still independent, there are at least $2^{\alpha(\mathcal{H})}$ independent sets in \mathcal{H} . A trivial upper bound can be found by counting all of the subsets of $V(\mathcal{H})$ of size at most $\alpha(\mathcal{H})$. For many hypergraphs \mathcal{H} , it turns out that the lower bound is closer to the truth, and so most efforts have gone into trying to improve the upper bound. For example, Alon [2] proved a conjecture of Granville (see [2]) from 1988 that when \mathcal{H} is an *n*-vertex, *d*-regular graph, then $i(\mathcal{H}) \leq 2^{(1+O(d^{-0.1}))n/2}$ (note that here $\alpha(\mathcal{H}) \leq n/2$).

1.1.1 Containers

The method of containers, which was first used by Kleitman and Winston [68, 69] and Sapozhenko [103, 104, 105], has become a very good tool for establishing better upper bounds for $i(\mathcal{H})$. In general a 'container theorem' roughly states the following: There exists a family $\mathcal{C}(\mathcal{H})$ of subsets of $V(\mathcal{H})$ known as 'containers' such that

- (i) Each container is 'almost' an independent set;
- (ii) Every independent set in \mathcal{H} is a subset of a container $C \in \mathcal{C}(\mathcal{H})$;
- (iii) There are 'not too many' containers, i.e. $|\mathcal{C}(\mathcal{H})|$ is small;
- (iv) No container is 'large'.

So one may choose a container $C \in C(\mathcal{H})$ and bound the number of independent sets which it contains (in general this number should not be too big since no container is large). Then since every independent set lies in some container, we obtain an upper bound for the total number of independent sets by multiplying by the number of containers. Often this approach gives good upper bounds on $i(\mathcal{H})$, in particular as in many applications of the method there are a sufficiently small number of containers. The container method is often useful in the probabilistic setting for bounding the size of the largest independent set.

The original method used by Kleitman and Winston involved the use of an algorithm, in which a container method for graphs was implicitly stated within. They used it to obtain upper bounds on the number of lattices, and graphs without cycles of length four. In 2004 Green [48] developed a method of containers using Fourier analysis, for counting the number of subsets of [n] avoiding solutions to a given linear equation. Recently Balogh, Morris and Samotij [8], and independently Saxton and Thomason [106] developed very general container theorems for independent sets in hypergraphs; both of which have seen numerous applications to a wide range of problems. These include problems in Ramsey theory, combinatorial number theory, positional games and list colourings of graphs. See the very recent survey article by Balogh, Morris and Samotij [9] for a detailed history and a number of illustrative examples of the method of hypergraph containers. Throughout this thesis we will apply the container method to a range of problems.

1.1.2 Removal lemmas

An important tool which often comes side by side with a container theorem is a so-called 'removal lemma'. This is often used in the step of the container method mentioned above, where one must bound the number of independent sets in a given container. The graph removal lemma states that for any graphs G, H on n, h vertices respectively, and any $\varepsilon > 0$, there exists a $\delta = \delta(\varepsilon, H) > 0$ such that if G contains at most δn^h copies of H, then it may be made H-free by removing at most εn^2 edges. By considering the hypergraph \mathcal{H} with vertex set E(G) and edge set corresponding to copies of H in G, we observe that an independent set in \mathcal{H} is an H-free subgraph of G, and G itself is 'almost' an independent set.

The triangle removal lemma (i.e. H is a triangle) was first used by Ruzsa and Szemerédi [102] in 1976. The general graph removal lemma was first explicitly stated by Alon, Duke, Lefmann, Rödl and Yuster [3] and by Füredi [43] in 1994. As with the container method, there has been much work on extending the graph removal lemma to a result for hypergraphs, most notably by Gowers [44, 45] and independently by Nagle, Rödl, Schacht and Skokan [84, 97]. Green also proved a removal lemma for abelian groups using Fourier analysis [48]. This combined with Green's container lemma gives an immediate result for the number of subsets of [n] avoiding solutions to a given linear equation. This removal lemma was also generalised to systems of linear equations by Král', Serra and Vena [74].

1.2 Our work

In Chapter 2, we state the aforementioned container and removal lemmas of Green, the general container result of Balogh, Morris and Samotij, and the general removal lemma of Král', Serra and Vena. All of these results will explicitly be required for proofs of results within Chapters 3–6. We now describe the content of these four chapters.

1.2.1 Solution-free sets of integers

Given a linear equation \mathcal{L} of the form $a_1x_1 + \cdots + a_kx_k = b$, we call a set $X \subseteq [n]$ weakly \mathcal{L} -free if it does not contain any 'non-trivial' solutions to \mathcal{L} in [n] (see the notation section below for a full definition). In Chapters 3 and 4 we prove a number of results concerning weakly \mathcal{L} -free subsets of [n] where \mathcal{L} is a homogeneous linear equation in three variables. In particular, our work is motivated by the following general questions.

- (i) What is the size of the largest weakly \mathcal{L} -free subset of [n]?
- (ii) How many weakly \mathcal{L} -free subsets of [n] are there?
- (iii) How many maximal weakly \mathcal{L} -free subsets of [n] are there?

A weakly \mathcal{L} -free set $X \subseteq [n]$ is maximal if it is not possible to add any other element of [n] to X so that X remains weakly \mathcal{L} -free. Notice that (iii) (as we showed with (i) and (ii) earlier) can also be phrased in terms of independent sets in hypergraphs; if \mathcal{H} has vertex set [n] and the edges of \mathcal{H} correspond to solutions to \mathcal{L} , then the corresponding question is to count the number of maximal independent sets in \mathcal{H} .

We make progress on all three of these questions. For each question we use tools from graph theory; for (i) and (ii) our methods are somewhat elementary. For (iii) our method is more involved and utilises container and removal lemmas of Green [48]. In Chapter 3 we focus mainly on equations of the form px + qy = z for some fixed positive integers p, q with $p \ge 2$. In Chapter 4 we focus on equations of the form px + qy = rz for some fixed positive integers $p \ge q \ge r$, and also obtain some generalisations of our results to equations with more than three variables. These two chapters are based respectively on the content of two papers with Andrew Treglown [56, 57].

1.2.2 Ramsey properties of graphs and the integers

Rather than forbidding a solution to some system of linear equations, one may instead choose to assign colours to numbers in [n], and then see if there is any guarantee of there being a monochromatic solution. A famous result of Schur [110] states that for any $r \in \mathbb{N}$, if n is sufficiently large, then however [n] is r-coloured, we can find a monochromatic triple $\{x, y, z\}$ with $x, y, z \in [n]$ such that x + y = z. Van der Waerden obtained a similar result [120] for arithmetic progressions. This was followed by the generalisation by Rado [88] which describes all systems of linear equations \mathcal{L} for which, for any $r \in \mathbb{N}$, if n is sufficiently large, then however [n] is r-coloured, we can find a monochromatic solution to \mathcal{L} in [n]. In the graph setting, Ramsey's theorem [89] states that for any $r \in \mathbb{N}$ and graph H, if n is sufficiently large, then however we r-colour the edges of K_n , there is a monochromatic copy of H.

Recently there has been a trend of obtaining extremal results in sparse random sets or graphs. For example, one may ask about Rado-type properties of $[n]_p$ (each element of [n] is included with probability p independently of all other elements). Given a fixed system of linear equations \mathcal{L} which satisfies Rado's partition theorem above, and integer $r \ge 2$, how small must we make p so that with high probability (with probability one as ntends to infinity; we will use the abbreviation w.h.p.) $[n]_p$ no longer has the property that however r-coloured, there is always a monochromatic solution to \mathcal{L} ? Results of Friedgut, Rödl and Schacht [42] and Rödl and Ruciński [93] together determine a *threshold* p_0 which depends only on \mathcal{L} . Similarly, a famous result of Rödl and Ruciński [90, 91, 92] determines the threshold p_0 for a given graph H, for when the Erdős–Rényi random graph $G_{n,p}$ has the property that however its edges are r-coloured, there is always a monochromatic copy of H. (Recall that $G_{n,p}$ has vertex set [n] in which each possible edge is present with probability p, independently of all other edges.)

Consider the hypergraphs with edge sets corresponding to solutions to a given system of linear equations or copies of a given graph H described earlier. Here a tuple of disjoint independent sets describes a tuple of disjoint solution-free sets/disjoint H-free graphs. In Chapter 5, we use the container method to prove results that correspond to problems concerning tuples of disjoint independent sets in hypergraphs. That is, we are able to tackle problems relating to the Rado property of sets of integers, and Ramsey properties of graphs and hypergraphs. In particular, we strengthen the random Rado theorem of Friedgut, Rödl and Schacht [42] described above by proving a so-called resilience version of the result. We also generalise the random Ramsey theorem of Rödl and Ruciński described above by providing a resilience analogue. This result also implies the *random version of Turán's theorem* due to Conlon and Gowers [28] and Schacht [108]. We also obtain hypergraph and asymmetric generalisations and counting results. This chapter is based on joint work with Katherine Staden and Andrew Treglown [55].

1.2.3 The Maker-Breaker Rado game on a random set of integers

Recently there has been a number of results which link extremal results in sparse random graphs or sets to results in Maker-Breaker games (see, for example [75, 86]). This is due in part to it being possible to phrase both problems in the language of independent sets in hypergraphs.

Given a system of linear equations Ax = b, the Maker-Breaker (A, b)-game on a set of integers X is the game where Maker and Breaker take turns claiming previously unclaimed

integers from X, and Maker's aim is to obtain a solution to Ax = b, whereas Breaker's aim is to prevent this. We can view the game as players taking turns claiming vertices from the hypergraph with edges corresponding to solutions to Ax = b. Maker's aim here is to obtain an edge of the hypergraph; if Maker fails to claim an edge, her set is an independent set and therefore a solution-free set.

When $X := [n]_p$, we determine the threshold probability p_0 for when the game is Maker or Breaker's win, for a large class of systems of linear equations. This class includes but is not limited to all single linear equations. The Maker's win statement also extends to a much wider class of systems of linear equations, which include those which satisfy Rado's partition theorem. Its proof involves the use of the container method. The proof of the Breaker's win statement draws on the method used by Rödl and Ruciński [93] to prove the 0-statement of the random Rado theorem. This chapter is based on the author's recently submitted paper [54].

1.3 Notation and preliminaries

1.3.1 Solutions to systems of linear equations

Throughout the thesis, unless otherwise stated, we will assume that A is a fixed integervalued matrix of dimension $\ell \times k$ and b a fixed integer-valued vector of dimension ℓ . We will let $\mathcal{L}(A, b)$ denote the associated system of linear equations Ax = b, noting that for brevity we will simply write \mathcal{L} if it is clear from the context which matrix A and vector bit refers to.

It is important to observe that in the problem of avoiding solutions to a given system of linear equations, there are a few different notions of what constitutes a solution. For example in the case of sum-free sets, should we care or not about whether we include solutions with repetition, e.g. (3,3,6), or not? Historically most problems have focused on studying the case where repetition is not allowed, however in some cases (especially Chapters 3 and 4) we wish to allow so called non-trivial solutions, which we will define shortly.

Given a set of integers X, first let $S(\mathcal{L}^s, X)$ be the set of all vectors $x = (x_1, \ldots, x_k) \in X^k$ such that Ax = b (i.e. the vector x is a solution to \mathcal{L} in X). If $S(\mathcal{L}^s, X)$ is empty then we call X strongly \mathcal{L} -free. If for a solution x, additionally the x_i are pairwise distinct, we call x a k-distinct solution to \mathcal{L} in X. Let $S(\mathcal{L}^d, X)$ refer to the set of all k-distinct solutions to \mathcal{L} in X. If $S(\mathcal{L}^d, X)$ is empty then we call X distinct \mathcal{L} -free.

Generally, we call a system of linear equations \mathcal{L} homogeneous if b = 0. In the case where we have a $1 \times k$ matrix, we simply have a linear equation of the form

$$a_1x_1 + \dots + a_kx_k = b \tag{1.3.1}$$

where $a_1, \ldots, a_k, b \in \mathbb{Z}$. If

$$\sum_{i \in [k]} a_i = b = 0$$

then we say that \mathcal{L} is translation-invariant. Let \mathcal{L} be a translation-invariant linear equation. Then notice that (x, \ldots, x) is a 'trivial' solution of (1.3.1) for any x. More generally, a solution (x_1, \ldots, x_k) to \mathcal{L} is said to be *trivial* if there exists a partition P_1, \ldots, P_ℓ of [k]so that:

- (i) $x_i = x_j$ for every i, j in the same partition class P_r ;
- (ii) For each $r \in [\ell]$, $\sum_{i \in P_r} a_i = 0$.

Let $S(\mathcal{L}^w, X)$ refer to the set of all non-trivial solutions to \mathcal{L} in X. If $S(\mathcal{L}^w, X)$ is empty then call X weakly \mathcal{L} -free. This definition of non-trivial was introduced by Ruzsa in [100]. Note that for non-translation-invariant \mathcal{L} there are no trivial solutions, and so the definition of weakly \mathcal{L} -free coincides with the definition of strongly \mathcal{L} -free. A definition of non-trivial for $\ell \times k$ matrices is given in [99], though we do not state it here since this case is not studied in this thesis. Note that we have

$$S(\mathcal{L}^d, X) \subseteq S(\mathcal{L}^w, X) \subseteq S(\mathcal{L}^s, X).$$

For shorthand notation, for $t \in \{d, w, s\}$ we will use \mathcal{L}^t -free for distinct/weakly/strongly \mathcal{L} -free sets respectively. We will also use $\mu(n, \mathcal{L}^t)$ to denote the size of the largest \mathcal{L}^t -free subset of [n]; observe that

$$\mu(n, \mathcal{L}^s) \leqslant \mu(n, \mathcal{L}^w) \leqslant \mu(n, \mathcal{L}^d).$$

If the equation \mathcal{L} (and type of solutions we are interested in) is clear from the context, then we simply say X is *solution-free*. In Chapters 3 and 4 we consider weakly \mathcal{L} -free sets, whereas in Chapters 5 and 6 we mainly consider distinct \mathcal{L} -free sets.

We also require definitions for r-colourings of sets of integers. Fix $r \in \mathbb{N}$ and $t \in \{d, w, s\}$. For each $i \in [r]$ let A_i be a matrix of dimension $\ell_i \times k_i$ and b_i a vector of dimension ℓ_i . We say a set X is $(\mathcal{L}_1^t, \ldots, \mathcal{L}_r^t)$ -free if there exists an r-colouring of X so that for each $i \in [r]$, the subset $Y \subseteq X$ with colour i is \mathcal{L}_i^t -free. We write $\mu(n, \mathcal{L}_1^t, \ldots, \mathcal{L}_r^t)$ to denote the size of the largest $(\mathcal{L}_1^t, \ldots, \mathcal{L}_r^t)$ -free subset of [n], and write $\mu(n, \mathcal{L}^t, r) := \mu(n, \mathcal{L}_1^t, \ldots, \mathcal{L}_r^t)$ if $\mathcal{L} = \mathcal{L}_i$ for all $i \in [r]$.

We also include some definitons which are important in both Chapters 5 and 6. We call a system of linear equations \mathcal{L} (and the matrix A in the case where b = 0) *irredundant* if there exists a k-distinct solution to Ax = b in \mathbb{N} , and *redundant* otherwise.

Call \mathcal{L} (and again A if b = 0) partition regular if for any finite colouring of \mathbb{N} , there is always a monochromatic solution (of any kind) to Ax = b.

Let (*) be the following matrix property:

(*) Under Gaussian elimination A does not have any row which consists of precisely

two non-zero rational entries.

Index the columns of an $\ell \times k$ matrix A by [k]. For a partition $W \cup \overline{W} = [k]$ of the columns of A, we denote by $A_{\overline{W}}$ the matrix obtained from A by restricting to the columns indexed by \overline{W} . Let rank $(A_{\overline{W}})$ be the rank of $A_{\overline{W}}$, where rank $(A_{\overline{W}}) = 0$ for $\overline{W} = \emptyset$. We set

$$m(A) := \max_{\substack{W \cup \overline{W} = [k] \\ |W| \ge 2}} \frac{|W| - 1}{|W| - 1 + \operatorname{rank}(A_{\overline{W}}) - \operatorname{rank}(A)}.$$
(1.3.2)

We remark that the denominator of m(A) is strictly positive provided that A is irredundant and satisfies (*). (Note that in Section 5.3.1 we will show that irredundant partition regular matrices are a strict subclass of irredundant matrices which satisfy (*).)

1.3.2 Notation

Let \mathcal{H} be a (hyper)graph. We write $V(\mathcal{H})$, $E(\mathcal{H})$ and $\mathcal{I}(\mathcal{H})$ to represent the vertex set, edge set and set of independent sets of \mathcal{H} , and $v(\mathcal{H})$, $e(\mathcal{H})$ and $i(\mathcal{H})$ for the respective numbers of each of these. Consider any subset $X \subseteq V(\mathcal{H})$. Let $\mathcal{H}[X]$ denote the induced subgraph of \mathcal{H} on the vertex set X and $\mathcal{H} \setminus X$ denote the induced subgraph of \mathcal{H} on the vertex set $V(\mathcal{H}) \setminus X$. For an edge set $Y \subseteq E(\mathcal{H})$, we define $\mathcal{H} - Y$ to be hypergraph with vertex set $V(\mathcal{H})$ and edge set $E(\mathcal{H}) \setminus Y$.

For a set X and a positive integer x, we define $\binom{X}{x}$ to be the set of all subsets of X of size x, and we define $\binom{X}{\leq x}$ to be the set of all subsets of X of size at most x. We use $\mathcal{P}(X)$ to denote the powerset of X, that is, the set of all subsets of X. If B is a family of subsets of X, then we define \overline{B} to be the complement family, that is, precisely the subsets of X which are not in B.

Given a hypergraph \mathcal{H} , for each $T \subseteq V(\mathcal{H})$, we define $\deg_{\mathcal{H}}(T) := |\{e \in E(\mathcal{H}) : T \subseteq e\}|$, and let $\Delta_{\ell}(\mathcal{H}) := \max\{\deg_{\mathcal{H}}(T) : T \subseteq V(\mathcal{H}) \text{ and } |T| = \ell\}.$

We write $x = a \pm b$ to say that the value of x is some real number in the interval [a - b, a + b].

For two constants $\alpha, \beta > 0$ we use the notation $\alpha \ll \beta$ (often within a hierarchy of constants) to mean that α is bounded by some unspecified function of β , so that the calculations we wish to hold concerning α and β do indeed hold.

1.3.3 Probabilistic tools

We will need the Markov inequality and Chernoff bounds of the following form (see e.g. [64, Theorem 2.1, Corollary 2.3]).

Proposition 1.1. Let X be a non-negative random variable. Then for all t > 0 we have $\mathbb{P}[X \ge t] \le \frac{\mathbb{E}[X]}{t}$.

Proposition 1.2. Suppose X has binomial distribution.

(i) For every $\lambda \ge 0$, we have

$$\mathbb{P}[X > \mathbb{E}[X] + \lambda] \leqslant \exp\left(-\frac{\lambda^2}{2(\mathbb{E}[X] + \lambda/3)}\right).$$

(ii) For every $0 < \varepsilon \leq 3/2$, we have

$$\mathbb{P}[|X - \mathbb{E}[X]| \ge \varepsilon \mathbb{E}[X]] \le 2 \exp\left(-\frac{\varepsilon^2}{3}\mathbb{E}[X]\right).$$

CHAPTER 2

CONTAINER AND REMOVAL LEMMAS

Here we introduce the container and removal lemmas which will be required in the thesis. Recall that the method of hypergraph containers was greatly advanced by the papers of Balogh, Morris and Samotij [8] and Saxton and Thomason [106]. The first result we require here is that of Balogh, Morris and Samotij. First we require some definitions. Let \mathcal{H} be a k-uniform hypergraph. A family of sets $\mathcal{F} \subseteq \mathcal{P}(V(\mathcal{H}))$ is called *increasing* if it is closed under taking supersets; in other words for every $A, B \subseteq V(\mathcal{H})$, if $A \in \mathcal{F}$ and $A \subseteq B$, then $B \in \mathcal{F}$. Suppose \mathcal{F} is an increasing family of subsets of $V(\mathcal{H})$ and let $\varepsilon \in (0, 1]$. We say that \mathcal{H} is $(\mathcal{F}, \varepsilon)$ -dense if $e(\mathcal{H}[A]) \ge \varepsilon e(\mathcal{H})$ for every $A \in \mathcal{F}$.

Theorem 2.1 ([8], Theorem 2.2). For every $k \in \mathbb{N}$ and all positive c and ε , there exists a positive constant C such that the following holds. Let \mathcal{H} be a k-uniform hypergraph and let $\mathcal{F} \subseteq \mathcal{P}(V(\mathcal{H}))$ be an increasing family of sets such that $|A| \ge \varepsilon v(\mathcal{H})$ for all $A \in \mathcal{F}$. Suppose that \mathcal{H} is $(\mathcal{F}, \varepsilon)$ -dense and $p \in (0, 1)$ is such that, for every $\ell \in [k]$,

$$\Delta_{\ell}(\mathcal{H}) \leqslant c \cdot p^{\ell-1} \frac{e(\mathcal{H})}{v(\mathcal{H})}.$$

Then there exists a family $S \subseteq \binom{V(\mathcal{H})}{\leqslant C_P \cdot v(\mathcal{H})}$ and functions $f : S \to \overline{\mathcal{F}}$ and $g : \mathcal{I}(\mathcal{H}) \to S$ such that for every $I \in \mathcal{I}(\mathcal{H})$, we have that $g(I) \subseteq I$ and $I \setminus g(I) \subseteq f(g(I))$.

Using the above notation, we refer to the set $\mathcal{C} := \{f(g(I)) \cup g(I) : I \in \mathcal{I}(\mathcal{H})\}$ as a set of *containers* and the $g(I) \in \mathcal{S}$ as *fingerprints*. We use this result to derive an analogous result for r-tuples of disjoint independent sets (see Theorems 5.14 and 5.15 in Chapter 5).

As mentioned in the introduction, container results often come side by side with a removal lemma or supersaturation lemma. The removal lemma from which other results we will require are a consequence of, is the following result of Král', Serra and Vena [74].

Lemma 2.2 ([74], **Theorem 2**). Let A be an $\ell \times k$ integer matrix of rank ℓ and b an integer vector of dimension ℓ . For every $\delta > 0$ there exist $n_0, \varepsilon > 0$ with the following property. Suppose $n \ge n_0$ is an integer and for $X \subseteq [n]$, there are at most $\varepsilon n^{k-\ell}$ solutions in $S(\mathcal{L}^s, X)$. Then we can write $X = B \cup C$ where B is strongly \mathcal{L} -free and $|C| \le \delta n$.

The result is stated with b = 0 in [74], though clearly can be extended to general b as stated above. Also within Lemma 2.2, since B is \mathcal{L}^s -free, it is also \mathcal{L}^w -free and \mathcal{L}^d -free. Thus, using little-o notation, we have $|X| \leq \mu(n, \mathcal{L}^w) + o(n)$ and $|X| \leq \mu(n, \mathcal{L}^d) + o(n)$. Hence we obtain the following r-colour supersaturation lemma.

Lemma 2.3. Fix $r \in \mathbb{N}$, $t \in \{d, w, s\}$ and for each $i \in [r]$, let A_i be an $\ell_i \times k_i$ integer matrix of rank ℓ_i and b_i be an integer vector of dimension ℓ_i . For every $\delta > 0$ there exist $n_0, \varepsilon > 0$ with the following property. Suppose $n \ge n_0$ is an integer and $X \subseteq [n]$ is r-coloured, and $|X| \ge \mu(n, \mathcal{L}_1^t, \ldots, \mathcal{L}_r^t) + \delta n$. Then there exists an $i \in [r]$ such that there are more than $\varepsilon n^{k_i - \ell_i}$ solutions in $S(\mathcal{L}_i^t, X)$ in colour i.

We use Lemma 2.3 in Chapters 5 and 6. Note that Lemma 2.2 extends an earlier removal lemma of Green for single linear equations, which we state here since we will use it directly in Chapters 3 and 4.

Lemma 2.4 ([48]). Fix a k-variable homogeneous linear equation \mathcal{L} . Suppose that $A \subseteq [n]$ is a set containing $o(n^{k-1})$ non-trivial solutions to \mathcal{L} . Then there exist B and C such that $A = B \cup C$ where B is weakly \mathcal{L} -free and |C| = o(n).

Finally, we also state Green's container lemma for single linear equations (Proposition 9.1 of [48]), since it will also be directly used in Chapters 3 and 4. Lemma 2.5(i)–(iii) is stated explicitly in Proposition 9.1 of [48]. Lemma 2.5(iv) follows as an immediate consequence of Lemma 2.5(i) and Lemma 2.4 above.

Lemma 2.5 ([48]). Fix a k-variable homogeneous linear equation \mathcal{L} . There exists a family \mathcal{F} of subsets of [n] with the following properties:

- (i) Every $F \in \mathcal{F}$ has at most $o(n^{k-1})$ non-trivial solutions to \mathcal{L} .
- (ii) If $S \subseteq [n]$ is weakly \mathcal{L} -free, then S is a subset of some $F \in \mathcal{F}$.
- (*iii*) $|\mathcal{F}| = 2^{o(n)}$.
- (iv) Every $F \in \mathcal{F}$ has size at most $\mu(n, \mathcal{L}^w) + o(n)$.

Note that Lemma 2.5 can be recovered from Theorem 2.1 and Lemma 2.2. In fact by using our r-colour container result (Theorem 5.15) we obtain an r-colour version of Lemma 2.5 (see Theorem 5.21).

CHAPTER 3

Solution-free sets of integers

3.1 Introduction

Recall from the introduction that a set $A \subseteq [n]$ is weakly \mathcal{L} -free if A does not contain any non-trivial solutions to \mathcal{L} in [n]. We will use the notation \mathcal{L}^w -free for weakly \mathcal{L} -free sets, and throughout this chapter and the next, only consider weakly \mathcal{L} -free sets.

The notion of an \mathcal{L}^w -free set encapsulates many fundamental topics in combinatorial number theory. Recall that in the case when \mathcal{L} is $x_1 + x_2 = x_3$ we call an \mathcal{L}^w -free set (or equivalently an \mathcal{L}^s -free set, since $x_1 + x_2 = x_3$ is not translation-invariant) a sumfree set. This is a notion that dates back to 1916 when Schur [110] proved that, if n is sufficiently large, any r-colouring of [n] yields a monochromatic triple x, y, z such that x + y = z. Sidon sets (when \mathcal{L} is $x_1 + x_2 = x_3 + x_4$) have also been extensively studied. For example, a classical result of Erdős and Turán [38] asserts that the largest Sidon set in [n] has size $(1+o(1))\sqrt{n}$. In the case when \mathcal{L} is $x_1 + x_2 = 2x_3$ an \mathcal{L}^w -free set is simply a progression-free set. Roth's theorem [98] states that the largest progression-free subset of [n] has size o(n). In [100, 101], Ruzsa instigated the study of solution-free sets for general linear equations.

In this chapter we prove a number of results concerning \mathcal{L}^{w} -free subsets of [n] where \mathcal{L}

is a homogeneous linear equation in *three variables*. In particular, our work is motivated by the following general questions:

- (i) What is the size of the largest \mathcal{L}^w -free subset of [n]?
- (ii) How many \mathcal{L}^w -free subsets of [n] are there?
- (iii) How many maximal \mathcal{L}^{w} -free subsets of [n] are there?

We make progress on all three of these questions. For each question we use tools from graph theory; for (i) and (ii) our methods are somewhat elementary. For (iii) our method is more involved and utilises container and removal lemmas of Green [48].

3.1.1 The size of the largest solution-free set

As highlighted above, a central question in the study of \mathcal{L}^w -free sets is to establish the size $\mu(n, \mathcal{L}^w)$ of the largest \mathcal{L}^w -free subset of [n]. It is not difficult to see that the largest sum-free subset of [n] has size $\lceil n/2 \rceil$, and this bound is attained by the set of odd numbers in [n] and by the interval $\lceil n/2 \rceil + 1, n \rceil$.

When \mathcal{L} is $x_1 + x_2 = 2x_3$, $\mu(n, \mathcal{L}^w) = o(n)$ by Roth's theorem. In fact, very recently Bloom [18] proved that there is a constant C such that every set $A \subseteq [n]$ with $|A| \ge Cn(\log \log n)^4 / \log n$ contains a three-term arithmetic progression. On the other hand, Behrend [15] showed that there is a constant c > 0 so that $\mu(n, \mathcal{L}^w) \ge n \exp(-c\sqrt{\log n})$. See [32, 50] for the best known lower bound on $\mu(n, \mathcal{L}^w)$ in this case.

More generally, it is known that $\mu(n, \mathcal{L}^w) = o(n)$ if \mathcal{L} is homogeneous and translationinvariant, and $\mu(n, \mathcal{L}^w) = \Omega(n)$ otherwise (see [100]). For other (exact) bounds on $\mu(n, \mathcal{L}^w)$ for various linear equations \mathcal{L} see, for example, [100, 101, 11, 31, 53].

In this chapter we mainly focus on \mathcal{L}^w -free subsets of [n] for linear equations \mathcal{L} of the form px + qy = z where $p \ge 2$ and $q \ge 1$ are fixed integers. For such equations the set of \mathcal{L}^w -free sets are precisely the same as the set of \mathcal{L}^s -free sets since \mathcal{L} is not translation-

invariant. Also notice that for such \mathcal{L} , the interval $[\lfloor n/(p+q) \rfloor + 1, n]$ is an \mathcal{L}^w -free set. Our first result implies that this is the largest such \mathcal{L}^w -free subset of [n].

Theorem 3.1. Let \mathcal{L} denote the equation px + qy = z where $p \ge q$ and $p \ge 2$, $p, q \in \mathbb{N}$. Let S be an \mathcal{L}^w -free subset of [n], and let $\min(S) = \lfloor \frac{n}{p+q} \rfloor - t$ where t is a non-negative integer.

(i) If
$$0 \leq t < \left(\frac{p+q-1}{p+q+p/q}\right) \lfloor \frac{n}{p+q} \rfloor$$
 then $|S| \leq \left\lceil \frac{(p+q-1)n}{p+q} \rceil - \lfloor \frac{p}{q}t \rfloor$.

(ii) If
$$t \ge \left(\frac{p+q-1}{p+q+p/q}\right) \lfloor \frac{n}{p+q} \rfloor$$
 then $|S| \le \frac{(q^2+1)n}{q^2+q+1}$ provided that $n \ge \frac{3(q^2+q+1)(q^3+p(q^2+q+1))}{q^2+1}$ and $n \ge \frac{5(q^2+q+1)(q^5+p(q^4+q^3+q^2+q+1))}{q^4+(p-1)q^3+q^2+1}$.

In both cases of Theorem 3.1 we observe that $|S| \leq n - \lfloor \frac{n}{p+q} \rfloor$, hence the following corollary holds.

Corollary 3.2. Let \mathcal{L} denote the equation px + qy = z where $p \ge q$ and $p \ge 2$, $p, q \in \mathbb{N}$. If n is sufficiently large depending on p and q, then $\mu(n, \mathcal{L}^w) = n - \lfloor \frac{n}{p+q} \rfloor$.

Roughly, Theorem 3.1 implies that every \mathcal{L}^w -free subset of [n] is 'interval like' or 'small'. In the case of sum-free subsets (i.e. when p = q = 1), a result of Deshouillers, Freiman, Sós and Temkin [30] provides very precise structural information on the sum-free subsets of [n]. Loosely speaking, they showed that a sum-free subset of [n] is 'interval like', 'small' or consists entirely of odd numbers.

In the case when p = q, Corollary 3.2 was proven by Hegarty [53] (without a lower bound on n).

3.1.2 The number of solution-free sets

Write $f(n, \mathcal{L}^w)$ for the number of \mathcal{L}^w -free subsets of [n]. In the case when \mathcal{L} is x + y = z, define $f(n) := f(n, \mathcal{L}^w)$.

By considering all possible subsets of [n] consisting of odd numbers, one observes that there are at least $2^{n/2}$ sum-free subsets of [n]. Cameron and Erdős [23] conjectured that in fact $f(n) = \Theta(2^{n/2})$. This conjecture was proven independently by Green [47] and Sapozhenko [104]. In fact, they showed that there are constants C_1 and C_2 such that $f(n) = (C_i + o(1))2^{n/2}$ for all $n \equiv i \mod 2$.

Results from [71, 107] imply that there are between $2^{(1.16+o(1))\sqrt{n}}$ and $2^{(6.45+o(1))\sqrt{n}}$ Sidon sets in [n]. There are also several results concerning the number of so-called (k, ℓ) -sum-free subsets of [n] (see, e.g., [17, 22, 109]).

More generally, given a linear equation \mathcal{L} , there are at least $2^{\mu(n,\mathcal{L}^w)} \mathcal{L}^w$ -free subsets of [n]. In light of the situation for sum-free sets one may ask whether, in general, $f(n,\mathcal{L}^w) = \Theta(2^{\mu(n,\mathcal{L}^w)})$. However, Cameron and Erdős [23] observed that this is false for homogeneous translation-invariant \mathcal{L} . In particular, given such an \mathcal{L}^w -free set, any translation of it is also \mathcal{L}^w -free.

Green [48] though showed that given a homogeneous linear equation \mathcal{L} , $f(n, \mathcal{L}^w) = 2^{\mu(n, \mathcal{L}^w) + o(n)}$ (where here the o(n) may depend on \mathcal{L}). Our next result implies that one can omit the term o(n) in the exponent for certain types of linear equation \mathcal{L} .

Theorem 3.3. Fix $p, q \in \mathbb{N}$ where (i) $q \ge 2$ and p > q(3q+2)/(2q-2) or (ii) q = 1 and $p \ge 3$. Let \mathcal{L} denote the equation px + qy = z. Then

$$f(n, \mathcal{L}^w) = \Theta(2^{\mu(n, \mathcal{L}^w)}).$$

3.1.3 The number of maximal solution-free sets

Given a linear equation \mathcal{L} , we say that $S \subseteq [n]$ is a maximal \mathcal{L}^w -free subset of [n] if it is \mathcal{L}^w -free and it is not properly contained in another \mathcal{L}^w -free subset of [n]. Write $f_{\max}(n, \mathcal{L}^w)$ for the number of maximal \mathcal{L}^w -free subsets of [n]. In the case when \mathcal{L} is x + y = z, define $f_{\max}(n) := f_{\max}(n, \mathcal{L}^w)$.

A significant proportion of the sum-free subsets of [n] lie in just two maximal sumfree sets, namely the set of odd numbers in [n] and the interval $[\lfloor n/2 \rfloor + 1, n]$. This led Cameron and Erdős [24] to ask whether $f_{\max}(n) = o(f(n))$ or even $f_{\max}(n) \leq f(n)/2^{\varepsilon n}$ for some constant $\varepsilon > 0$. Luczak and Schoen [79] answered this question in the affirmative, showing that $f_{\max}(n) \leq 2^{n/2-2^{-28}n}$ for sufficiently large n. Later, Wolfovitz [121] proved that $f_{\max}(n) \leq 2^{3n/8+o(n)}$. Very recently, Balogh, Liu, Sharifzadeh and Treglown [6, 7] proved the following: For each $1 \leq i \leq 4$, there is a constant C_i such that, given any $n \equiv i \mod 4$, $f_{\max}(n) = (C_i + o(1))2^{n/4}$.

Except for sum-free sets, the problem of determining the number of maximal solutionfree subsets of [n] remains wide open. In this chapter we give a number of bounds on $f_{\max}(n, \mathcal{L}^w)$ for homogeneous linear equations \mathcal{L} in three variables. The next result gives a general upper bound for such \mathcal{L} . Given a three-variable linear equation \mathcal{L} , an \mathcal{L} -triple is a multiset $\{x, y, z\}$ which forms a solution to \mathcal{L} . (In other words, the set of all \mathcal{L} -triples in X corresponds to $S(\mathcal{L}^s, X)$.) Let $\mu^*(n, \mathcal{L})$ denote the number of elements $x \in [n]$ that do not lie in any \mathcal{L} -triple in [n].

Theorem 3.4. Let \mathcal{L} be a fixed homogenous three-variable linear equation. Then

$$f_{\max}(n, \mathcal{L}^w) \leqslant 3^{(\mu(n, \mathcal{L}^w) - \mu^*(n, \mathcal{L}))/3 + o(n)}.$$

Theorem 3.4 together with the aforementioned result of Green shows that $f_{\max}(n, \mathcal{L}^w)$ is significantly smaller than $f(n, \mathcal{L}^w)$ for all homogeneous three-variable linear equations \mathcal{L} that are not translation-invariant. So in this sense it can be viewed as a generalisation of the result of Luczak and Schoen. The proof of Theorem 3.4 is a simple application of container and removal lemmas of Green [48]. The same idea was used to prove results in [10, 6, 7]. Although at first sight the bound in Theorem 3.4 may seem crude, perhaps surprisingly there are equations \mathcal{L} where the value of $f_{\max}(n, \mathcal{L}^w)$ is close to this bound (see Proposition 3.19 in Section 3.5).

On the other hand, the following result shows that there are linear equations where

the bound in Theorem 3.4 is far from tight.

Theorem 3.5. Let \mathcal{L} denote the equation px + qy = z where $p \ge q \ge 2$ are integers so that $p \le q^2 - q$ and gcd(p,q) = q. Then

$$f_{\max}(n, \mathcal{L}^w) \leqslant 2^{(\mu(n, \mathcal{L}^w) - \mu^*(n, \mathcal{L}))/2 + o(n)}.$$

In the case when \mathcal{L} is the equation 2x + 2y = z we provide a matching lower bound. Again though, we suspect there are equations \mathcal{L} where the bound in Theorem 3.5 is far from tight. The proof of Theorem 3.5 applies Theorem 3.1 as well as the container and removal lemmas of Green [48].

We also provide another upper bound on $f_{\max}(n, \mathcal{L}^w)$ for a more general class of linear equations.

Theorem 3.6. Let \mathcal{L} denote the equation px + qy = z where $p \ge q$, $p \ge 2$ and $p, q \in \mathbb{N}$. Then

$$f_{\max}(n, \mathcal{L}^w) \leqslant 2^{\mu(\lfloor \frac{n-p}{q} \rfloor, \mathcal{L}^w) + o(n)}$$

Further, if $q \ge 2$ and p > q(3q-2)/(2q-2) or q = 1 and $p \ge 3$ then

$$f_{\max}(n, \mathcal{L}^w) = O(2^{\mu(\lfloor \frac{n-p}{q} \rfloor, \mathcal{L}^w)}).$$

In Section 3.5 we provide lower bounds on $f_{\max}(n, \mathcal{L}^w)$ for all equations \mathcal{L} of the form px + qy = z where $p, q \ge 2$ are integers; see Proposition 3.21.

Our results suggest that, in contrast to the case of $f(n, \mathcal{L}^w)$, it is unlikely there is a 'simple' general asymptotic formula for $f_{\max}(n, \mathcal{L}^w)$ for all homogeneous linear equations \mathcal{L} . It would be extremely interesting to make further progress on this problem.

The chapter is organised as follows. In the next section we collect together a number of useful tools. In Section 3.3 we prove Theorem 3.1. Theorem 3.3 is proven in Section 3.4. We prove our results on the number of maximal \mathcal{L}^w -free sets in Section 3.5. In Chapter 4 we obtain further results on the number of maximal \mathcal{L}^w -free sets, and so see Section 4.6 for a note on which of our results produces the best upper bound on $f_{\max}(n, \mathcal{L}^w)$ for a given equation \mathcal{L} .

3.2 Containers and independent sets in graphs

3.2.1 Container and removal lemmas

Recall from the introduction that we can phrase the problem in terms of independent sets in hypergraphs. Let \mathcal{H} denote the hypergraph with vertex set [n] and edges corresponding to non-trivial solutions to \mathcal{L} . Then an independent set in \mathcal{H} is precisely an \mathcal{L}^w -free set. We will use the removal and container results of Green from Chapter 2 (that is, Lemmas 2.4 and 2.5), and the following result (which is an immediate consequence).

Theorem 3.7 ([48]). Fix a homogeneous linear equation \mathcal{L} . Then $f(n, \mathcal{L}^w) = 2^{\mu(n, \mathcal{L}^w) + o(n)}$.

We will use these results to deduce upper bounds on the number of maximal \mathcal{L}^{w} -free sets (Theorems 3.4, 3.5 and 3.6).

3.2.2 Independent sets in graphs

First observe the following obvious bound on the number of independent sets in a graph.

Fact 3.8. Let G be a graph and let A_1, \ldots, A_r be a partition of V(G). Then $i(G) \leq \prod_{i=1}^r i(G[A_i])$.

The following simple lemma will be used in the proof of Theorem 3.3.

Lemma 3.9. Let G be a graph on n vertices and M be a matching in G which consists of e edges. Suppose that $v \in V(G)$ lies in M. Then the number of independent sets in G which contain v is at most $3^{e-1} \cdot 2^{n-2e}$. **Proof.** First note that the number of independent sets in G which contain v is at most $i(G \setminus X)$ where X consists of v and its neighbour in M. Let A_1, \ldots, A_e be a partition of the vertex set $V(G \setminus X)$, where if $1 \leq i \leq e - 1$ then A_i contains precisely the two vertices from some edge in M. So $|A_e| = n - 2e$. Clearly $i(G[A_i]) = 3$ for $1 \leq i \leq e - 1$ and $i(G[A_e]) \leq 2^{n-2e}$. The result then follows by Fact 3.8.

3.2.3 Link graphs and maximal independent sets

We obtain many of our results by counting the number of maximal independent sets in various auxiliary graphs. Similar techniques were used in [121, 6, 7], and in the graph setting in [10, 5]. To be more precise, let B and S be disjoint subsets of [n] and fix a three-variable linear equation \mathcal{L} . The link graph $L_S[B]$ of S on B has vertex set B, and an edge set consisting of the following two types of edges:

- (i) Two vertices x and y are adjacent if there exists an element $z \in S$ such that $\{x, y, z\}$ is an \mathcal{L} -triple;
- (ii) There is a loop at a vertex x if there exists an element $z \in S$ or elements $z, z' \in S$ such that $\{x, x, z\}$ or $\{x, z, z'\}$ is an \mathcal{L} -triple.

Notice that since the only possible trivial solutions to a three-variable linear equation \mathcal{L} are of the form $\{x, x, x\}$, all the edges in $L_S[B]$ correspond to non-trivial \mathcal{L} -triples.

The following simple lemma was stated in [6, 7] for sum-free sets, but extends to three-variable linear equations.

Lemma 3.10. Fix a three-variable linear equation \mathcal{L} . Suppose that B, S are disjoint \mathcal{L}^w -free subsets of [n]. If $I \subseteq B$ is such that $S \cup I$ is a maximal \mathcal{L}^w -free subset of [n], then I is a maximal independent set in $G := L_S[B]$.

Let MIS(G) denote the number of maximal independent sets in G. Suppose we have a container $F \in \mathcal{F}$ as in Lemma 2.5 and suppose $F = A \cup B$ where B is \mathcal{L}^w -free. Observe that any maximal \mathcal{L}^{w} -free subset of [n] in F can be found by first choosing an \mathcal{L}^{w} -free set $S \subseteq A$, and then extending S in B. Note that by Lemma 3.10, the number of possible extensions of S in B (which we shall refer to as N(S, B)) is bounded from above by the number of maximal independent sets in the link graph $L_{S}[B]$ (i.e. we have $N(S, B) \leq \text{MIS}(L_{S}[B])$). Hence Lemma 3.10 is a useful tool for bounding the number of maximal \mathcal{L}^{w} -free subsets of [n].

In particular, we will apply the following result in combination with Lemma 3.10. The first part was proven by Moon and Moser [82] and the second part by Hujter and Tuza [62]. We use the first condition in the proof of Theorems 3.4 and 3.5.

Theorem 3.11. Suppose that G is a graph on n vertices possibly with loops. Then the following bounds hold.

- (i) $MIS(G) \leq 3^{n/3}$;
- (ii) $MIS(G) \leq 2^{n/2}$ if G is additionally triangle-free.

To prove Theorem 3.5 we will combine Theorem 3.11(ii) and the following result.

Lemma 3.12. Let \mathcal{L} denote the equation px + qy = z where $p \ge q \ge 2$ and $p, q \in \mathbb{N}$. Let $A \subseteq [1, u]$ and let $B \subseteq [u + 1, n]$ for some $u \in [n]$. Consider the link graph $G := L_A[B]$ of A on B. If $q^2 \ge p + q$ then G is triangle-free.

Proof. Suppose that $q^2 \ge p + q$ and suppose for a contradiction there is a triangle in G with vertices $b_1 < b_2 < b_3$. By definition of the link graph, there exist $s_1, s_2, s_3 \in A$ such that $\{b_1, b_2, s_1\}, \{b_2, b_3, s_2\}, \{b_1, b_3, s_3\}$ are \mathcal{L} -triples.

Since all numbers in A are smaller than all numbers in B we have $1 \leq s_1, s_2, s_3 < b_1 < b_2 < b_3$. Also, since $p \geq q \geq 2$, for each of our \mathcal{L} -triples $\{b_i, b_j, s_k\}$ (where $b_i < b_j$) it follows that b_j must play the role of z in \mathcal{L} .

Let $r_i \in \{p,q\}$ for each $i \in [6]$, where $r_1 \neq r_2$, $r_3 \neq r_4$ and $r_5 \neq r_6$, and consider the three equations $r_1b_1 + r_2s_1 = b_2$, $r_3b_2 + r_4s_2 = b_3$ and $r_5b_1 + r_6s_3 = b_3$; observe that each of the possible ordered tuples (r_1, \ldots, r_6) correspond to possible solutions. Combining the second and third equations gives $b_2 = (r_5b_1 + r_6s_3 - r_4s_2)/r_3$. Then combining this with the first equation gives $(r_1r_3 - r_5)b_1 + r_2r_3s_1 + r_4s_2 = r_6s_3$. Now since $s_3 < b_1$ and all terms are at least 1, for such an inequality to hold we must have $r_1r_3 - r_5 < r_6$. Since $r_5 \neq r_6$ this means we have $r_1r_3 < p+q$. Hence as $r_1, r_3 \in \{p,q\}$, in order for G to have a triangle at least one of $p^2 < p+q$, $q^2 < p+q$ and pq < p+q must be satisfied. Since $p \ge q \ge 2$, the first and third are not true and so we must have $q^2 < p+q$, a contradiction.

We also use link graphs as a means to obtain lower bounds on the number of maximal \mathcal{L}^{w} -free sets. We apply the following result in Propositions 3.19 and 3.21.

Lemma 3.13. Fix a three-variable linear equation \mathcal{L} . Suppose that B, S are disjoint \mathcal{L}^w -free subsets of [n]. Let H be an induced subgraph of the link graph $L_S[B]$. Then $f_{\max}(n, \mathcal{L}^w) \ge \mathrm{MIS}(H)$.

Proof. Suppose I and J are different maximal independent sets in H. First note that $S \cup I$ and $S \cup J$ are \mathcal{L}^w -free by definition of the link graph. Both cannot lie in the same maximal \mathcal{L}^w -free subset of [n]. To see this, observe by definition of I and J, there exists $i \in I \setminus J$. There must exist $s \in S$, $j \in J$ such that $\{i, j, s\}$ forms an \mathcal{L} -triple, else $J \cup \{i\}$ would be an independent set in H, which contradicts the maximality of J. Hence any maximal \mathcal{L}^w -free subset of [n] containing $S \cup J$ does not contain i. Similarly there exists $j \in J \setminus I$ such that any maximal \mathcal{L}^w -free subset of [n] containing $S \cup I$ does not contain j. The result immediately follows.

3.3 The size of the largest solution-free set

Throughout this section, \mathcal{L} will denote the equation px + qy = z where $p \ge q$ and $p \ge 2$, $p,q \in \mathbb{N}$. The aim of this section is to determine the size of the largest \mathcal{L}^w -free subset of [n]. In fact, we will prove a richer structural result on \mathcal{L}^w -free sets (Theorem 3.1). For this, we will introduce the following auxiliary graph G_m : Let $m \in [n]$ be fixed. We define the graph G_m to have vertex set [m, n] and edges between c and pm + qc for all $c \in [m, n]$ such that $pm + qc \le n$. We will also make use of these auxiliary graphs in Section 3.4.

Fact 3.14.

- (i) The size of the largest \mathcal{L}^w -free subset S of [n] with $\min(S) = m$ is at most the size of the largest independent set in G_m which contains m.
- (ii) The number of L^w-free subsets S of [n] with min(S) = m is at most the number of independent sets in G_m which contain m.

Proof. Let S be an \mathcal{L}^w -free subset of [n] with $\min(S) = m$. Since $\{m, c, pm + qc\}$ is an \mathcal{L} -triple contained in [n] for all $c \in [m, n]$ such that $pm + qc \leq n$, S cannot contain both c and pm + qc. Hence any \mathcal{L}^w -free subset of [n] with minimum element m is also an independent set in G_m which contains m (although the converse does not necessarily hold). This immediately implies (i) and (ii).

Note that G_m is a union of disjoint paths (and possibly isolated vertices). We refer to the connected components of G_m as the *path components*. Given G_m , we define $y_0 := n$, and for $i \ge 1$ define $y_i := \max\{v \in V(G_m) | pm + qv \le y_{i-1}\}$. Thus we have $y_i = \lfloor \frac{y_{i-1} - pm}{q} \rfloor$. For G_m we also define k to be the largest i such that $y_i \in [m, n]$, and refer to k as the *path parameter* of G_m . We define the *size* of a path component to be the number of vertices in it, and we define $N(G_m, i)$ to be the number of path components of size i in G_m . **Fact 3.15.** The graph G_m consists entirely of disjoint path components, where for each $1 \leq i \leq k-1$ there are $y_{i-1}+y_{i+1}-2y_i$ path components of size i, there are $y_{k-1}-2y_k+m-1$ path components of size k and $y_k - m + 1$ path components of size k + 1.

Proof. Every vertex $c \in V(G_m)$ satisfying $y_{j+1} < c \leq y_j$ for some $0 \leq j \leq k-1$ is in a path in G_m which contains precisely j vertices which are larger than it, whereas every vertex $c > y_j$ is not in such a path. All the vertices in $[m, y_k]$ are in paths which contain precisely k vertices which are larger than it, all vertices in $[y_k+1, y_{k-1}]$ are in paths which contain precisely k-1 vertices which are larger than it, and so on.

Let A_i be the interval $[y_i + 1, y_{i-1}]$ for $1 \le i \le k$ and let A_{k+1} be the interval $[m, y_k]$. There are $|[m, y_k]| = y_k - m + 1$ path components of size k + 1 in G_m . For $i \le k$ all vertices in A_i are the smallest vertex in a path on i vertices, however they may not be the smallest vertex in their path component. In fact, by definition of the y_i , all paths which start in A_j for some j must include precisely one vertex from each set $A_{j-1}, A_{j-2}, \ldots, A_1$. This means that for $i \le k$, the number of path components of size i in G_m is precisely $|A_i| - |A_{i+1}|$. For $i \le k - 1$ this is $y_{i-1} + y_{i+1} - 2y_i$ and for i = k this is $y_{k-1} - 2y_k + m - 1$.

We use the graphs G_m and the above facts to obtain the bound for the size of the largest \mathcal{L}^w -free subset of [n] as stated in Theorem 3.1. For (ii), we will show that a largest independent set in G_m has size at most $(q^2 + 1)n/(q^2 + q + 1)$. Before going into full details of the proof of Theorem 3.1, we briefly explain why this is a reasonable target for an upper bound.

Since G_m consists of path components of different sizes, one picks an independent set of maximum size by selecting $\lceil k/2 \rceil$ vertices from each path component of size k. (That is, we select one vertex from a path component of size 1 or 2, two vertices from a path component of size 3 or 4, etc.) Note that the ratio of vertices selected (percentage of vertices chosen from a path of given size) is always 1/2 if k is even, while it tends towards 1/2 from above if k is odd and increasing. We show that there is at least one path component of size 3, and thus the 'worst possible' case should be having path components of size at most 3. The relative number of path components of size 3 compared to those of size 2 and 1 leads to the bound $(q^2 + 1)n/(q^2 + q + 1)$. For an example, let \mathcal{L} be the equation 2x + 2y = z, suppose n = 21 and m = 1; see Figure 3.1 below. Here we can see that a largest independent set is $[1,3] \cup [10,21]$; the ratio of selected vertices here is $15/21 = 5/7 = (q^2 + 1)/(q^2 + q + 1)$. We now proceed with the formal proof.



Proof of Theorem 3.1. Let t be a non-negative integer. To prove (i) suppose that $t < (\frac{p+q-1}{p+q+p/q})\lfloor \frac{n}{p+q} \rfloor$. Suppose S is an \mathcal{L}^w -free set contained in $[\lfloor \frac{n}{p+q} \rfloor - t, n]$ where $m := \lfloor \frac{n}{p+q} \rfloor - t \in S$. By Fact 3.14(i) it suffices to prove that the largest independent set in G_m containing m has size at most $\lceil \frac{(p+q-1)n}{p+q} \rceil - \lfloor \frac{p}{q}t \rfloor$. Since $|V(G_m)| = \lceil \frac{(p+q-1)n}{p+q} \rceil + t + 1$ it suffices to show that any independent set I in G_m satisfies $|V(G_m) \setminus I| \ge \lfloor (p+q)t/q \rfloor + 1$.

For $0 \leq i \leq \lfloor (p+q)t/q \rfloor$, there is an edge between m+i and (p+q)m+qi. Note that since $i \leq \lfloor (p+q)t/q \rfloor$ and $q \leq p$ we have that the largest vertex in any of these edges is indeed at most n:

$$(p+q)(\lfloor \frac{n}{p+q} \rfloor - t) + qi \leqslant n - (p+q)t + q\lfloor (p+q)t/q \rfloor \leqslant n - (p+q)t + q(p+q)t/q = n.$$

Since I can only contain one vertex from each of these edges, we have proven (i), provided that these edges are disjoint. It suffices to show that $\lfloor \frac{n}{p+q} \rfloor + \lfloor pt/q \rfloor < (p+q)m =$ $(p+q)(\lfloor \frac{n}{p+q} \rfloor - t)$ since the left hand side is the largest element of the set $\{m+i: 0 \leq i \leq \lfloor (p+q)t/q \rfloor\}$. But this immediately follows since $t < (\frac{p+q-1}{p+q+p/q})\lfloor \frac{n}{p+q} \rfloor$. To prove (ii) let $t \ge (\frac{p+q-1}{p+q+p/q})\lfloor \frac{n}{p+q} \rfloor$ and suppose S is an \mathcal{L}^w -free subset of [n] with $m := \min(S) = \lfloor \frac{n}{p+q} \rfloor - t$. By Fact 3.14(i) |S| is at most the size of the largest independent set in G_m which contains m. We will first show that G_m has path parameter $k \ge 2$, and then the case q = 1 follows easily. Define $\ell := \lfloor k/2 \rfloor$ and

$$C_k := \left(\frac{\sum_{i=0}^{2\ell+1} (-1)(-q)^i + p \sum_{i=0}^{\ell} q^{2i}}{q^{2\ell+1} + p \sum_{i=0}^{2\ell} q^i}\right)$$

We will show that if $q \ge 2$ then the largest independent set in G_m has size at most $C_k n + k$. We then further bound this from above by $(q^2 + 1)n/(q^2 + q + 1)$ for n sufficiently large.

Note that by Fact 3.15, to prove that $k \ge 2$ for G_m it suffices to show that there is a path on 3 vertices in G_m . By definition of k, m lies on a path P on k + 1 vertices. Write $P = v_0 v_1 \cdots v_k$ where $m = v_0$ and observe that $v_j = (q^j + p \sum_{i=0}^{j-1} q^i)m$ for $0 \le j \le k$. To prove $k \ge 2$ it suffices to show that there is indeed a vertex $(q^2 + pq + p)m$ in $V(G_m)$, i.e. $(q^2 + pq + p)m \le n$. Note that since $t \ge (\frac{p+q-1}{p+q+p/q})\lfloor \frac{n}{p+q} \rfloor$, we have $m = \lfloor \frac{n}{p+q} \rfloor - t \le (\frac{p+q+p/q-p-q+1}{p+q+p/q})\lfloor \frac{n}{p+q} \rfloor = (\frac{p+q}{q^2+pq+p})\lfloor \frac{n}{p+q} \rfloor$. Hence $(q^2 + pq + p)m \le n$ as desired.

When q = 1 observe that $y_i = y_{i-1} - pm$, so for $i \leq k - 1$ by Fact 3.15 we have $N(G_m, i) = y_{i-1} + y_{i+1} - 2y_i = y_i + pm + y_i - pm - 2y_i = 0$. Hence G_m consists entirely of a union of path components of size either k or k + 1. Since at most $\lceil i/2 \rceil$ vertices of a path on i vertices can be in an independent set and $k \geq 2$, the largest independent set in G_m has size at most $2n/3 = (q^2 + 1)n/(q^2 + q + 1)$ in this case, as desired. So now consider the case when $q \geq 2$. We calculate the maximum size of an independent set in G_m :

$$\sum_{i=1}^{k+1} \lceil i/2 \rceil \cdot N(G_m, i)$$
$$= \left(\sum_{i=1}^{k-1} \lceil i/2 \rceil \cdot (y_{i-1} + y_{i+1} - 2y_i)\right) + \lceil k/2 \rceil (y_{k-1} + m - 1 - 2y_k) + \lceil (k+1)/2 \rceil (y_k - m + 1) = y_0 + \left(\sum_{i=1}^k (\lceil (i-1)/2 \rceil - 2\lceil i/2 \rceil + \lceil (i+1)/2 \rceil) y_i\right) + (m-1)(\lceil k/2 \rceil - \lceil (k+1)/2 \rceil).$$
(3.3.1)

Here we used Fact 3.15 in the first equality. For *i* odd, the coefficient of y_i in (3.3.1) is (i-1)/2 - 2(i+1)/2 + (i+1)/2 = -1. For *i* even, the coefficient of y_i in (3.3.1) is i/2 - 2i/2 + (i+2)/2 = 1.

The following bounds are obtained from the definition of y_i and k:

(a)
$$\left(n - q^{j} + 1 - pm \sum_{i=0}^{j-1} q^{i}\right)/q^{j} \leq y_{j} \leq \left(n - pm \sum_{i=0}^{j-1} q^{i}\right)/q^{j};$$

(b) $n/\left(q^{k+1} + p \sum_{i=0}^{k} q^{i}\right) < m \leq n/\left(q^{k} + p \sum_{i=0}^{k-1} q^{i}\right).$

Let $\ell := \lfloor k/2 \rfloor$ (note $k \ge 2$ so $\ell \ge 1$). First suppose k is odd, i.e. $k = 2\ell + 1$. Using (3.3.1), the size of the largest independent set in G_m is bounded above by

$$y_{0} + \left(\sum_{i=1}^{k} \left(\left\lceil (i-1)/2 \right\rceil - 2\left\lceil i/2 \right\rceil + \left\lceil (i+1)/2 \right\rceil\right)y_{i}\right) + (m-1)\left(\left\lceil k/2 \right\rceil - \left\lceil (k+1)/2 \right\rceil\right)$$

$$= y_{0} - y_{1} + y_{2} - y_{3} + \dots + y_{2\ell} - y_{2\ell+1}$$

$$\stackrel{(a)}{\leq} n - \left(\frac{n-pm-q+1}{q}\right) + \left(\frac{n-pm(1+q)}{q^{2}}\right) - \left(\frac{n-pm(1+q+q^{2})-q^{3}+1}{q^{3}}\right)$$

$$+\dots - \left(\frac{n - \left(pm\sum_{i=0}^{2\ell} q^i\right) - q^{2\ell+1} + 1}{q^{2\ell+1}}\right)$$

= $n\left(1 - \frac{1}{q} + \frac{1}{q^2} - \dots - \frac{1}{q^{2\ell+1}}\right) + m\left(\frac{p}{q} + \frac{p}{q^3} + \dots + \frac{p}{q^{2\ell+1}}\right) + \frac{q - 1}{q} + \frac{q^3 - 1}{q^3}$
+ $\dots + \frac{q^{2\ell+1} - 1}{q^{2\ell+1}}$

$$\stackrel{\text{(b)}}{\leqslant} \frac{n}{q^{2\ell+1}} \Big(\sum_{i=0}^{2\ell+1} (-1)(-q)^i \Big) + \left(\frac{n}{q^{2\ell+1} + p \sum_{i=0}^{2\ell} q^i} \right) \left(\frac{p \sum_{i=0}^{i} q^{2i}}{q^{2\ell+1}} \right) + \frac{k+1}{2}$$

$$= \left(\frac{\left[\sum_{i=0}^{2\ell+1} (-1)(-q)^i \right] (q^{2\ell+1} + p \sum_{i=0}^{2\ell} q^i) + p \sum_{i=0}^{\ell} q^{2i}}{q^{2\ell+1} (q^{2\ell+1} + p \sum_{i=0}^{2\ell} q^i)} \right) n + \frac{k+1}{2}$$

$$= \left(\frac{\sum_{i=0}^{2\ell+1} (-q)^{i+2\ell+1} + p \sum_{i=0}^{\ell} q^{2i+2\ell+1}}{q^{2\ell+1} (q^{2\ell+1} + p \sum_{i=0}^{2\ell} q^i)} \right) n + \frac{k+1}{2} = \left(\frac{\sum_{i=0}^{2\ell+1} (-1)(-q)^i + p \sum_{i=0}^{\ell} q^{2i}}{q^{2\ell+1} + p \sum_{i=0}^{2\ell} q^i} \right) n + \frac{k+1}{2}$$

$$= C_k n + \frac{k+1}{2} \leqslant C_k n + k.$$

By definition, $m \ge y_{k+1} + 1$ and for k even, we have $C_k = C_{k+1}$. So if k is even $(k = 2\ell)$ then we have

$$y_{0} + \left(\sum_{i=1}^{k} \left(\left[(i-1)/2\right] - 2\left[i/2\right] + \left[(i+1)/2\right]\right)y_{i}\right) + (m-1)\left(\left[k/2\right] - \left[(k+1)/2\right]\right)$$
$$= y_{0} - y_{1} + y_{2} - y_{3} + \dots + y_{2\ell} - m + 1 \leq y_{0} - y_{1} + y_{2} - y_{3} + \dots + y_{2\ell} - y_{2\ell+1}$$
$$\leq C_{k+1}n + \frac{k+2}{2} \leq C_{k}n + k.$$

The penultimate inequality follows by using calculations from the odd case. The last inequality follows since $k \ge 2$ and $C_k = C_{k+1}$. Thus we have shown that $|S| \le C_k n + k$ and we know that $k \ge 2$. It remains to show that

$$C_k n + k \leqslant \frac{(q^2 + 1)n}{q^2 + q + 1} \tag{3.3.2}$$

for $k \ge 2$ and n sufficiently large.

We know that $m \leq n/(q^k + p\sum_{i=0}^{k-1} q^i)$ and so $n \geq q^k + p\sum_{i=0}^{k-1} q^i$, therefore condition (3.3.2) is met if

$$\left(\frac{q^2+1}{q^2+q+1} - C_k\right) \left(q^k + p\sum_{i=0}^{k-1} q^i\right) \ge k.$$
(3.3.3)

Claim 3.16. For $k \ge 6$, (3.3.3) holds.

Proof. We use induction on k. Recall that $p \ge q \ge 2$. For the base case k = 6 we directly calculate (3.3.3). First note that

$$\begin{aligned} & \frac{q^2+1}{q^2+q+1} - \frac{q^7-q^6+q^5-q^4+q^3-q^2+q-1+p(q^6+q^4+q^2+1)}{q^7+p(q^6+q^5+q^4+q^3+q^2+q+1)} \\ = & \frac{(q^6+(p-1)q^5+q^4+(p-1)q^3+q^2+1)}{(q^2+q+1)(q^7+p(q^6+q^5+q^4+q^3+q^2+q+1))}, \end{aligned}$$

and so we have

$$\left(\frac{q^2+1}{q^2+q+1} - C_6\right) \left(q^6 + p(q^5+q^4+q^3+q^2+q+1)\right)$$

$$=\frac{(q^6+(p-1)q^5+q^4+(p-1)q^3+q^2+1)(q^6+p(q^5+q^4+q^3+q^2+q+1))}{(q^2+q+1)(q^7+p(q^6+q^5+q^4+q^3+q^2+q+1))}.$$

Since $p \ge q \ge 2$ every power of q in the numerator has a coefficient of at least 1 in both expressions, hence the numerator as a single polynomial in q has positive coefficients. Hence we can make our fraction smaller by dropping lower powers of q. We then make further use of $p \ge q \ge 2$ to get the desired result:

$$\begin{split} &\frac{(q^6+(p-1)q^5+q^4+(p-1)q^3+q^2+1)(q^6+p(q^5+q^4+q^3+q^2+q+1))}{(q^2+q+1)(q^7+p(q^6+q^5+q^4+q^3+q^2+q+1))}\\ \geqslant &\frac{q^{12}+(2p-1)q^{11}+(p^2+1)q^{10}+(p^2+2p-1)q^9}{(q^2+q+1)(q^7+p(q^6+q^5+q^4+q^3+q^2+q+1))}\\ \geqslant &\frac{q^{12}+(2p-1)q^{11}+(p^2+1)q^{10}+(p^2+2p-1)q^9}{(p+1)q^{10}}\\ =&\frac{q^2+(2p-1)q+(p^2+1)}{p+1}+\frac{p^2+2p-1}{(p+1)q}\\ \geqslant &\frac{p^2+4p+3}{p+1}+\frac{p^2+p}{(p+1)q}=p+3+p/q\geqslant 6=k. \end{split}$$

For the inductive step, assume that (3.3.3) holds for k. It suffices to show that $C_k \ge C_{k+1}$ as then the result holds for k+1:

$$\left(\frac{q^2+1}{q^2+q+1} - C_{k+1}\right) \left(q^{k+1} + p\sum_{i=0}^k q^i\right) \ge \left(\frac{q^2+1}{q^2+q+1} - C_k\right) \left(q^{k+1} + p\sum_{i=0}^k q^i\right)$$
$$\ge q \left(\frac{q^2+1}{q^2+q+1} - C_k\right) \left(q^k + p\sum_{i=0}^{k-1} q^i\right) \ge qk \ge k+1.$$

For k even, we have $C_k = C_{k+1}$ by definition. For k odd, consider the following calculations:

(i)
$$D_1 := q^{k+2} \left(\sum_{i=0}^k (-1)(-q)^i \right) - q^k \left(\sum_{i=0}^{k+2} (-1)(-q)^i \right) = -q^{k+1} + q^k,$$

(ii) $D_2 := pq^{k+2} \left(\sum_{i=0}^{(k-1)/2} q^{2i} \right) - pq^k \left(\sum_{i=0}^{(k+1)/2} q^{2i} \right) = -pq^k,$

(iii)
$$D_3 := p\left(\sum_{i=0}^{k+1} q^i\right) \left(\sum_{i=0}^k (-1)(-q)^i\right) - p\left(\sum_{i=0}^{k-1} q^i\right) \left(\sum_{i=0}^{k+2} (-1)(-q)^i\right) = pq^{k+1} - pq^k,$$

(iv)
$$D_4 := p^2 \left(\sum_{i=0}^{k+1} q^i\right) \left(\sum_{i=0}^{(k-1)/2} q^{2i}\right) - p^2 \left(\sum_{i=0}^{k-1} q^i\right) \left(\sum_{i=0}^{(k+1)/2} q^{2i}\right) = p^2 q^k.$$

Using these we have

$$\begin{split} C_k - C_{k+1} &= \frac{\left(\sum_{i=0}^k (-1)(-q)^i\right) + p\left(\sum_{i=0}^{(k-1)/2} q^{2i}\right)}{q^k + p\left(\sum_{i=0}^{k-1} q^i\right)} - \frac{\left(\sum_{i=0}^{k+2} (-1)(-q)^i\right) + p\left(\sum_{i=0}^{(k+1)/2} q^{2i}\right)}{q^{k+2} + p\left(\sum_{i=0}^{k+1} q^i\right)} \\ &= \frac{D_1 + D_2 + D_3 + D_4}{\left(q^k + p\left(\sum_{i=0}^{k-1} q^i\right)\right) \left(q^{k+2} + p\left(\sum_{i=0}^{k+1} q^i\right)\right)} \\ &= \frac{(p-1)q^{k+1} + (p^2 - 2p + 1)q^k}{\left(q^k + p\left(\sum_{i=0}^{k-1} q^i\right)\right) \left(q^{k+2} + p\left(\sum_{i=0}^{k+1} q^i\right)\right)} \geqslant 0, \end{split}$$

where the last inequality follows since $p, q \ge 2$.

The claim is not a result which generally holds for $2 \le k \le 5$ so instead we directly calculate how large *n* should be to satisfy (3.3.2) in these cases. For k = 3 and k = 5 we obtain $n \ge \frac{3(q^3+p(q^2+q+1))(q^2+q+1)}{q^2+1}$ and $n \ge \frac{5(q^5+p(q^4+q^3+q^2+q+1))(q^2+q+1)}{q^4+(p-1)q^3+q^2+1}$ respectively. For

k = 2 and k = 4 we obtain weaker bounds. Hence taking n to be sufficiently large (larger than these two bounds), we have $C_k n + k \leq \frac{(q^2+1)n}{q^2+q+1}$ for all $k \geq 2$.

3.4 The number of solution-free sets

Recall Theorem 3.7 states that $f(n, \mathcal{L}^w) = 2^{\mu(n, \mathcal{L}^w) + o(n)}$ for any fixed homogeneous linear equation \mathcal{L} . The aim of this section is to replace the term o(n) here with a constant for many equations \mathcal{L} . This will be achieved in Theorem 3.18, which immediately implies Theorem 3.3. Denote by $f(n, \mathcal{L}^w, m)$ the number of \mathcal{L}^w -free subsets of [n] with minimum element m. We first give bounds on $f(n, \mathcal{L}^w, m)$ for linear equations \mathcal{L} of the form px + qy = z.

Lemma 3.17. Let \mathcal{L} denote the equation px + qy = z where $p \ge q$ and $p \ge 2$, $p, q \in \mathbb{N}$.

- (i) If $m \ge \lfloor \frac{n}{p+q} \rfloor + 1$ then $f(n, \mathcal{L}^w, m) = 2^{n-m}$.
- (ii) If n is sufficiently large depending on p and q and $m = \lfloor \frac{n}{p+q} \rfloor$ then $f(n, \mathcal{L}^w, m) \leq 2^{\mu(n, \mathcal{L}^w) 1}$.
- (iii) If n is sufficiently large depending on p and q, $q \ge 2$, $m = \lfloor \frac{n}{p+q} \rfloor t$ for some positive integer t and G_m has path parameter 1, then $f(n, \mathcal{L}^w, m) \le 2^{\mu(n, \mathcal{L}^w) 3/5 + t(3q-2p)/(5q)}$.
- (iv) If $q \ge 2$, $m = \lfloor \frac{n}{p+q} \rfloor t$ for some positive integer t and G_m has path parameter $k \ge 2$, then $f(n, \mathcal{L}^w, m) \le (4/3) \cdot 2^{(5q^2 2q + 2)n/(5q^2)}$.
- (v) If q = 1, G_m has path parameter ℓ , and $m = \lfloor \frac{n}{\ell p+1} \rfloor t$ for some integer t, then $f(n, \mathcal{L}^w, m) \leq 2^{(7\ell p+3p)n/(10\ell p+10)+(t(7-3p)+7)/10}$.

Proof. First note that (i) is trivial since all subsets $S \subseteq [n]$ with $\min(S) \ge \lfloor \frac{n}{p+q} \rfloor + 1$ are \mathcal{L}^w -free. By Fact 3.14(ii) we know that $f(n, \mathcal{L}^w, m)$ is at most the number of independent

sets in G_m which contain m. For (ii), there is one edge between $m = \lfloor \frac{n}{p+q} \rfloor$ and $(p + q)m \leq n$ in G_m , hence there are at most $2^{n-\lfloor \frac{n}{p+q} \rfloor -1} = 2^{\mu(n,\mathcal{L}^w)-1}$ independent sets in G_m containing m.

For (iii) suppose $q \ge 2$ and $m = \lfloor \frac{n}{p+q} \rfloor - t$ for some $t \in \mathbb{N}$. Notice that G_m contains a matching on $y_1 - m + 1$ edges, namely there is an edge between c and pm + qc for $c \in [m, y_1]$. Observe that $3/4 \le 2^{-2/5}$ and also

$$y_1 - m = \left\lfloor \frac{n - pm}{q} \right\rfloor - m \ge \frac{n - (p + q)m - q}{q} \ge \frac{t(p + q)}{q} - 1$$

Hence by Lemma 3.9 the total number of independent sets in G_m which contain m is at most

$$2^{n-m-2(y_1-m)-1}3^{y_1-m} \leqslant 2^{\mu(n,\mathcal{L}^w)-1+t}(3/4)^{y_1-m}$$
$$\leqslant 2^{\mu(n,\mathcal{L}^w)-1+t}(3/4)^{t(p+q)/q-1} \leqslant 2^{\mu(n,\mathcal{L}^w)-3/5+t(3q-2p)/(5q)}$$

as desired.

For (iv) suppose $q \ge 2$, $m = \lfloor \frac{n}{p+q} \rfloor - t$ for some positive integer t and G_m has path parameter $k \ge 2$. First note that

$$y_1 - y_2 = \left\lfloor \frac{n - pm}{q} \right\rfloor - \left\lfloor \frac{\lfloor \frac{n - pm}{q} \rfloor - pm}{q} \right\rfloor \geqslant \frac{n - pm - q}{q} - \frac{n - pm - qpm}{q^2}$$
$$= \frac{(q - 1)n + pm - q^2}{q^2} \geqslant \frac{(q - 1)n}{q^2} - 1.$$

Define F(i) to be the *i*th Fibonacci number where F(1) = F(2) = 1. There are F(i+2) independent sets (including the empty set) in a path of length *i*. Recall the following Fibonacci identity: $F(i+2)F(i) - F(i+1)^2 = (-1)^{i+1}$. If *i* is even and a > b

then

$$\left(\frac{F(i)F(i+2)}{F(i+1)^2}\right)^a \left(\frac{F(i+1)F(i+3)}{F(i+2)^2}\right)^b = \left(\frac{F(i+1)^2 - 1}{F(i+1)^2}\right)^a \left(\frac{F(i+2)^2 + 1}{F(i+2)^2}\right)^b \leqslant 1.$$

Also observe that by omitting $(F(i+1)F(i+3)/F(i+2)^2)^b$ the inequality still holds. By use of Fact 3.15 and applying the above bounds, we can bound from above the number of independent sets in G_m as required:

$$2^{y_0+y_2-2y_1}3^{y_1+y_3-2y_2}5^{y_2+y_4-2y_3}\dots F(k+1)^{y_{k-2}+y_k-2y_{k-1}}F(k+2)^{y_{k-1}+m-2y_k-1}F(k+3)^{y_k-m+1}$$

$$=2^{y_0+y_2-2y_1}3^{y_1-2y_2}5^{y_2}\left(\frac{3\cdot 8}{5^2}\right)^{y_3}\left(\frac{5\cdot 13}{8^2}\right)^{y_4}\dots\left(\frac{F(k+1)\cdot F(k+3)}{F(k+2)^2}\right)^{y_k}\left(\frac{F(k+2)}{F(k+3)}\right)^{m-1}$$

$$\leqslant 2^{y_0+y_2-2y_1}3^{y_1-2y_2}5^{y_2}\leqslant 2^{y_0+y_2-2y_1+y_2}3^{y_1-y_2}=2^{y_0}(3/4)^{y_1-y_2}\leqslant 2^n(3/4)^{(q-1)n/q^2-1}$$

$$\leqslant (4/3)\cdot 2^{n-2(q-1)n/(5q^2)}=(4/3)\cdot 2^{(5q^2-2q+2)n/(5q^2)}.$$

For (v), since $y_i = n - ipm$ Fact 3.15 implies that if G_m has path parameter ℓ , then G_m is a union of paths of length ℓ and $\ell + 1$. We use the bound $F(i) \leq 2^{(7i-11)/10}$ (a simple proof by induction which holds for $i \geq 2$). Since $m \leq y_\ell = n - \ell pm$ we can write $m = \lfloor \frac{n}{\ell p+1} \rfloor - t$ for some integer $t \geq 0$. Now using these bounds, we have

$$F(\ell+2)^{y_{\ell-1}-2y_{\ell}+m-1}F(\ell+3)^{y_{\ell}-m+1} = F(\ell+2)^{(\ell p+p+1)m-n-1}F(\ell+3)^{n-(\ell p+1)m+1}$$
$$\leq 2^{(3+7\ell)((\ell p+p+1)m-n-1)/10+(10+7\ell)(n-(\ell p+1)m+1)/10} = 2^{(7n+(3p-7)m+7)/10}$$
$$\leq 2^{(7+7n+(3p-7)(n/(\ell p+1)-t))/10} = 2^{(7\ell p+3p)n/(10\ell p+10)+(t(7-3p)+7)/10}.$$

Theorem 3.18. Let \mathcal{L} denote the equation px + qy = z where $p, q \in \mathbb{N}$ and

- (i) $q \ge 2$ and p > q(3q+2)/(2q-2) or;
- (ii) q = 1 and $p \ge 3$.

Then $f(n, \mathcal{L}^w) \leq (3/2 + o(1) + C)2^{\mu(n, \mathcal{L}^w)}$ where for (i) $C := \frac{2^{-2p/(5q)}}{1 - 2^{(3q-2p)/(5q)}}$ and for (ii) $C := \frac{2^{(14-3p)/10}}{1 - 2^{(7-3p)/10}}.$

Proof. For both cases by Lemma 3.17(i)–(ii) there are at most $3 \cdot 2^{\mu(n,\mathcal{L}^w)-1} \mathcal{L}^w$ -free subsets S of [n] where $\min(S) \ge \lfloor \frac{n}{p+q} \rfloor$. For (i), first consider \mathcal{L}^w -free subsets arising from Lemma 3.17(iv). Since $k \ge 2$,

$$m \leqslant y_2 = \left\lfloor \frac{\left\lfloor \frac{n-pm}{q} \right\rfloor - pm}{q} \right\rfloor \leqslant \frac{n-pm-qpm}{q^2}$$

and so $m \leqslant n/(q^2 + pq + p)$. Now as $n \to \infty$,

$$\frac{n/(q^2 + pq + p) \cdot (4/3) \cdot 2^{(5q^2 - 2q + 2)n/(5q^2)}}{2^{\mu(n,\mathcal{L}^w)}} = \frac{2^{\log_2(4n/(3(q^2 + pq + p))) + (5q^2 - 2q + 2)n/(5q^2)}}{2^{\mu(n,\mathcal{L}^w)}} \to 0,$$

as long as we have $2^{(5q^2-2q+2)n/(5q^2)} \ll 2^{\mu(n,\mathcal{L}^w)}$. This is satisfied if $(5q^2 - 2q + 2)/(5q^2) < (p+q-1)/(p+q)$ which when rearranged, gives p > q(3q+2)/(2q-2).

For \mathcal{L}^w -free subsets arising from Lemma 3.17(iii), set $a := 2^{\mu(n,\mathcal{L}^w)-3/5}$, $r := 2^{(3q-2p)/(5q)}$ and let u be the largest t such that G_m with $m = \lfloor \frac{n}{p+q} \rfloor - t$ has path parameter 1. Then since p > q(3q+2)/(2q-2) > 3q/2 we have |r| < 1 and so

$$\sum_{t=1}^{u} 2^{\mu(n,\mathcal{L}^w) - 3/5 + t(3q - 2p)/(5q)} \leqslant \sum_{t=1}^{\infty} ar^t = \sum_{t=0}^{\infty} (ar)r^t = \frac{ar}{1 - r} = \frac{2^{\mu(n,\mathcal{L}^w) - 2p/(5q)}}{1 - 2^{(3q - 2p)/(5q)}}$$

Altogether this implies that $f(n, \mathcal{L}^w) \leq (3/2 + o(1) + C)2^{\mu(n, \mathcal{L}^w)}$ where $C := \frac{2^{-2p/(5q)}}{1 - 2^{(3q-2p)/(5q)}}$.

For (ii), since $p \ge 3$, if G_m has path parameter $k \ge 2$ then we have $f(n, \mathcal{L}, m) \le 2^{17pn/(20p+10)+7/10}$. We also have $m \le y_2 = n - 2pm$ and so $m \le n/(2p+1)$. Now as

 $n \to \infty$,

$$\frac{n/(2p+1) \cdot 2^{17pn/(20p+10)+7/10}}{2^{\mu(n,\mathcal{L}^w)}} = \frac{2^{\log_2(n/(2p+1))+17pn/(20p+10)+7/10}}{2^{\mu(n,\mathcal{L}^w)}} \to 0,$$

since $2^{17pn/(20p+10)+7/10} \ll 2^{\mu(n,\mathcal{L}^w)}$.

Now consider G_m with path parameter k = 1. Set $a := 2^{pn/(p+1)+7/10}$, set $r := 2^{(7-3p)/10}$ and let u be the largest t such that G_m with $m := \lfloor \frac{n}{p+q} \rfloor - t$ has path parameter 1. Since $p \ge 3$ we have |r| < 1 and so

$$\sum_{t=1}^{u} 2^{pn/(p+1)+t(7-3p)/10+7/10} \leqslant \sum_{t=1}^{\infty} ar^{t} = \sum_{t=0}^{\infty} (ar)r^{t} = \frac{ar}{1-r}$$
$$= \frac{2^{pn/(p+1)+(14-3p)/10}}{1-2^{(7-3p)/10}} \leqslant \frac{2^{\mu(n,\mathcal{L}^{w})+(14-3p)/10}}{1-2^{(7-3p)/10}}.$$

Therefore, Lemma 3.17 implies that $f(n, \mathcal{L}^w) \leq (3/2 + o(1) + C)2^{\mu(n, \mathcal{L}^w)}$ where $C := \frac{2^{(14-3p)/10}}{1-2^{(7-3p)/10}}$.

3.5 The number of maximal solution-free sets

3.5.1 A general upper bound

Let \mathcal{L} be a three-variable linear equation. Let $\mathcal{M}(n, \mathcal{L})$ denote the set of elements $x \in [n]$ such that $x \in [n]$ does not lie in any \mathcal{L} -triple in [n]. Define $\mu^*(n, \mathcal{L}) := |\mathcal{M}(n, \mathcal{L})|$. For example, if \mathcal{L} is translation-invariant then $\{x, x, x\}$ is an \mathcal{L} -triple for all $x \in [n]$ so $\mathcal{M}(n, \mathcal{L}) = \emptyset$ and $\mu^*(n, \mathcal{L}) = 0$.

Let \mathcal{L} denote the equation px + qy = z where $p \ge 2$, $p \ge q$ and $p, q \in \mathbb{N}$. Write $t := \gcd(p,q)$. Then notice that $\mathcal{M}(n,\mathcal{L}) \supseteq \{s \in [n] : s > \lfloor (n-p)/q \rfloor, t \nmid s\}$. This follows since if $s > \lfloor (n-p)/q \rfloor$ then $ps + q \ge qs + p > n$ and so s cannot play the role of x or y in an \mathcal{L} -triple in [n]. If $t \nmid s$ then as $t \mid (px + qy)$ for any $x, y \in [n]$ we have that

s cannot play the role of z in an \mathcal{L} -triple in [n]. Actually, for large enough n we have $\mathcal{M}(n, \mathcal{L}) = \{s \in [n] : s > \lfloor (n-p)/q \rfloor, t \nmid s\}$ for all such \mathcal{L} .

We need to show that if $u \in [n]$ satisfies $u \notin \{s \in [n] : s > \lfloor (n-p)/q \rfloor, t \nmid s\}$ then u lies in an \mathcal{L} -triple. If $u \leq \lfloor (n-p)/q \rfloor$ then $p + qu \leq n$ so u lies in the \mathcal{L} -triple $\{1, u, p + qu\}$. So suppose $u > \lfloor (n-p)/q \rfloor$ and u is divisible by t. Then since n is sufficiently large, u is sufficiently large and hence can be written as u = pa + qb where a, b are positive integers. Therefore u lies in the \mathcal{L} -triple $\{u, a, b\}$.

To prove that such a, b exist, observe that first since u, p, q are all divisible by t, write $u' := u/t, r_1 := p/t$ and $r_2 := q/t$. Then $u' = r_1a + r_2b$; the largest u' such that there does not exist a, b positive integers such that $r_1a + r_2b = s$ is r_1r_2 (see [116]). So we simply require $u > \lfloor (n-p)/q \rfloor \ge t(r_1r_2+1)$.

We now prove Theorem 3.4.

Theorem 3.4. Let \mathcal{L} be a fixed homogenous three-variable linear equation. Then

$$f_{\max}(n, \mathcal{L}^w) \leqslant 3^{(\mu(n, \mathcal{L}^w) - \mu^*(n, \mathcal{L}))/3 + o(n)}$$

Proof. Let \mathcal{F} denote the set of containers obtained by applying Lemma 2.5. Since every \mathcal{L}^w -free subset of [n] lies in at least one of the $2^{o(n)}$ containers, it suffices to show that every $F \in \mathcal{F}$ houses at most $3^{(\mu(n,\mathcal{L}^w)-\mu^*(n,\mathcal{L}))/3+o(n)}$ maximal \mathcal{L}^w -free subsets.

Let $F \in \mathcal{F}$. By Lemmas 2.5(i) and 2.4, $F = A \cup B$ where |A| = o(n), $|B| \leq \mu(n, \mathcal{L}^w)$ and B is \mathcal{L}^w -free. Note that we can add all the elements of $\mathcal{M}(n, \mathcal{L})$ to B (and thus F) whilst ensuring that $|B| \leq \mu(n, \mathcal{L}^w)$ and B is \mathcal{L}^w -free. So we may assume that $\mathcal{M}(n, \mathcal{L}) \subseteq B$.

Each maximal \mathcal{L}^w -free subset of [n] in F can be found by picking a subset $S \subseteq A$ which is \mathcal{L}^w -free, and extending it in B. The number of ways of doing this is the number of ways of choosing the subset S multiplied by the number of ways of extending a fixed S in B, which we denote by N(S, B). Since |A| = o(n), there are $2^{o(n)}$ choices for S. It therefore suffices to show that for any $S \subseteq A$, we have $N(S, B) \leq 3^{(\mu(n, \mathcal{L}^w) - \mu^*(n, \mathcal{L}))/3}$.

Consider the link graph $G := L_S[B]$. Then by definition, $\mathcal{M}(n, \mathcal{L})$ is an independent set in G. Thus, $MIS(G) = MIS(G \setminus \mathcal{M}(n, \mathcal{L}))$. Further, Lemma 3.10 and Theorem 3.11(i) imply that

$$N(S,B) \leqslant \mathrm{MIS}(G) = \mathrm{MIS}(G \setminus \mathcal{M}(n,\mathcal{L})) \leqslant 3^{|B \setminus \mathcal{M}(n,\mathcal{L})|/3} \leqslant 3^{(\mu(n,\mathcal{L}^w) - \mu^*(n,\mathcal{L}))/3},$$

as desired.

As mentioned in the introduction of this chapter, Theorem 3.4 together with Theorem 3.7 shows that $f_{\max}(n, \mathcal{L}^w)$ is significantly smaller than $f(n, \mathcal{L}^w)$ for all homogeneous threevariable linear equations \mathcal{L} that are not translation-invariant. So in this sense it can be viewed as a generalisation of a result of Luczak and Schoen [79] on sum-free sets.

Let \mathcal{L} denote the equation px + y = z for some $p \in \mathbb{N}$. Notice that in this case we have $\mu^*(n, \mathcal{L}) = 0$ for n > p. The next result implies that if p is large then $f_{\max}(n, \mathcal{L}^w)$ is close to the bound in Theorem 3.4. So for such equations \mathcal{L} , Theorem 3.4 is close to best possible.

Proposition 3.19. Given $p \in \mathbb{N}$ where $p \ge 2$, let \mathcal{L} denote the equation px + y = z and let n be sufficiently large depending on p. Then

$$f_{\max}(n, \mathcal{L}^w) \ge 3^{\mu(n, \mathcal{L}^w)/3 - 2pn/(3(p+1)(3p^2 - 1)) - p - 5}.$$

Proof. Given $p, n \in \mathbb{N}$, let \mathcal{L} denote the equation px + y = z. Set $s := \lfloor \frac{(p-1)n}{3p^2-1} \rfloor$ and $a := \lfloor \frac{n-s}{p} \rfloor$. Consider the link graph $G := L_{\{s,2s\}}[a+1, a+3ps]$. Observe that:

$$2s \leqslant \frac{(2p-2)n}{3p^2-1} < \frac{n}{p+1} < \frac{(3p-1)n}{3p^2-1} = \frac{n}{p} - \frac{(p-1)n}{3p^3-p} \leqslant \frac{n-s}{p} < a+1;$$
$$a+3ps = \left\lfloor \frac{n-s}{p} \right\rfloor + 3ps \leqslant \frac{n}{p} + \left(3p - \frac{1}{p}\right)s = \frac{n}{p} + \frac{3p^2-1}{p} \left\lfloor \frac{(p-1)n}{3p^2-1} \right\rfloor$$

$$\leqslant \frac{n+n(p-1)}{p} = n.$$

As a consequence, the sets $\{s, 2s\}$ and [a + 1, a + 3ps] (a subset of $[\lfloor \frac{n}{p+1} \rfloor + 1, n]$) are disjoint \mathcal{L}^w -free sets in [n], and so Lemma 3.13 implies that $f_{\max}(n, \mathcal{L}^w) \ge \mathrm{MIS}(G)$. It remains to show that G contains at least $3^{\mu(n, \mathcal{L}^w)/3 - 2pn/(3(p+1)(3p^2-1))-6}$ maximal independent sets.

Observe that for each $i \in [ps]$ there is an edge in G between a + i and a + ps + i (since $\{s, a + i, a + i + ps\}$ is an \mathcal{L} -triple), an edge between a + i + ps and a + i + 2ps (since $\{s, a + i + ps, a + i + 2ps\}$ is an \mathcal{L} -triple) and an edge between a + i and a + i + 2ps (since $\{2s, a + i, a + i + 2ps\}$ is an \mathcal{L} -triple). Also since a > (n - s)/p - 1, we have p(a + 1) + s > n and hence there are no further edges in G.

Hence G is a collection of ps disjoint triangles, where 4 vertices in G have loops ((p+1)s, (p+2)s, (2p+1)s and (2p+2)s). So G has at least 3^{ps-4} maximal independent sets. Now observe:

$$ps - 4 - \frac{\mu(n, \mathcal{L}^w)}{3} = p \left\lfloor \frac{(p-1)n}{3p^2 - 1} \right\rfloor - 4 - \frac{n}{3} + \frac{1}{3} \left\lfloor \frac{n}{p+1} \right\rfloor$$
$$\geqslant \left(\frac{p^2 - p}{3p^2 - 1} - \frac{1}{3} + \frac{1}{3(p+1)} \right) n - p - 5$$
$$= \left(\frac{-2p}{3(p+1)(3p^2 - 1)} \right) n - p - 5,$$

as required.

3.5.2 Upper bounds for px + qy = z

Let \mathcal{L} denote the equation px + qy = z where $p \ge q$, $p \ge 2$ and $p, q \in \mathbb{N}$. For such \mathcal{L} , the next simple result provides an alternative bound to Theorem 3.4.

Lemma 3.20. Let \mathcal{L} denote the equation px + qy = z where $p \ge q$, $p \ge 2$ and $p, q \in \mathbb{N}$.

Then $f_{\max}(n, \mathcal{L}^w) \leq f(\lfloor (n-p)/q \rfloor, \mathcal{L}^w).$

Proof. Set $C := \lfloor \lfloor \frac{n-p}{q} \rfloor$ and $B := \lfloor \lfloor \frac{n-p}{q} \rfloor + 1, n \rfloor$. In particular, B is \mathcal{L}^w -free. Notice that every maximal \mathcal{L}^w -free subset of [n] can be found by selecting an \mathcal{L}^w -free subset $S \subseteq C$ and then extending it in B to a maximal one. Suppose we have such an \mathcal{L}^w -free subset S. By Lemma 3.10, the number of such extensions of S is at most $MIS(L_S[B])$.

For any \mathcal{L} -triple $\{x, y, z\}$ in [n] satisfying px + qy = z, since $z \leq n$, we must have $x \leq \frac{n-q}{p}$ and $y \leq \frac{n-p}{q}$. Hence $x, y \in C$. This means that there are no \mathcal{L} -triples in [n] which contain more than one element from B. Thus the link graph $L_S[B]$ must only contain isolated vertices and loops. So $L_S[B]$ has precisely one maximal independent set. Hence the number of maximal \mathcal{L}^w -free subsets of [n] is bounded by the number of choices of S in C which are \mathcal{L}^w -free, i.e. $f(\lfloor (n-p)/q \rfloor, \mathcal{L}^w)$.

Lemma 3.20 together with Theorems 3.3 and 3.7 immediately implies Theorem 3.6.

The next result gives a further upper bound on $f_{\max}(n, \mathcal{L}^w)$ for certain linear equations \mathcal{L} . Notice that for such \mathcal{L} , Theorem 3.5 yields a better bound than Theorem 3.4.

Theorem 3.5. Let \mathcal{L} denote the equation px + qy = z where $p \ge q \ge 2$ are integers so that $p \le q^2 - q$ and gcd(p,q) = q. Then

$$f_{\max}(n, \mathcal{L}^w) \leqslant 2^{(\mu(n, \mathcal{L}^w) - \mu^*(n, \mathcal{L}))/2 + o(n)}.$$

Proof. Let \mathcal{F} denote the set of containers obtained by applying Lemma 2.5. Since every \mathcal{L}^w -free subset of [n] lies in at least one of the $2^{o(n)}$ containers, it suffices to show that every $F \in \mathcal{F}$ houses at most $2^{(\mu(n,\mathcal{L}^w)-\mu^*(n,\mathcal{L}))/2+o(n)} \mathcal{L}^w$ -free sets.

Let $F \in \mathcal{F}$. By Lemmas 2.5(i) and 2.4, $F = A \cup B$ where |A| = o(n), $|B| \leq \mu(n, \mathcal{L}^w)$ and B is \mathcal{L}^w -free. Note that we can add all the elements of $\mathcal{M}(n, \mathcal{L})$ to B (and thus F) whilst ensuring that $|B| \leq \mu(n, \mathcal{L}^w)$ and B is \mathcal{L}^w -free. So we may assume that $\mathcal{M}(n, \mathcal{L}) \subseteq B$. Either we have $\min(B) > \lfloor \frac{n}{p+q} \rfloor$ or we use Theorem 3.1 to say that either $\min(B) = \lfloor \frac{n}{p+q} \rfloor - t$ for some non-negative integer $t < (\frac{p+q-1}{p+q+p/q}) \lfloor \frac{n}{p+q} \rfloor$ and $|B| \leq \lfloor \frac{(p+q-1)n}{p+q} \rceil - \lfloor \frac{p}{q}t \rfloor$, or $|B| \leq \frac{(q^2+1)n}{q^2+q+1}$.

Case 1: $\min(B) = \lfloor \frac{n}{p+q} \rfloor - t$ for $0 \leq t < (\frac{p+q-1}{p+q+p/q}) \lfloor \frac{n}{p+q} \rfloor$, or $\min(B) > \lfloor \frac{n}{p+q} \rfloor$ (in which case set t := 0). Write $F = X \cup Y$ where $Y \subseteq \lfloor \lfloor \frac{n}{p+q} \rfloor + 1, n$] is \mathcal{L}^w -free, and $X \subseteq [1, \lfloor \frac{n}{p+q} \rfloor]$. Note that |X| = t' + o(n) and $|Y| \leq \lceil \frac{(p+q-1)n}{p+q} \rceil - \lfloor \frac{p}{q}t \rfloor - t' + o(n)$ where $0 \leq t' \leq t$. Also $\mathcal{M}(n, \mathcal{L}) \subseteq Y$. Choose $S \subseteq X$ to be \mathcal{L}^w -free. Consider the link graph $L_S[Y]$ and observe that by Lemma 3.10, $N(S, Y) \leq \mathrm{MIS}(L_S[Y])$. (Recall N(S, Y) denotes the number of extensions of S in Y to a maximal \mathcal{L}^w -free set.)

Since $p \leq q^2 - q$, by Lemma 3.12 $L_S[Y]$ is triangle-free. By definition, $\mathcal{M}(n, \mathcal{L})$ is an independent set in $L_S[Y]$ and so $\mathrm{MIS}(L_S[Y]) = \mathrm{MIS}(L_S[Y \setminus \mathcal{M}(n, \mathcal{L})])$. Therefore Theorem 3.11(ii) implies that $\mathrm{MIS}(L_S[Y]) \leq 2^{(|Y| - |\mathcal{M}(n, \mathcal{L})|)/2}$. Overall, this implies that the number of \mathcal{L}^w -free sets contained in F is at most

$$2^{|X|} \times 2^{(|Y| - |\mathcal{M}(n,\mathcal{L})|)/2} \leqslant 2^{t' + o(n) + (\mu(n,\mathcal{L}^w) - \mu^*(n,\mathcal{L}) - \lfloor \frac{p}{q}t \rfloor - t')/2} \leqslant 2^{(\mu(n,\mathcal{L}^w) - \mu^*(n,\mathcal{L}))/2 + o(n)},$$

as desired.

Case 2: $|B| \leq \frac{(q^2+1)n}{q^2+q+1}$. In this case $|F| \leq \frac{(q^2+1)n}{q^2+q+1} + o(n)$. Choose any \mathcal{L}^w -free $S \subseteq A$ (note there are at most $2^{o(n)}$ choices for S). Consider the link graph $L_S[B]$ and observe by Lemma 3.10 that $N(S, B) \leq \text{MIS}(L_S[B])$. Similarly as in Case 1 we have that $\text{MIS}(L_S[B])=\text{MIS}(L_S[B'])$ where $B' := B \setminus \mathcal{M}(n, \mathcal{L})$. By Theorem 3.11(i),

$$\mathrm{MIS}(L_S[B']) \leqslant 3^{|B'|/3} \leqslant 3^{((q^2+1)n/(3(q^2+q+1))-\mu^*(n,\mathcal{L})/3)} \leqslant 2^{(\mu(n,\mathcal{L}^w)-\mu^*(n,\mathcal{L}))/2+o(n)}.$$

The last inequality follows since $\mu(n, \mathcal{L}^w) = n - \lfloor n/(p+q) \rfloor$ and $\mathcal{M}(n, \mathcal{L}) = \{s : s > \lfloor (n-p)/q \rfloor, q \nmid s\}$ since gcd(p,q) = q.

To see this, first note that

$$\mu^*(n, \mathcal{L}) = \frac{(q-1)^2 n}{q^2} - o(n).$$

Hence for the inequality to hold we require that

$$9^{((q^2+1)/(q^2+q+1)-(q^2-2q+1)/(q^2))} < 8^{((p+q-1)/(p+q)-(q^2-2q+1)/(q^2))}.$$

Let $a := \log_9 8$. This rearranges to give

$$p > \frac{(1-a)(q^4-q) + q^3 + q^2}{(2a-1)q^3 + (a-1)(q^2+q-1)}.$$

Since $p \ge q$ it suffices to show that $(3a-2)q^3 + (a-2)(q^2+q) + (2-2a) > 0$. This indeed holds since $q \ge 2$.

Overall, this implies that the number of \mathcal{L}^{w} -free sets contained in F is at most $2^{(\mu(n,\mathcal{L}^{w})-\mu^{*}(n,\mathcal{L}))/2+o(n)}$, as desired.

The proof of Theorem 3.5 actually generalises to some other equations px + qy = zwhere $gcd(p,q) \neq q$ (but still $p \leq q^2 - q$). However, in these cases Theorem 3.6 produces a better upper bound on $f_{max}(n, \mathcal{L}^w)$.

3.5.3 Lower bounds for px + qy = z

The following result provides lower bounds on $f_{\max}(n, \mathcal{L}^w)$ for all equations \mathcal{L} of the form px + qy = z where $p \ge q \ge 2$.

Proposition 3.21. Let \mathcal{L} denote the equation px + qy = z where $p \ge q \ge 2$ are integers. Suppose that n > 2p. In each case $f_{\max}(n, \mathcal{L}^w) \ge 2^{\ell}$ where ℓ is defined as follows:

(i)
$$\ell := (n(q-1) - pq + q - 2q^2)/q^2$$
 if $p \ge q^2$,

(ii)
$$\ell := (n(p-q) - p^2 + q^2 - 2pq)/(pq)$$
 if $q ,$

(*iii*) $\ell := (n - 6q)/2q$ if p = q.

Proof. For each case, we shall let $B := \lfloor \lfloor \frac{n}{p+q} \rfloor + 1, n \rfloor$, and consider the link graph $G := L_{\{1\}}[B]$. Since B and $\{1\}$ are \mathcal{L}^w -free, by Lemma 3.13 it suffices to show that there is an induced subgraph of G which contains at least 2^{ℓ} maximal independent sets. For each case we will find an induced perfect matching on 2ℓ vertices in G. (Note there are 2^{ℓ} maximal independent sets in such a matching.)

More specifically, for each case we shall find an interval I := [a, b] for some $a, b \in V(G)$ and let $J := \{qi + p | i \in I\}$. Note that all edges in G (other than at most one loop) are of the form $\{i, qi + p\}$ and $\{i, pi + q\}$. By our choice of I and J, $G[I \cup J]$ will form a perfect matching on 2|I| vertices if the following conditions hold:

- (1) qa + p > b (which ensures that $I \cap J = \emptyset$),
- (2) $qb + p \leq n$ (which ensures that $J \subseteq [n]$),
- (3) pa+q > n (which ensures that the only edges in $G[I \cup J]$ are of the form $\{i, qi+p\}$),
- (4) p + q < a (which ensures that there is no loop at a vertex in $G[I \cup J]$).

Notice that actually we do not require condition (3) to hold in the case when p = q. Indeed, this is because in this case an edge $\{i, pi + q\}$ in G is the same as the edge $\{i, qi + p\}$. Further, there is at most one loop in G (if $p + q \in B$). So even if (4) does not hold we will obtain an induced matching in G on 2|I| - 2 vertices.

Thus, to obtain an induced matching in G on 2|I| - 2 vertices it suffices to choose a and b so that (1)–(3) hold except when p = q when we only require that (1) and (2) hold.

By choosing $b := \lfloor (n-p)/q \rfloor$, (2) holds since $qb+p = q \lfloor (n-p)/q \rfloor + p \leq q(n-p)/q + p = n$.

If $p \ge q^2$ then set $a := \lfloor (n-q)/q^2 \rfloor + 1$. Then $a \in B$ and further $pa + q \ge q^2a + q > q^2((n-q)/q^2) + q = n$ and $qa + p \ge qa + q^2 > q((n-q)/q^2) + q^2 = n/q - 1 + q^2 > \lfloor (n-p)/q \rfloor = b$. So (1) and (3) hold.

If $q then set <math>a := \lfloor (n-q)/p \rfloor + 1$. So $a \in B$. Further, pa+q > p((n-q)/p)+q = n and $qa+p > q((n-q)/p)+p = qn/p-q^2/p+p > qn/q^2-q+p > n/q > \lfloor (n-p)/q \rfloor = b$. So (1) and (3) hold.

If p = q set $a := \lfloor n/(p+q) \rfloor + 1 = \lfloor n/(2q) \rfloor + 1 \in B$. Observe that $qa+q > qn/2q+q > n/2 > \lfloor (n-q)/q \rfloor = b$ since $q \ge 2$. So (1) holds.

Now calculating the size of the interval I = [a, b] in each case proves the result:

- If $a = \lfloor (n-q)/q^2 \rfloor + 1$, then $|I| 1 = \lfloor (n-p)/q \rfloor (\lfloor (n-q)/q^2 \rfloor + 1) \ge (n-p)/q 1 (n-q)/q^2 1 = (n(q-1) pq + q 2q^2)/q^2$.
- If $a = \lfloor (n-q)/p \rfloor + 1$, then $|I| 1 = \lfloor (n-p)/q \rfloor (\lfloor (n-q)/p \rfloor + 1) \ge (n-p)/q 1 (n-q)/p 1 = (n(p-q) p^2 + q^2 2pq)/(pq).$
- If $a = \lfloor n/(p+q) \rfloor + 1$ and p = q, then $|I| 1 = \lfloor (n-p)/q \rfloor (\lfloor n/(p+q) \rfloor + 1) \ge (n-p)/q 1 n/(p+q) 1 = (pn (p+2q)(p+q))/(q(p+q)) = (qn 6q^2)/(2q^2) = (n 6q)/2q.$

- 11			_
	L		

Although the lower bounds in Proposition 3.21 do not meet the upper bounds in Theorems 3.5 and 3.6 in general, Theorem 3.5 and Proposition 3.21(iii) do immediately imply the following, where we determine $\log(f_{\max}(n, \mathcal{L}^w))$ asymptotically.

Theorem 3.22. Let \mathcal{L} denote the equation 2x + 2y = z. Then $f_{\max}(n, \mathcal{L}^w) = 2^{n/4 + o(n)}$.

In the next chapter, we give a general upper bound on $f_{\max}(n, \mathcal{L}^w)$ for equations \mathcal{L} of the form px + qy = rz where $p \ge q \ge r$ are fixed positive integers (see Theorem 4.5). In particular, our result shows that in the case when $p = q \ge 2$, r = 1 the lower bound in Proposition 3.21(iii) is correct up to an o(n) term in the exponent.

3.6 Concluding remarks

The results in the chapter show that the parameter $f_{\max}(n, \mathcal{L}^w)$ can exhibit very different behaviour depending on the linear equation \mathcal{L} . Indeed, Theorem 3.4 gives a 'crude' general upper bound on $f_{\max}(n, \mathcal{L}^w)$ for all homogeneous three-variable linear equations \mathcal{L} . (It is crude in the sense that, in the proof, we do not use any structural information about the link graphs.) However, this bound is close to the correct value of $f_{\max}(n, \mathcal{L}^w)$ for certain equations \mathcal{L} (Proposition 3.19). On the other hand, for many equations this bound is far from tight (Theorem 3.5). Further, for some equations (x + y = z and 2x + 2y = z)the value of $f_{\max}(n, \mathcal{L}^w)$ is tied to the property that any triangle-free graph on n vertices contains at most $2^{n/2}$ maximal independent sets. Theorem 3.6 and upper bounds we obtain in the next chapter suggest though that the value of $f_{\max}(n, \mathcal{L}^w)$ for other equations \mathcal{L} may depend on completely different factors. Further progress on understanding the possible behaviour of $f_{\max}(n, \mathcal{L}^w)$ would be extremely interesting.

CHAPTER 4

More on solution-free sets of

INTEGERS

4.1 Introduction

In the last chapter we obtained results concerning \mathcal{L}^w -free subsets of [n] where \mathcal{L} is a fixed linear equation. We made progress on all three of our general motivating questions:

- (i) What is the size of the largest \mathcal{L}^w -free subset of [n]?
- (ii) How many \mathcal{L}^w -free subsets of [n] are there?
- (iii) How many maximal \mathcal{L}^w -free subsets of [n] are there?

Recall that we denote the answers to the above questions by $\mu(n, \mathcal{L}^w)$, $f(n, \mathcal{L}^w)$ and $f_{\max}(n, \mathcal{L}^w)$ respectively. In this chapter we use a new lemma to obtain further results for the above three questions. In particular we obtain results for equations of the form px + qy = rz where $p \ge q \ge r$ and $p, q, r \in \mathbb{N}$. In Chapter 3 we used various auxiliary graphs as a means to bound the size and number of elements in a solution-free subset of [n]. In Section 4.2 we introduce a new auxiliary graph which similarly can be used for this purpose.

We also extend our results to equations in more than three variables. The following simple but key proposition allows us to easily do this.

Proposition 4.1. Let \mathcal{L}_1 denote the equation $p_1x_1 + \cdots + p_kx_k = b$ where $p_1, \ldots, p_k, b \in \mathbb{Z}$ and let \mathcal{L}_2 denote the equation $(p_1 + p_2)x_1 + p_3x_2 + \cdots + p_kx_{k-1} = b$. Then $\mu(n, \mathcal{L}_1^w) \leq \mu(n, \mathcal{L}_2^w)$ and $f(n, \mathcal{L}_1^w) \leq f(n, \mathcal{L}_2^w)$.

The proposition is just a simple consequence of the observation that any solution to the equation \mathcal{L}_2 gives rise to a solution to the equation \mathcal{L}_1 . So all \mathcal{L}_1^w -free subsets of [n] are also \mathcal{L}_2^w -free.

4.1.1 The size of the largest solution-free set

The first key question in the study of \mathcal{L}^w -free sets listed in the introduction, is to establish the size $\mu(n, \mathcal{L}^w)$ of the largest \mathcal{L}^w -free subset of [n]. When \mathcal{L} is a homogeneous equation in two variables, the value of $\mu(n, \mathcal{L}^w)$ is known exactly and an extremal \mathcal{L}^w -free set can be found by greedy choice. See [53] for further details. For homogeneous linear equations in three variables, the picture is not as clear. First note we may assume without loss of generality that the equation is of the form px + qy = rz, where p, q, r are fixed positive integers, and gcd(p, q, r) = 1.

Now consider the following two natural candidates for extremal sets. Let t := gcd(p, q)and let a be the unique non-negative integer $0 \leq a < t$ such that n - a is divisible by t. The interval

$$I_n := \left[\left\lfloor \frac{r(n-a)}{p+q} \right\rfloor + 1, n \right]$$

is \mathcal{L}^w -free. To see this observe that since gcd(p,q,r) = 1 and gcd(p,q) = t, any solution (x, y, z) to \mathcal{L} with $x, y, z \in I_n$ must have z divisible by t. Since px + qy > r(n-a), z must lie in [n - a + 1, n]; however then z is not divisible by t and so I_n is \mathcal{L}^w -free. Note that when r = 1, $I_n := \lfloor \frac{r(n-a)}{p+q} \rfloor + 1$, $n = \lfloor \frac{rn}{p+q} \rfloor + 1$, n

hold in general when r > 1. Notice also that I_n is only a candidate for an extremal set if r is 'small'. Indeed, if r > p + q and n is sufficiently large then $I_n = \emptyset$. The set

$$T_n := \{ x \in [n] : x \not\equiv 0 \mod t \}$$

is also \mathcal{L}^w -free: note that in any solution (x, y, z) to \mathcal{L} , z must be divisible by t since gcd(r, t) = 1. But T_n contains no elements divisible by t.

This raises the following question.

Question 4.2. For which \mathcal{L} do we have $\mu(n, \mathcal{L}^w) = \max\{|I_n|, |T_n|\}$?

When \mathcal{L} is x + y = z it is easy to see that $\mu(n, \mathcal{L}^w) = \lceil n/2 \rceil$ and the interval $I_n = \lfloor \lfloor n/2 \rfloor + 1, n \rfloor$ is an extremal set of this size. In the previous chapter we established that when \mathcal{L} is the equation px+qy = z with $p, q \in \mathbb{N}, p \ge 2$, we have $\mu(n, \mathcal{L}^w) = n - \lfloor n/(p+q) \rfloor$ for sufficiently large n. Again this bound is attained by the interval I_n . Our first result of this chapter determines a further class of equations (of the form px + qy = rz) for which I_n or T_n gives an \mathcal{L} -free set of maximum size.

Theorem 4.3. Let \mathcal{L} denote the equation px + qy = rz where $p \ge q \ge r$ and p, q, r are fixed positive integers satisfying gcd(p, q, r) = 1. Let t := gcd(p, q). Write $r_1 := p/t$ and $r_2 := q/t$.

- (i) If q divides p and $p + q \leq rq$ then $\mu(n, \mathcal{L}^w) = |T_n| = \lceil (q-1)n/q \rceil$;
- (ii) If q divides p and $p+q \ge rq$ then $\mu(n, \mathcal{L}^w) = |I_n| = \lceil (p+q-r)(n-a)/(p+q)\rceil + a$ where a is the unique non-negative integer $0 \le a < q$ such that n-a is divisible by q;
- (iii) If q does not divide p, t > 1 and

$$r > (r_1 r_2 - r_1 - r_2 + 4) r_2 \left(r_1 + 1 + \frac{r_2 - 1}{r_1^2 + (r_1 - 1)(r_2 - 1)} \right)$$

then
$$\mu(n, \mathcal{L}^w) = |T_n| = \lceil (t-1)n/t \rceil$$
.

Theorem 4.3(ii) was already proven (for large enough n) in the last chapter (see Corollary 3.2) in the special case when r = 1. (Note though that Corollary 3.2 determines $\mu(n, \mathcal{L}^w)$ for many equations \mathcal{L} not covered by Theorem 4.3.) Previously, Hegarty [53] proved Theorem 4.3(i) in the case when p = q. In Section 4.3 we also give a generalisation of Theorem 4.3 concerning some linear equations with more variables (see Corollary 4.14).

Notice that in the case when q divides p, Theorem 4.3 gives a dichotomy for the value of $\mu(n, \mathcal{L}^w)$: when $p + q \leq rq$ the set T_n is a largest \mathcal{L}^w -free subset of [n], whilst when $p + q \geq rq$ the interval I_n is a largest \mathcal{L}^w -free subset of [n]. Theorem 4.3 does not provide us with as much information for the case when q does not divide p; note though it is not true that a similar dichotomy occurs in this case. Take the equation 3x + 2y = 2z; here we have $|I_n| \approx 3n/5$ and $|T_n| = 0$. However the set $A_n := \{x \in [n] : x \not\equiv 0 \mod 2 \text{ or} x > 2n/3\}$ has size $|A_n| \approx 2n/3$ and is \mathcal{L}^w -free, since any solution (x, y, z) to \mathcal{L} must have that x is even and $x \leq 2n/3$. It would be very interesting to fully resolve the case where $p \geq q \geq r$ and q does not divide p.

For equations px + qy = rz where r is bigger than p, q, there are a range of cases where an extremal set is known and it is neither I_n nor T_n ; see [11, 31, 53] for these, and also other results on the size of the largest \mathcal{L}^w -free subset of [n] for various \mathcal{L} .

4.1.2 The number of solution-free sets

Recall Green [48] proved that $f(n, \mathcal{L}^w) = 2^{\mu(n, \mathcal{L}^w) + o(n)}$ for any fixed homogeneous linear equation \mathcal{L} . In the last chapter we were able to replace the term o(n) in the exponent with a constant for certain types of linear equation \mathcal{L} . In Section 4.4 we find further linear equations where we can omit the term o(n):

Theorem 4.4. Let \mathcal{L} denote the equation px + qy = rz where $p \ge q \ge r$, p,q,r are fixed positive integers satisfying gcd(p,q,r) = 1. Let t := gcd(p,q). Write $r_1 := p/t$ and $r_2 := q/t$. If

$$r > (r_1 r_2 - r_1 - r_2 + 4) r_2 \left(r_1 + 1 + \frac{r_2 - 1}{r_1^2 + (r_1 - 1)(r_2 - 1)} \right)$$

then $f(n, \mathcal{L}^w) = \Theta(2^{\mu(n, \mathcal{L}^w)}).$

By applying Proposition 4.1 we also obtain equations \mathcal{L} with more than three variables for which $f(n, \mathcal{L}^w) = \Theta(2^{\mu(n, \mathcal{L}^w)})$.

4.1.3 The number of maximal solution-free sets

In this chapter we prove the following result.

Theorem 4.5. Let \mathcal{L} denote the equation px + qy = rz where $p \ge q \ge r$ and p, q, r are fixed positive integers satisfying gcd(p,q,r) = 1. Let t := gcd(p,q). Then

$$f_{\max}(n, \mathcal{L}^w) \leq 2^{\frac{Crn}{q} + o(n)} \text{ where } C := 1 - \frac{t}{p+q} \Big(\frac{p^2 + (p-t)(q-t)}{p^2} \Big).$$

For a wide class of equations \mathcal{L} this is the current best known upper bound on $f_{\max}(n, \mathcal{L}^w)$; see Section 4.6 for more details. In the case when $p = q \ge 2$ and r = 1, the upper bound given by Theorem 4.5 is actually exact up to the error term in the exponent.

Theorem 4.6. Let \mathcal{L} denote the equation qx + qy = z where $q \ge 2$ is an integer. Then

$$f_{\max}(n, \mathcal{L}^w) = 2^{n/2q + o(n)}.$$

In Section 4.5 we will also generalise Theorem 4.5 to consider some linear equations with more variables (see Theorem 4.18).

For the proof of both Theorems 4.3 and 4.5, a simple but crucial tool is a result (Lemma 4.9) which ensures a certain auxiliary graph contains a large collection of disjoint

edges. As in the previous chapter, we also make use of container and removal lemmas of Green [48].

In the next section we collect together a number of useful tools and lemmas. We prove our results on the size of the largest solution-free subset of [n], on the number of solution-free subsets of [n], and on the number of maximal solution-free subsets of [n], in Sections 4.3, 4.4 and 4.5 respectively.

4.2 Link hypergraphs and the main lemmas

4.2.1 Link hypergraphs

One can turn the problem of counting the number of maximal \mathcal{L}^{w} -free subsets of [n] into one of counting maximal independent sets in an auxiliary graph. Similar techniques were used in [121, 6, 7], and in the graph setting in [10, 5]. Recall that in the previous chapter we defined the link graph, which we used to deal with equations with three variables. Since we will consider equations with more than three variables in this chapter, we need to generalise this definition.

Consider the following generalisation of a link graph $L_S[B]$ to that of a link hypergraph: Let B and S be disjoint subsets of [n] and let \mathcal{L} denote the equation $p_1x_1 + \cdots + p_kx_k = 0$ where p_1, \ldots, p_k are fixed non-zero integers. The link hypergraph $L_S[B]$ of S on B has vertex set B; It has an edge set consisting of hyperedges between $s \leq k$ distinct vertices v_1, \ldots, v_s of B, whenever there is a solution (x_1, \ldots, x_k) to \mathcal{L} in which $\{x_1, \ldots, x_k\} \subseteq$ $S \cup \{v_1, \ldots, v_s\}$ and $\{v_1, \ldots, v_s\} \subseteq \{x_1, \ldots, x_k\}$. In this definition one could have edges corresponding to trivial solutions. However in our applications, since we only consider non-translation-invariant equations, there are no trivial solutions.

The link graph lemmas used in the previous chapter (Lemmas 3.10 and 3.13) can easily be extended to the hypergraph case. **Lemma 4.7.** Let \mathcal{L} denote a non-translation-invariant linear equation. Suppose that B, S are disjoint \mathcal{L}^w -free subsets of [n]. If $I \subseteq B$ is such that $S \cup I$ is a maximal \mathcal{L}^w -free subset of [n], then I is a maximal independent set in the link hypergraph $L_S[B]$.

As with Lemma 3.10 in the previous chapter, the above result can be used in conjunction with the container lemma as follows. Let $F = A \cup B$ be a container as in Lemma 2.5 where |A| = o(n) and B is \mathcal{L}^w -free. Observe that any maximal \mathcal{L}^w -free subset of [n] in F can be found by first selecting an \mathcal{L}^w -free subset $S \subseteq A$, and then extending S in B. Then the number of extensions of S in B is bounded by $MIS(L_S[B])$ by Lemma 4.7.

We can also use link graphs to obtain lower bounds.

Lemma 4.8. Let \mathcal{L} denote a non-translation-invariant linear equation. Suppose that B, S are disjoint \mathcal{L}^w -free subsets of [n]. Let H be an induced subgraph of the link graph $L_S[B]$. Then $f_{\max}(n, \mathcal{L}^w) \ge \mathrm{MIS}(H)$.

The proof is analogous to that of Lemma 3.13.

4.2.2 The main lemmas

Here we use a specific link graph as a means to bound the number of elements in a solution-free subset of [n].

Let \mathcal{L} denote the equation px + qy = rz where $p \ge q \ge r$ and p, q, r are fixed positive integers satisfying gcd(p,q,r) = 1. Let t := gcd(p,q) and write $r_1 := p/t, r_2 := q/t$. Fix $M \in [n]$ such that M is divisible by t. We define the graph G_M to have vertex set $[\lceil rM/q \rceil - 1]$ and an edge between x and y whenever px + qy = rM.

Lemma 4.9. The graph G_M contains a collection E of vertex-disjoint edges where

$$|E| = \left\lfloor \frac{rM}{r_2(p+q)} \right\rfloor + (r_1r_2 - r_1 - r_2 + 1) \left\lfloor \left\lfloor \frac{rM}{r_1(p+q)} - \frac{1}{r_2} \right\rfloor \frac{1}{r_1r_2} \right\rfloor$$

and at most one edge in E is a loop.

Proof. All edges in G_M are pairs of the form $\{s, (rM - sp)/q\}$ for some $s \in \mathbb{N}$ since ps + q(rM - sp)/q = rM. Since $p = r_1t$ and $q = r_2t$ where r_1 and r_2 are coprime, for a fixed integer s precisely one element in $\{(rM - (s - j)p)/q : 0 \leq j < r_2\}$ is an integer. (Note here we are using that M is divisible by t.) In other words there exists a unique $x \in \mathbb{N}, 1 \leq x \leq r_2$ such that (rM - xp)/q is an integer, and all edges in G_M are of the form $\{x + ar_2, (rM - xp)/q - ar_1\}$ for some non-negative integer a. In particular there is an edge incident to $x + ar_2$ provided a satisfies $(rM - xp)/q - ar_1 \in \mathbb{N}$.

Write y := (rM - xp)/q. Note that if $x + ar_2 \leq rM/(p+q)$ then

$$y - ar_1 = \frac{rM - (x + ar_2)p}{q} \ge \frac{rM - prM/(p+q)}{q} = \frac{rM}{p+q}$$

Hence there are $\lfloor rM/(r_2(p+q)) \rfloor$ distinct edges in G_M of the form $\{x + ar_2, y - ar_1\}$ with $x + ar_2 \leq rM/(p+q) \leq y - ar_1$. Note that one of these edges may be a loop. (This will be at rM/(p+q) in the case when $rM/(p+q) \in \mathbb{N}$.) Call this collection of edges E_1 . Our next aim is to find an additional collection E_2 of edges in G_M that is vertex-disjoint from E_1 .

Note that $x + ar_2 \equiv x \mod r_2$ and $y - ar_1 \equiv y \mod r_1$. Also p(rM/p) + q(0) = rMand $\lceil rM/p \rceil \leq \lceil rM/q \rceil$, hence there are at least

$$\left\lfloor \left(\left\lceil \frac{rM}{p} \right\rceil - 1 - \left\lfloor \frac{rM}{p+q} \right\rfloor \right) / r_2 \right\rfloor \geqslant \left\lfloor \left(\frac{rM}{p} - \frac{rM}{p+q} - 1 \right) / r_2 \right\rfloor = \left\lfloor \frac{rM}{r_1(p+q)} - \frac{1}{r_2} \right\rfloor$$

edges in G_M of the form $\{x + ar_2, y - ar_1\}$ with $x + ar_2 > rM/(p+q)$. Consider a set of r_1r_2 edges $\{\{x + ar_2 + br_2, y - ar_1 - br_1\} : 0 \leq b < r_1r_2\}$ for a fixed a. Since r_1 and r_2 are coprime, precisely r_2 of these edges (1 in r_1 of them) have $x + ar_2 + br_2 \equiv y \mod r_1$, and precisely r_1 of these edges have $y - ar_1 - br_1 \equiv x \mod r_2$. (Also, precisely 1 edge satisfies both.) In all other cases since $x + ar_2 + br_2 \not\equiv y \mod r_1$ and $y - ar_1 - br_1 \not\equiv x \mod r_2$, the edge $\{x + ar_2 + br_2, y - ar_1 - br_1\}$ is vertex-disjoint from E_1 . Hence we obtain a set E_2

of at least $(r_1r_2 - r_1 - r_2 + 1) \lfloor \lfloor rM/(r_1(p+q)) - 1/r_2 \rfloor/(r_1r_2) \rfloor$ additional distinct edges. Thus $E := E_1 \cup E_2$ is our desired set.

Observe that the graph G_M is a link graph $L_S[B]$, where $S := \{M\}$ and $B := [\lceil rM/q \rceil - 1]$. If we wish to extend a solution-free set S into a solution-free subset of $S \cup B$, then we must pick an independent set in $L_S[B]$. Similarly here if we wish to obtain a solution-free subset of [n] which contains M divisible by t, then we must pick an independent set in G_M . This is the idea behind the following key lemma, which allows us to bound the number of elements in such an \mathcal{L}^w -free set.

Lemma 4.10. Let \mathcal{L} denote the equation px + qy = rz where $p \ge q \ge r$ and p, q, r are fixed positive integers satisfying gcd(p, q, r) = 1. Let t := gcd(p, q) and write $r_1 := p/t$ and $r_2 := q/t$. Let S be an \mathcal{L}^w -free subset of [n]. If $M \in S$ is divisible by t, then S contains at most

$$\left\lceil \frac{rM}{q} \right\rceil - 1 - \left\lfloor \frac{rM}{r_2(p+q)} \right\rfloor - (r_1r_2 - r_1 - r_2 + 1) \left\lfloor \left\lfloor \frac{rM}{r_1(p+q)} - \frac{1}{r_2} \right\rfloor \frac{1}{r_1r_2} \right\rfloor$$

elements from $[\lceil rM/q \rceil - 1]$.

Proof. Consider the graph G_M and observe that its edges correspond to \mathcal{L} -triples: since $p \ge q \ge r$ there is an edge between x and y precisely when $\{x, y, M\}$ is an \mathcal{L} -triple. Hence if $I \subseteq V(G_M)$ is such that $I \cup \{M\}$ is an \mathcal{L}^w -free subset of [n] then I is an independent set in G_M . As a consequence if we find a set of vertex-disjoint edges in G_M of size J, then S contains at most $\lceil rM/q \rceil - 1 - J$ elements from $\lceil rM/q \rceil - 1 \rceil$. The result then follows by applying Lemma 4.9.

First note that if \mathcal{L} denotes the equation x + y = z, then in Lemma 4.10 we are simply saying that if a sum-free set S contains M, then it cannot contain both 1 and M - 1, it cannot contain both 2 and M - 2, and so on. So in a sense this lemma is a generalisation of the proof that sum-free subsets of [n] cannot contain more than [n/2] elements.

Let \mathcal{L} denote the equation px + qy = rz where $p \ge q \ge r$ and p, q, r are fixed positive integers satisfying gcd(p,q,r) = 1 and let t := gcd(p,q). Recall that $T_n := \{x \in [n] : x \not\equiv 0 \mod t\}$ is \mathcal{L}^w -free. Lemma 4.10 roughly implies that every \mathcal{L}^w -free subset of [n] must have 'not too many small elements' or must 'look like' T_n . Clearly this lemma gives rise to an upper bound on the size of the largest \mathcal{L}^w -free subset of [n]. In Section 4.5 we also show that this lemma can be used to obtain an upper bound on the number of maximal \mathcal{L}^w -free subsets of [n].

Recall from the introduction that we can use the following simple proposition to extend our results for linear equations with three variables to linear equations with more than three variables.

Proposition 4.1. Let \mathcal{L}_1 denote the equation $p_1x_1 + \cdots + p_kx_k = b$ where $p_1, \ldots, p_k, b \in \mathbb{Z}$ and let \mathcal{L}_2 denote the equation $(p_1 + p_2)x_1 + p_3x_2 + \cdots + p_kx_{k-1} = b$. Then $\mu(n, \mathcal{L}_1^w) \leq \mu(n, \mathcal{L}_2^w)$ and $f(n, \mathcal{L}_1^w) \leq f(n, \mathcal{L}_2^w)$.

Proof. If $(p_1 + p_2)x_1 + p_3x_2 + \cdots + p_kx_{k-1} = b$ for some $x_i \in [n], 1 \leq i \leq k-1$, then $p_1x_1 + p_2x_1 + p_3x_2 + \cdots + p_kx_{k-1} = b$. Hence any solution to \mathcal{L}_2 in [n] gives rise to a solution to \mathcal{L}_1 in [n]. So if $A \subseteq [n]$ is \mathcal{L}_1^w -free, then A is also \mathcal{L}_2^w -free. Hence the size of the largest \mathcal{L}_2^w -free set is at least the size of the largest \mathcal{L}_1^w -free set, and also there are at least as many \mathcal{L}_2^w -free sets as there are \mathcal{L}_1^w -free sets. \Box

We will also make use of the following trivial fact.

Fact 4.11. Suppose \mathcal{L}_1 is a linear equation and \mathcal{L}_2 is a positive integer multiple of \mathcal{L}_1 . Then the set of \mathcal{L}_1^w -free subsets of [n] is precisely the set of \mathcal{L}_2^w -free subsets of [n]. In particular $\mu(n, \mathcal{L}_1^w) = \mu(n, \mathcal{L}_2^w)$, $f(n, \mathcal{L}_1^w) = f(n, \mathcal{L}_2^w)$ and $f_{\max}(n, \mathcal{L}_1^w) = f_{\max}(n, \mathcal{L}_2^w)$.

The two results above allow us to extend the use of Lemma 4.10 to equations with more than three variables.

Lemma 4.12. Let \mathcal{L} denote the equation $p_1x_1 + \cdots + p_kx_k = 0$ where $p_i \in \mathbb{Z}$. Suppose there is a partition of the p_i into three non-empty parts P_1 , P_2 and P_3 where $p' := \sum_{p_j \in P_1} p_j$, $q' := \sum_{p_j \in P_2} p_j$ and $r' := -\sum_{p_j \in P_3} p_j$ satisfy $p' \ge q' \ge r' \ge 1$. Let $t' := \gcd(p', q', r')$ and write p := p'/t', q := q'/t' and r := r'/t'. Let $t := \gcd(p, q)$ and write $r_1 := p/t$ and $r_2 := q/t$. Let S be an \mathcal{L}^w -free subset of [n]. If $M \in S$ is divisible by t, then S contains at most

$$\left\lceil \frac{rM}{q} \right\rceil - 1 - \left\lfloor \frac{rM}{r_2(p+q)} \right\rfloor - (r_1r_2 - r_1 - r_2 + 1) \left\lfloor \left\lfloor \frac{rM}{r_1(p+q)} - \frac{1}{r_2} \right\rfloor \frac{1}{r_1r_2} \right\rfloor$$

elements from $[\lceil rM/q \rceil - 1]$.

Proof. Let \mathcal{L}_2 denote the equation px + qy = rz. Now observe by repeatedly applying Proposition 4.1 and Fact 4.11 that any \mathcal{L}^w -free set is also an \mathcal{L}_2^w -free set. Hence S must be \mathcal{L}_2^w -free and so we simply apply Lemma 4.10.

This bounds the number of 'small elements' in solution-free sets for equations with more than three variables, and in Theorem 4.18 we will use this lemma to obtain a result for the number of maximal solution-free sets.

4.3 The size of the largest solution-free set

The aim of this section is to use our results from the previous section to obtain bounds on $\mu(n, \mathcal{L}^w)$ for linear equations \mathcal{L} of the form px + qy = rz with $p \ge q \ge r$ positive integers and also linear equations with more than three variables. As previously mentioned we can use Lemma 4.10 to obtain a bound on the size of a solution-free set.

Corollary 4.13. Let \mathcal{L} denote the equation px + qy = rz where $p \ge q \ge r$ and p, q, r are fixed positive integers satisfying gcd(p,q,r) = 1. Let S be an \mathcal{L}^w -free subset of [n] and suppose M is the largest element of S divisible by t := gcd(p,q). Write $r_1 := p/t$ and $r_2 := q/t$. Then

$$|S| \leq M - \left\lfloor \frac{rM}{r_2(p+q)} \right\rfloor - (r_1r_2 - r_1 - r_2 + 1) \left\lfloor \left\lfloor \frac{rM}{r_1(p+q)} - \frac{1}{r_2} \right\rfloor \frac{1}{r_1r_2} \right\rfloor + \left\lceil \frac{(n-M)(t-1)}{t} \right\rceil.$$
Proof. By Lemma 4.10, S contains at most $\lceil rM/q \rceil - 1 - \lfloor rM/(r_2(p+q)) \rfloor - (r_1r_2 - r_2) \rfloor$

 $r_1 - r_2 + 1$ $\lfloor \lfloor rM/(r_1(p+q)) - 1/r_2 \rfloor/(r_1r_2) \rfloor$ elements from $\lceil rM/q \rceil - 1 \rceil$. It also cannot contain any element larger than M and divisible by t.

Note in the statement of Corollary 4.13 we are implicitly assuming that M exists. If it does not then $|S| \leq \lceil n(t-1)/t \rceil$.

We are now ready to prove Theorem 4.3, which determines $\mu(n, \mathcal{L}^w)$ for a wide class of equations of the form px + qy = rz where $p \ge q \ge r$ and p, q, r are fixed positive integers.

Theorem 4.3. Let \mathcal{L} denote the equation px + qy = rz where $p \ge q \ge r$ and p, q, r are fixed positive integers satisfying gcd(p, q, r) = 1. Let t := gcd(p, q). Write $r_1 := p/t$ and $r_2 := q/t$.

- (i) If q divides p and $p + q \leq rq$ then $\mu(n, \mathcal{L}^w) = \lceil (q-1)n/q \rceil$;
- (ii) If q divides p and $p+q \ge rq$ then $\mu(n, \mathcal{L}^w) = \lceil (p+q-r)(n-a)/(p+q) \rceil + a$ where a is the unique non-negative integer $0 \le a < q$ such that n-a is divisible by q;
- (iii) If q does not divide p, t > 1 and

$$r > (r_1 r_2 - r_1 - r_2 + 4) r_2 \left(r_1 + 1 + \frac{r_2 - 1}{r_1^2 + (r_1 - 1)(r_2 - 1)} \right)$$

then $\mu(n, \mathcal{L}^w) = \lfloor (t-1)n/t \rfloor$.

Proof. Let S be an \mathcal{L}^w -free subset of [n] and suppose M is the largest element of S divisible by t. If S does not contain an element divisible by t, set M := 0. If q divides p

then t = q and $r_2 = 1$ and hence by Corollary 4.13 we have

$$|S| \leqslant \left\lceil \frac{(p+q-r)M}{p+q} \right\rceil + \left\lceil \frac{(n-M)(q-1)}{q} \right\rceil.$$
(4.3.1)

(This is true even in the case M = 0.)

If $p + q \leq rq$ then $|S| \leq \lceil (q-1)M/q \rceil + \lceil (n-M)(q-1)/q \rceil = \lceil n(q-1)/q \rceil$ since M is divisible by q. Observe that the set $T_n := \{x \in [n] : x \not\equiv 0 \mod t\}$ is an \mathcal{L}^w -free set obtaining this size, and so this proves (i).

For (ii) we will show that (4.3.1) is an increasing function of M (when restricted to running through M divisible by t) and hence it will be maximised by taking M = n - a. Then $|S| \leq \lceil (p+q-r)(n-a)/(p+q) \rceil + a$. Observe that the interval $I_n := \lfloor \lfloor r(n-a)/(p+q) \rfloor + 1, n \rfloor$ is an \mathcal{L}^w -free set obtaining this size and so this proves (ii), provided (4.3.1) is an increasing function of M.

Since M must be divisible by t = q, write M' := M/q and so (4.3.1) can be written as

$$\left\lceil \frac{((r_1+1)q-r)M'}{r_1+1} \right\rceil + \left\lceil \frac{n(q-1)}{q} \right\rceil - M'(q-1) = M' + \left\lceil \frac{-rM'}{r_1+1} \right\rceil + \left\lceil \frac{n(q-1)}{q} \right\rceil.$$

Now observe that the difference between successive terms M' and M' + 1 is given by

$$M' + 1 + \left\lceil \frac{-r(M'+1)}{r_1 + 1} \right\rceil - M' - \left\lceil \frac{-rM'}{r_1 + 1} \right\rceil = 1 + \left\lceil \frac{-rM'}{r_1 + 1} - \frac{r}{r_1 + 1} \right\rceil - \left\lceil \frac{-rM'}{r_1 + 1} \right\rceil \ge 0$$

where the inequality follows since $r_1 + 1 \ge r$. Hence (4.3.1) is an increasing function of M as required.

For (iii) if M = 0 then $|S| \leq \lceil n(t-1)/t \rceil$ as required. So assume $M \ge t$. Then by Corollary 4.13 we have

$$\begin{split} S| &\leq M - \left\lfloor \frac{rM}{r_2(p+q)} \right\rfloor - (r_1r_2 - r_1 - r_2 + 1) \left\lfloor \left\lfloor \frac{rM}{r_1(p+q)} - \frac{1}{r_2} \right\rfloor \frac{1}{r_1r_2} \right\rfloor \\ &+ \left\lceil \frac{(n-M)(t-1)}{t} \right\rceil \\ &\leq M - \frac{rM}{r_2(p+q)} + 1 - \frac{r_1r_2 - r_1 - r_2 + 1}{r_1r_2} \left(\frac{rM}{r_1(p+q)} - \frac{1}{r_2} - 1 \right) + r_1r_2 - r_1 - r_2 + 1 \\ &- \frac{M(t-1)}{t} + \left\lceil \frac{n(t-1)}{t} \right\rceil \\ &\leq \left\lceil \frac{n(t-1)}{t} \right\rceil + r_1r_2 - r_1 - r_2 + 3 - M \left(\frac{r(r_1^2 + (r_1 - 1)(r_2 - 1))}{tr_1^2 r_2(r_1 + r_2)} - \frac{1}{t} \right) \\ &= \left\lceil \frac{n(t-1)}{t} \right\rceil + r_1r_2 - r_1 - r_2 + 3 - M \left(\frac{r}{tr_2} \left(r_1 + 1 + \frac{r_2 - 1}{r_1^2 + (r_1 - 1)(r_2 - 1)} \right)^{-1} - \frac{1}{t} \right) \\ &\leq \left\lceil \frac{n(t-1)}{t} \right\rceil + r_1r_2 - r_1 - r_2 + 3 - M \left(\frac{r_1r_2 - r_1 - r_2 + 4}{t} - \frac{1}{t} \right) \\ &\leq \left\lceil \frac{n(t-1)}{t} \right\rceil + r_1r_2 - r_1 - r_2 + 3 - M \left(\frac{r_1r_2 - r_1 - r_2 + 4}{t} - \frac{1}{t} \right) \\ &\leq \left\lceil \frac{n(t-1)}{t} \right\rceil + r_1r_2 - r_1 - r_2 + 3 - M \left(\frac{r_1r_2 - r_1 - r_2 + 4}{t} - \frac{1}{t} \right) \end{split}$$

where the penultimate inequality follows by our lower bound on r and the last inequality follows by using $M \ge t$.

For Theorem 4.3(iii) it is easy to check that actually given the conditions on r we must always have t > 1 (we just state t > 1 in the theorem for clarity). As an example, $p := 3t, q := 2t, r \ge 41$, and $t \ge r/2$ gives a set of equations which satisfy the conditions of Theorem 4.3(iii).

Theorem 4.3 together with Proposition 4.1 yield results for $\mu(n, \mathcal{L}^w)$ where \mathcal{L} is an equation with more than three variables.

Corollary 4.14. Let \mathcal{L} denote the equation $a_1x_1 + \cdots + a_kx_k + b_1y_1 + \cdots + b_\ell y_\ell = c_1z_1 + \cdots + c_m z_m$ where the $a_i, b_i, c_i \in \mathbb{N}$ and $p' := \sum_i a_i, q' := \sum_i b_i$ and $r' := \sum_i c_i$ satisfy $p' \ge q' \ge r'$. Let $t' := \gcd(p', q', r')$ and write p := p'/t', q := q'/t' and r := r'/t'. Let $t := \gcd(p, q)$.

- (i) If m = 1, $\ell = 1$, $q' = b_1$ divides a_i for all $1 \leq i \leq k$ and $p + q \leq rq$ then $\mu(n, \mathcal{L}^w) = \lceil (q-1)n/q \rceil;$
- (ii) If q divides p and $p + q \ge rq$ then $\lceil (p + q r)n/(p + q) \rceil \le \mu(n, \mathcal{L}^w) \le \lceil (p + q r)(n a)/(p + q) \rceil + a$ where a is the unique non-negative integer $0 \le a < q$ such that n a is divisible by q;
- (iii) Write $r_1 := p/t$ and $r_2 := q/t$. If q does not divide p, m = 1, tt' divides a_i and b_j for $1 \le i \le k, 1 \le j \le \ell$ and

$$r > (r_1 r_2 - r_1 - r_2 + 4) r_2 \left(r_1 + 1 + \frac{r_2 - 1}{r_1^2 + (r_1 - 1)(r_2 - 1)} \right)$$

then $\mu(n, \mathcal{L}^w) = [(t-1)n/t].$

Proof. Let \mathcal{L}_2 denote the equation px + qy = rz. Then we have $\mu(n, \mathcal{L}^w) \leq \mu(n, \mathcal{L}_2^w)$ by repeated use of Proposition 4.1 and Fact 4.11. Then the use of Theorem 4.3(i), (ii) and (iii) respectively for each of the three cases stated gives the required upper bound.

For the lower bounds for each case we show that a suitable \mathcal{L}_2^w -free set is also \mathcal{L}^w -free. For (i) and (iii) consider the \mathcal{L}_2^w -free set $T := \{x \in [n] : x \not\equiv 0 \mod t\}$. Since tt' divides a_i and b_j for $1 \leq i \leq k, 1 \leq j \leq \ell$ (noting for (i) that tt' = qt' = q'), any solution $\{x_1, \ldots, x_k, y_1, \ldots, y_\ell, z_1\}$ to \mathcal{L} must have z_1 divisible by t. But T contains no elements divisible by t and hence $\mu(n, \mathcal{L}^w) \geq \lceil (t-1)n/t \rceil$.

For (ii) consider the \mathcal{L}_2^w -free set $\lfloor \lfloor rn/(p+q) \rfloor + 1, n \rfloor$. Observe that

$$(a_1 + \dots + a_k + b_1 + \dots + b_\ell)(\lfloor rn/(p+q) \rfloor + 1) = (p'+q')(\lfloor r'n/(p'+q') \rfloor + 1)$$

> $r'n = (c_1 + \dots + c_m)n$

and so there are no solutions to \mathcal{L} in the interval $[\lfloor rn/(p+q) \rfloor + 1, n]$, and so $\mu(n, \mathcal{L}^w) \ge \lceil (p+q-r)n/(p+q) \rceil$.

Note that for Corollary 4.14(ii) the lower bound is very close to the upper bound, and in particular matches it in the case where r = 1.

4.4 The number of solution-free sets

In this section we will apply Lemma 4.9 and Proposition 4.1 to obtain upper bounds on the number of solution-free sets for various equations. Let $f(n, \mathcal{L}^w, M)$ denote the number of \mathcal{L}^w -free subsets of [n] with maximum element M.

Lemma 4.15. Let \mathcal{L} denote the equation px + qy = rz where $p \ge q \ge r$ and p, q, r are fixed positive integers satisfying gcd(p,q,r) = 1. Assume that $M \in [n]$ is divisible by t := gcd(p,q). Write $r_1 := p/t$ and $r_2 := q/t$, and let $c := 2^{(2-\log_2 3)(r_1r_2 - r_1 - r_2 + 4) - 1}$. Then

$$f(n, \mathcal{L}^w, M) \leq c \cdot 2^{M(1 - rt(2 - \log_2 3)(p^2 + (p - t)(q - t))/(p^2 q(p + q)))}.$$

Proof. Note that if I is an \mathcal{L}^w -free set with maximum element M, then $I \cap [\lceil rM/q \rceil - 1]$ is an independent set in G_M . Since M is divisible by t, by Lemma 4.9 there exists a matching H in G_M of size

$$J := \left\lfloor \frac{rM}{r_2(p+q)} \right\rfloor + (r_1r_2 - r_1 - r_2 + 1) \left\lfloor \left\lfloor \frac{rM}{r_1(p+q)} - \frac{1}{r_2} \right\rfloor \frac{1}{r_1r_2} \right\rfloor - 1$$

$$\geqslant rtM \left(\frac{p^2 + (p-t)(q-t)}{p^2q(p+q)} \right) - r_1r_2 + r_1 + r_2 - 4.$$

Let A_1, \ldots, A_{J+1} be a partition of $V(G_M)$ where if $1 \leq i \leq J$ then A_i contains precisely the two vertices from some edge in H. So $|A_{J+1}| = \lceil rM/q \rceil - 1 - 2J$. Recalling that we use i(G) for the number of independent sets in G, we have $i(G_M[A_i]) = 3$ for $1 \leq i \leq J$ and $i(G_M[A_{J+1}]) \leq 2^{\lceil rM/q \rceil - 1 - 2J}$. So Lemma 3.8 implies that $i(G_M) \leq 3^J 2^{\lceil rM/q \rceil - 1 - 2J}$. Observe that every \mathcal{L}^w -free subset of [n] with maximum element M can be found by picking an independent set $I \subseteq V(G_M)$ and extending it in $[\lceil rM/q \rceil, M-1]$. There are at most $2^{M-\lceil rM/q \rceil}$ ways to form this extension. Hence we have

$$f(n, \mathcal{L}^w, M) \leq i(G_M) \cdot 2^{M - \lceil rM/q \rceil} \leq 3^J 2^{M - 2J - 1} = 2^{M - (2 - \log_2 3)J - 1}$$
$$\leq 2^{M(1 - rt(2 - \log_2 3)(p^2 + (p - t)(q - t))/(p^2q(p + q)))} 2^{(2 - \log_2 3)(r_1r_2 - r_1 - r_2 + 4) - 1}.$$

Theorem 4.16. Let \mathcal{L} denote the equation px + qy = rz where $p \ge q \ge r$ and p, q, r are fixed positive integers satisfying gcd(p,q,r) = 1. Let t := gcd(p,q). Suppose

$$r > (r_1 r_2 - r_1 - r_2 + 4) r_2 \left(r_1 + 1 + \frac{r_2 - 1}{r_1^2 + (r_1 - 1)(r_2 - 1)} \right).$$

Then $f(n, \mathcal{L}^w) \leq C \cdot 2^{\mu(n, \mathcal{L}^w)}$ where

$$C := \frac{2^{(2-\log_2 3)(r_1r_2 - r_1 - r_2 + 4)}}{1 - 2^{1 - rt^2(2 - \log_2 3)(p^2 + (p-t)(q-t))/(p^2q(p+q))}}.$$

Proof. If q does not divide p, since $r > (r_1r_2 - r_1 - r_2 + 4)r_2(r_1 + 1 + (r_2 - 1)/(r_1^2 + (r_1 - 1)(r_2 - 1)))$ we have $\mu(n, \mathcal{L}^w) = \lceil (t - 1)n/t \rceil$ by Theorem 4.3(iii). If q divides p then our condition on r implies $rq \ge p + q$ and so we have $\mu(n, \mathcal{L}^w) = \lceil (q - 1)n/q \rceil = \lceil (t - 1)n/t \rceil$ by Theorem 4.3(i).

Every \mathcal{L}^{w} -free subset A of [n] can be written as $A = B \cup D$ where the maximum element M of B is divisible by t and min D > M and no element of D is divisible by t. (Note here B or D could be empty. If B is empty then we define M := 0.) Thus

$$f(n,\mathcal{L}^w) \leq \sum_{\substack{M \equiv 0 \bmod t \\ 0 \leq M \leq n}} f(n,\mathcal{L}^w,M) \cdot 2^{\lceil (n-M)(t-1)/t \rceil} = \sum_{\substack{M \equiv 0 \bmod t \\ 0 \leq M \leq n}} f(n,\mathcal{L}^w,M) \cdot 2^{\mu(n,\mathcal{L}^w) - M(t-1)/t}$$
where we define $f(n, \mathcal{L}^w, 0) := 1$.

Let $b := 1 - rt(2 - \log_2 3)(p^2 + (p-t)(q-t))/(p^2q(p+q)), d := 2^{bt-t+1}$ and $a := c \cdot 2^{\mu(n,\mathcal{L}^w)}$ where c is as stated in Lemma 4.15. Note that |d| < 1 if $r > r_2(r_1 + 1 + (r_2 - 1)/(r_1^2 + (r_1 - 1)(r_2 - 1)))/(2 - \log_2 3)$. But this is true since $(r_1r_2 - r_1 - r_2 + 4) > 1/(2 - \log_2 3)$. By using Lemma 4.15 we obtain

$$\begin{split} f(n,\mathcal{L}^w) &\leqslant \sum_{\substack{M \equiv 0 \, \text{mod} \, t \\ 0 \leqslant M \leqslant n}} c \cdot 2^{M(1-rt(2-\log_2 3)(p^2+(p-t)(q-t))/(p^2q(p+q)))} 2^{\mu(n,\mathcal{L}^w)-M(t-1)/t} \\ &= \sum_{\substack{M \equiv 0 \, \text{mod} \, t \\ 0 \leqslant M \leqslant n}} a \cdot 2^{M(b-(t-1)/t)} = \sum_{i=0}^{\lfloor n/q \rfloor} a \cdot 2^{it(b-(t-1)/t)} \\ &= \sum_{i=0}^{\lfloor n/q \rfloor} a d^i \leqslant \sum_{i=0}^{\infty} a d^i = \frac{a}{1-d} \\ &= C \cdot 2^{\mu(n,\mathcal{L}^w)} \text{ where } C := \frac{2^{(2-\log_2 3)(r_1r_2-r_1-r_2+3)}}{1-2^{1-rt^2(2-\log_2 3)(p^2+(p-t)(q-t))/(p^2q(p+q))}}. \end{split}$$

Note Theorem 4.16 implies Theorem 4.4. Recall Theorem 3.3 from the last chapter. Both of these involve equations with three variables. To prove the following corollary, we use Theorems 4.4 and 3.3 respectively, together with repeated use of Proposition 4.1.

Corollary 4.17. Let \mathcal{L} denote one of the following equations:

(i) $a_1x_1 + \dots + a_kx_k + b_1y_1 + \dots + b_\ell y_\ell = r'z$ where $a_i, b_i, r' \in \mathbb{N}$ and $p' = \sum_i a_i$, $q' = \sum_i b_i \text{ satisfy } p' \ge q' \ge r'$. Let $t' := \gcd(p', q', r')$ and write p := p'/t', q := q'/t'and r := r'/t'. Write $r_1 := p/t$ and $r_2 := q/t$. Suppose additionally tt' divides a_i and b_j for $1 \le i \le k, 1 \le j \le \ell$ and

$$r > (r_1 r_2 - r_1 - r_2 + 4) r_2 \left(r_1 + 1 + \frac{r_2 - 1}{r_1^2 + (r_1 - 1)(r_2 - 1)} \right).$$

(ii) $a_1x_1 + \dots + a_kx_k + b_1y_1 + \dots + b_\ell y_\ell = z$ where the $a_i, b_i \in \mathbb{N}$ and $p = \sum_i a_i$, $q = \sum_i b_i$ satisfy $p \ge q$ and additionally either q = 1 and $p \ge 3$ or $q \ge 2$ and p > q(3q-2)/(2q-2).

Then $f(n, \mathcal{L}^w) = \Theta(2^{\mu(n, \mathcal{L}^w)}).$

Proof. Let \mathcal{L}_2 denote the equation px + qy = rz. By repeated use of Proposition 4.1 and Fact 4.11 we have $f(n, \mathcal{L}^w) \leq f(n, \mathcal{L}^w_2)$. By Theorems 4.4 and 3.3 respectively for cases (i) and (ii), we have $f(n, \mathcal{L}^w_2) \leq C \cdot 2^{\mu(n, \mathcal{L}^w_2)}$ for some constant C. But by Corollary 4.14 we have $\mu(n, \mathcal{L}^w) = \mu(n, \mathcal{L}^w_2)$ and so $f(n, \mathcal{L}^w) \leq C \cdot 2^{\mu(n, \mathcal{L}^w)}$.

4.5 The number of maximal solution-free sets

We start this section with the proof of Theorem 4.5.

Theorem 4.5. Let \mathcal{L} denote the equation px + qy = rz where $p \ge q \ge r$ and p, q, r are fixed positive integers satisfying gcd(p,q,r) = 1. Let t := gcd(p,q). Then

$$f_{\max}(n, \mathcal{L}^w) \leqslant 2^{\frac{Crn}{q} + o(n)} \text{ where } C := 1 - \frac{t}{p+q} \Big(\frac{p^2 + (p-t)(q-t)}{p^2} \Big).$$

Proof. First note that C lies between 1/2 and 1 - t/(p+q). To see this, note that if q divides p, then $C = 1 - q/(p+q) \ge 1/2$ since $p \ge q$. Otherwise, p > q > t, and so $(p-t)(q-t) < p^2$. Hence $t(p^2+(p-t)(q-t))/(p^2(p+q)) < 2t/(p+q) \le 2(q/2)/(p+q) < 1/2$ and so C > 1/2. We observe that $C \le 1 - t/(p+q)$ since $p \ge q \ge t$.

Let \mathcal{F} denote the set of containers obtained by applying Lemma 2.5. Since every maximal \mathcal{L}^w -free subset of [n] lies in at least one of the $2^{o(n)}$ containers, it suffices to show that every $F \in \mathcal{F}$ houses at most $2^{Crn/q+o(n)}$ maximal \mathcal{L}^w -free sets.

Let $F \in \mathcal{F}$. By Lemmas 2.5(i) and 2.4, $F = A \cup B$ where |A| = o(n), $|B| \leq \mu(n, \mathcal{L}^w)$ and B is \mathcal{L}^w -free. Define $M := \max\{x \in B : x \equiv 0 \mod t\}$ and u :=

 $\max\{\lfloor rM/q \rfloor, \lfloor rn/2q \rfloor\}$. Every maximal \mathcal{L}^w -free set which lies in such a container can be constructed by:

- (i) Picking $S_1 \subseteq A$ to be \mathcal{L}^w -free;
- (ii) Adding a set $S_2 \subseteq [u] \cap B$ so that $S_1 \cup S_2$ is \mathcal{L}^w -free;
- (iii) Choosing a set $S_3 \subseteq [u+1,n] \cap B$ so that $S_1 \cup S_2 \cup S_3$ is a maximal \mathcal{L}^w -free subset of [n].

There are $2^{o(n)}$ ways to pick S_1 . If $M \leq n/2$ then $u = \lfloor rn/2q \rfloor$ and so there are at most $2^{rn/2q} \leq 2^{Crn/q}$ ways to pick S_2 so that $S_1 \cup S_2$ is \mathcal{L}^w -free. Write $r_1 := p/t$ and $r_2 := q/t$. If $M \geq n/2$ then since M is divisible by t, we apply Lemma 4.10 to show that

$$\begin{split} |[u] \cap B| &= |[\lfloor rM/q \rfloor] \cap B| \\ &\leqslant \left\lfloor \frac{rM}{q} \right\rfloor - \left\lfloor \frac{rM}{r_2(p+q)} \right\rfloor - (r_1r_2 - r_1 - r_2 + 1) \left\lfloor \left\lfloor \frac{rM}{r_1(p+q)} - \frac{1}{r_2} \right\rfloor \frac{1}{r_1r_2} \right\rfloor \\ &= \frac{CrM}{q} + o(n). \end{split}$$

Hence there are at most $2^{CrM/q+o(n)} \leq 2^{Crn/q+o(n)}$ ways to pick S_2 so that $S_1 \cup S_2$ is \mathcal{L}^w -free.

Let $B' := [u+1, n] \cap B$. For step (iii) we calculate the number of extensions of $S_1 \cup S_2$ into B'. Observe by Lemma 4.7, this is bounded above by $MIS(L_{S_1 \cup S_2}[B'])$. We will show that this link graph has only one maximal independent set. Then combining steps (i)-(iii) we have that F contains at most $2^{o(n)} \times 2^{Crn/q+o(n)} = 2^{Crn/q+o(n)}$ maximal \mathcal{L}^w -free sets as desired.

If the link graph only contains loops and isolated vertices, then it has only one maximal independent set. For it to have an edge between distinct vertices, we either must have $x, z \in B', y \in S_1 \cup S_2$ such that px + qy = rz or py + qx = rz, or we must have $x, y \in B'$, $z \in S_1 \cup S_2$ such that px + qy = rz. The first of these events does not occur since otherwise $rz \ge q(x + y) > qx \ge q(\lfloor rM/q \rfloor + 1) > rM$ and so z > M. Note that since z is part of the solution px + qy = rz and gcd(p,q,r) = 1, it must be divisible by t. However this contradicts z > M as we have $z \in B$, but M was defined to be the largest element in B divisible by t.

If M > n/2 then the second event does not occur since $rz = px + qy \ge q(x+y) \ge 2q(\lfloor rM/q \rfloor + 1) > 2rM > rn$ and so z > n. If $M \le n/2$ then the second event does not occur since $rz = px + qy \ge q(x+y) \ge 2q(\lfloor rn/2q \rfloor + 1) > rn$ and so again z > n.

Note that when r = 1, Theorem 4.5 gives us new results for equations of the form px + qy = z. Recall that in the last chapter we found results for such equations. In Section 4.6 we give a summary describing which result gives the best upper bound for various values of p and q.

When \mathcal{L} denotes the equation qx + qy = z for some positive integer $q \ge 2$, Proposition 3.21(iii) gives a lower bound of $f_{\max}(n, \mathcal{L}^w) \ge 2^{(n-6q)/2q}$. Combining this with Theorem 4.5 allows us to determine $\log(f_{\max}(n, \mathcal{L}^w))$ asymptotically.

Theorem 4.6. Let \mathcal{L} denote the equation qx + qy = z where $q \ge 2$ is an integer. Then

$$f_{\max}(n, \mathcal{L}^w) = 2^{n/2q + o(n)}.$$

By adapting the proof of Theorem 4.5 we obtain the following result for $f_{\max}(n, \mathcal{L}^w)$ for some equations with more than three variables.

Theorem 4.18. Let \mathcal{L} denote the equation $p_1x_1 + \cdots + p_kx_k = rz$ where $p_1, \ldots, p_k, r \in \mathbb{N}$ satisfy $gcd(p_1, \ldots, p_k, r) = 1$ and $p_1 \ge \cdots \ge p_k \ge r$. Suppose that $p := \sum_{i=1}^{k-1} p_i$ and $q := p_k$ satisfy $t := gcd(p, q) = gcd(p_1, \ldots, p_k)$. Then

$$f_{\max}(n, \mathcal{L}^w) \leqslant 2^{\frac{Crn}{q} + o(n)} \text{ where } C := 1 - \frac{t}{p+q} \Big(\frac{p^2 + (p-t)(q-t)}{p^2} \Big).$$

Proof. We follow the proof used in Theorem 4.5 precisely (except for using Lemma 4.12 instead of Lemma 4.10 in step (ii)) up until counting the number of ways of extending $S_1 \cup S_2$ to a maximal \mathcal{L}^w -free set in $B' := [u+1,n] \cap B$. Observe by Lemma 4.7, this is bounded above by $\operatorname{MIS}(L_{S_1 \cup S_2}[B'])$ since B' and $S_1 \cup S_2$ are \mathcal{L}^w -free. To see that B' is \mathcal{L}^w -free, suppose (x_1, \ldots, x_k, z) is a solution within B' and note that $rz = p_1x_1 + \cdots + p_kx_k > p_kx_k \ge q(rM/q) = rM$ and so z > M. (Here we needed that each p_i is positive.) Since $\operatorname{gcd}(p_1, \ldots, p_k, r) = 1$ and $\operatorname{gcd}(p, q) = \operatorname{gcd}(p_1, \ldots, p_k)$ we have $\operatorname{gcd}(t, r) = 1$ and so in any solution to \mathcal{L} , z must be divisible by t. However this contradicts z > M as we have $z \in B$, but M was defined to be the largest element in B divisible by t.

We will show that this link hypergraph $L_{S_1 \cup S_2}[B']$ has only one maximal independent set (and hence the number of maximal \mathcal{L}^w -free sets contained in F is at most $2^{Crn/q+o(n)}$ as required). If the link hypergraph only contains loops and isolated vertices, then it has only one maximal independent set.

For it to have a hyperedge between at least two vertices, there must exist a solution (x_1, \ldots, x_k, z) where either there is a hyperedge with distinct vertices $x_i, z \in B'$ for some $1 \leq i \leq k$ and $\{x_1, \ldots, x_{i-1}, x_{i+1}, \ldots, x_k\} \subseteq B' \cup S_1 \cup S_2$, or there is a hyperedge with distinct vertices $x_i, x_j \in B'$ for some $1 \leq i < j \leq k$ and

$$\{x_1, \dots, x_{i-1}, x_{i+1}, \dots, x_{j-1}, x_{j+1}, \dots, x_k, z\} \subseteq B' \cup S_1 \cup S_2.$$

Suppose the first event occurs with (x_1, \ldots, x_k, z) . Then $rz = p_1x_1 + \cdots + p_kx_k > p_ix_i \ge p_kx_i = qx_i \ge q(rM/q) = rM$ and so z > M. But since z is part of a solution, it must be divisible by t. This contradicts $z \in B$, since M was defined to be the largest element in B divisible by t.

If M > n/2 then the second event does not occur since $rz = p_1x_1 + \cdots + p_kx_k > p_k(x_i + x_j) \ge 2q(\lfloor rM/q \rfloor + 1) > 2rM > rn$ and so z > n. If $M \le n/2$ then the second

event does not occur since $rz = p_1x_1 + \cdots + p_kx_k > p_k(x_i + x_j) \ge 2q(\lfloor rn/(2q) \rfloor + 1) > rn$ and so again z > n.

We end the section with a lower bound.

Proposition 4.19. Let \mathcal{L} denote the equation qx + qy = rz where q > r and q, r are fixed positive integers satisfying gcd(q, r) = 1. Then

$$f_{\max}(n, \mathcal{L}^w) \ge 2^{\lceil \lfloor rn/2q - rq/2 \rfloor (q-1)/q \rceil - 1}.$$

Proof. Let *B* be the \mathcal{L}^w -free set $\{z \in [n] : z \not\equiv 0 \mod q\}$. Let $M := \max\{z \in [n] : rz/q^2 \in [n]\}$; so $M > n - q^2$. Let $S := \{M\}$ and consider the link graph $L_S[B]$. Note that if $i \in B$ where i < rM/q then $rM/q - i \in B$. This follows since $rM/q^2 \in \mathbb{N}$ and so $rM/q - i \not\equiv 0 \mod q$. Hence there is an edge in $L_S[B]$ between every such i and rM/q - i since q(i + rM/q - i) = rM. By running through all $i \in B$ we obtain a total of $\lceil \lfloor rM/2q \rfloor (q-1)/q \rceil$ disjoint edges in $L_S[B]$ of which at most one is a loop (at rM/2q if it is an integer not congruent to 0 modulo q). Hence we obtain an *induced* matching E in $L_S[B]$ of size $\lceil \lfloor rM/2q \rfloor (q-1)/q \rceil - 1 \ge \lceil \lfloor rn/2q - rq/2 \rfloor (q-1)/q \rceil - 1$. It is easy to see that the matching E contains $2^{|E|}$ maximal independent sets. Since E is an induced subgraph of $L_S[B]$, by applying Lemma 4.8 we obtain the result.

Question 4.20. Let \mathcal{L} denote the equation qx + qy = rz where $q > r \ge 2$ and q, r are fixed positive integers satisfying gcd(q, r) = 1. Does $f_{max}(n, \mathcal{L}^w) = 2^{rn(q-1)/(2q^2) + o(n)}$?

We remark that it turns out that for equations \mathcal{L} as in Question 4.20, we have that $2^{rn(q-1)/(2q^2)+o(n)} = 2^{(\mu(n,\mathcal{L}^w)-\mu^*(n,\mathcal{L}))/2+o(n)}$.

4.6 Best upper bounds on $f_{\max}(n, \mathcal{L}^w)$

In this section we give a summary of the best known upper bound on $f_{\max}(n, \mathcal{L}^w)$ for equations of the form px + qy = rz where $p \ge q \ge r$. First we recall some of the results from the previous chapter.

Theorem 3.4. Let \mathcal{L} be a fixed homogeneous three-variable linear equation. Then

$$f_{\max}(n, \mathcal{L}^w) \leqslant 3^{(\mu(n, \mathcal{L}^w) - \mu^*(n, \mathcal{L}))/3 + o(n)}.$$

Theorem 3.5. Let \mathcal{L} denote the equation px + qy = z where $p \ge q \ge 2$ are integers so that $p \le q^2 - q$ and gcd(p,q) = q. Then

$$f_{\max}(n, \mathcal{L}^w) \leqslant 2^{(\mu(n, \mathcal{L}^w) - \mu^*(n, \mathcal{L}))/2 + o(n)}.$$

Theorem 3.6. Let \mathcal{L} denote the equation px + qy = z where $p \ge q$, $p \ge 2$ and $p, q \in \mathbb{N}$. Then

$$f_{\max}(n, \mathcal{L}^w) \leqslant 2^{\mu(\lfloor \frac{n-p}{q} \rfloor, \mathcal{L}^w) + o(n)}.$$

We can generalise Theorem 3.6 for r > 1 by following a very similar proof to that of Lemma 3.20 to obtain the following.

Theorem 4.21. Let \mathcal{L} denote the equation px + qy = rz where $p \ge q \ge r$ and $p, q, r \in \mathbb{N}$. Then

$$f_{\max}(n, \mathcal{L}^w) \leqslant 2^{\mu(\lfloor \frac{rn}{q} \rfloor, \mathcal{L}^w) + o(n)}.$$

The upper bounds given by Theorems 3.5, 3.6 and 4.21 are all superseded by the bound given by Theorem 4.5. We prove this here and also compare Theorem 4.5 with Theorem 3.4.

Proposition 4.22. Let \mathcal{L} denote the equation px + qy = rz where $p \ge q \ge r$, $p \ge 2$ and p, q, r are fixed positive integers satisfying gcd(p, q, r) = 1. Let t := gcd(p, q) and let $a := \log_2 3$. The best upper bound on $f_{max}(n, \mathcal{L}^w)$ given by Theorems 3.4, 3.5, 3.6, 4.5 and 4.21 is: (i) $f_{\max}(n, \mathcal{L}^w) \leq 3^{(\mu(n, \mathcal{L}^w) - \mu^*(n, \mathcal{L}))/3 + o(n)}$ if

(a)
$$r = 1$$
, $gcd(p,q) = q$, $p \ge \max \{q^2, (q^2 - q)a/(q(3 - 2a) + a)\}$, and $q \le 9$;
(b) $r \ge 2$, $\mu(n, \mathcal{L}^w) = \lceil (t-1)n/t \rceil$, and additionally (1) $p \ne q$ or (2) $2 \le q \le 18$;
(c) $r \ge 2$, q divides p , $p + q \ge rq$ and additionally (1) $p \ne q$ or (2) $2 \le q \le 18$;

(*ii*)
$$f_{\max}(n, \mathcal{L}^w) \leq 2^{Crn/q+o(n)}$$
 where $C := 1 - t(p^2 + (p-t)(q-t))/(p^2(p+q))$ if

(a)
$$r = 1$$
, $gcd(p,q) \neq q$ or $q > 9$ or $p < q^2$ or $p < (q^2 - q)a/(q(3 - 2a) + a)$,
(b) $r \ge 2$, $\mu(n, \mathcal{L}^w) = \lceil (t - 1)n/t \rceil$, and $p = q \ge 19$.

Proof. First suppose that r = 1 (and so $\mu(n, \mathcal{L}^w) = \lceil (p+q-1)n/(p+q) \rceil$). Note that $C \leq 1 - t/(p+q) = (p+q-t)/(p+q) \leq (p+q-1)/(p+q)$ and so the exponent given by Theorem 4.5 is at most the exponent given by Theorem 3.6. For Theorem 3.5 we require gcd(p,q) = q and $p \leq q^2 - q$. In this case C = p/(p+q) and $(p+q-1)/(2(p+q)) - (q-1)^2/(2q^2) = (2pq+q^2-p-q)/(2q^2(p+q)) \geq p/(q(p+q)) = C/q$ and so the exponent given by Theorem 4.5 is at most the exponent given by Theorem 3.5.

For r = 1 it remains to check when the bound given by Theorem 3.4 is better than the bound given by Theorem 4.5. Since Theorem 3.5 gives a better bound than Theorem 3.4 when gcd(p,q) = q and $p \leq q^2 - q$, it suffices to consider the case when gcd(p,q) = q and $p \geq q^2$, and when $gcd(p,q) \neq q$. For the latter, when $t = gcd(p,q) \neq q$, it certainly suffices to show that $2\mu(\lfloor (n-p)/q \rfloor, \mathcal{L}^w) \leq \mu(n, \mathcal{L}^w) - \mu^*(n, \mathcal{L}) + o(n)$, since Theorem 4.5 gives a better bound than Theorem 3.6. In this case we have $\mu^*(n, \mathcal{L}) = (q-1)(t-1)n/(qt) + o(n)$, and hence it suffices to show that $t \leq (pq+q^2-p-q)/(p+2q-2)$. First note that $t \leq q/2$ and so $q \neq 1$. Now observe that $t(p+2q-2) \leq q(p+2q-2)/2 = pq/2+q^2-q \leq pq+q^2-p-q$ and so our inequality on t holds as required. Now suppose gcd(p,q) = q and $p \geq q^2$. To prove (i)(a) it suffices to show that

$$3^{\frac{(p+q-1)n}{3(p+q)} - \frac{(q-1)^2n}{3q^2}} \leqslant 2^{\frac{pn}{q(p+q)}},$$

or rearranging

$$p(q(3-2a)+a) \ge (q^2-q)a.$$

If $q \ge 10$ then (q(3-2a) + a) is negative, but then we would require p negative, a contradiction. Hence we must have $q \le 9$ and then the inequality holds if $p > (q^2 - q)a/(q(3-2a) + a)$.

Now suppose that $r \ge 2$ and $\mu(n, \mathcal{L}^w) = \lceil (t-1)n/t \rceil$. Then $\mu(n, \mathcal{L}^w) - \mu^*(n, \mathcal{L}) = r(t-1)n/(qt) + o(n)$ and $3^{x/3} < 2^x$ and so Theorem 3.4 gives a better bound than Theorem 3.6. We wish to know when

$$3^{\frac{r}{q}\frac{t-1}{3t}} < 2^{\frac{r}{q}(1-\frac{t}{p+q}(\frac{p^2+(p-t)(q-t)}{p^2}))}$$

Write $r_1 := p/t$ and $r_2 := q/t$. The above rearranges to give

$$t((a-3)r_1^2(r_1+r_2) + 3r_1^2 + 3(r_1-1)(r_2-1)) < ar_1^2(r_1+r_2).$$

The right hand side is positive and the left hand side is negative unless $r_1 = r_2 = 1$. In this case p = q = t and so we now require $3^{(q-1)/(3q)} < 2^{1/2}$, which holds when $q \leq 18$.

Finally suppose that $r \ge 2$, q divides p and $p + q \ge rq$ (so $\mu(n, \mathcal{L}^w) = \lceil (p + q - r)n/(p+q)\rceil$). Since q divides p, we have t = q and $p = r_1q$, and so Theorem 4.5 gives a bound of $2^{rpn/(q(p+q))+o(n)}$. This is better than Theorem 4.21 which gives a bound of $2^{r(p+q-r)n/(q(p+q))+o(n)}$ since $q \ge r$. Therefore we wish to know when

$$3^{\frac{p+q-r}{3(p+q)} - \frac{(q-r)(q-1)}{3q^2}} < 2^{\frac{rp}{q(p+q)}}.$$

Rearranging, we require $r_1(a(q+qr-r)/3-qr) \leq a(r-q)/3$. Now note a(q+qr-r)/3-qris negative when $r \geq 2$, so this rearranges to give $r_1 \geq (q-r)/(r-q-rq+3rq/a)$. If p > q (so $r_1 \geq 2$), it suffices to have $2 \geq (q-r)/(r-q-rq+3rq/a)$ or rearranging, $q(r(2-6/a)+3) \leq 3r$. This holds since r(2-6/a)+3 is negative for $r \geq 2$. Otherwise p = q, and so since $p+q \geq rq$, we have that r = 2. So we require $1 \geq (q-2)/((6/a-3)q+2)$ which holds when $q \leq 18$. (In this final case, $\mu(n, \mathcal{L}^w) = \lceil (t-1)n/t \rceil = \lceil (p+q-r)n/(p+q) \rceil = \lceil (q-1)n/q \rceil$.)

4.7 Concluding Remarks

The crucial trick used in the proof of Theorems 4.5 and 4.18 was to choose our sets S carefully so that the link hypergraphs $L_S[B]$ each contain precisely one maximal independent set. In other applications of this method (see [6, 7] and Chapter 3) the approach was to instead obtain other structural properties of the link graphs (such as being triangle-free) to ensure there are not too many maximal independent sets in $L_S[B]$. It would be interesting to see if the approach of this chapter can be applied to obtain other results in the area.

Although we have found an initial bound on $f_{\max}(n, \mathcal{L}^w)$ for some equations with more than three variables, we still do not know in general if there are significantly fewer maximal \mathcal{L}^w -free subsets of [n] than there are \mathcal{L}^w -free subsets of [n]. Progress on giving general upper bounds on the number of maximal independent sets in (non-uniform) hypergraphs should (through the method of link hypergraphs) yield results in this direction.

CHAPTER 5

RAMSEY PROPERTIES OF GRAPHS AND THE INTEGERS

5.1 Introduction

In this chapter we use the container method to prove results that correspond to problems concerning tuples of *disjoint* independent sets in hypergraphs. An overarching aim is to demonstrate that with the container method at hand, one can give relatively short and elementary proofs of fundamental results concerning Ramsey properties of graphs and the integers. Moreover, our results give us a precise understanding about how *resiliently* typical graphs and sets of integers of a given density possess a given Ramsey property. In particular, one of our main results is a resilience random Ramsey theorem (Theorem 5.7). This result provides a unified framework for studying both the Ramsey and Turán problems in the setting of random (hyper)graphs. In particular, Theorem 5.7 implies the (so-called 1-statements of the) random Ramsey theorem due to Rödl and Ruciński [90, 91, 92] and the random version of Turán's theorem [28, 108]. Moreover, Theorem 5.7 also resolves a general subcase of the asymmetric random Ramsey conjecture of Kohayakawa and Kreuter [70]. Since Theorem 5.7 unifies and generalises several fundamental results concerning Ramsey and Turán properties of random (hyper)graphs, we survey these topics in Sections 5.1.1.2–5.1.1.4 before we state this result in Section 5.1.1.5.

We also prove a sister result to Theorem 5.7, a resilience strengthening of the random Rado theorem (Theorem 5.11). Again the container method allows us to give a rather short proof of this result. We further provide results on the enumeration of Ramsey graphs (Theorem 5.12) and sets of integers without a given Ramsey property (Theorem 5.13).

As mentioned the results we prove all correspond to problems concerning tuples of disjoint independent sets in hypergraphs. In particular, from the container theorem of Balogh, Morris and Samotij one can easily obtain an analogous result for tuples of independent sets in hypergraphs (see Theorem 5.14). It turns out that many Ramsey-type questions (and other problems) can be naturally phrased in this setting. For example, by Schur's theorem we know that, if n is large, then whenever one r-colours the elements of $[n] := \{1, \ldots, n\}$ there is a monochromatic solution to x + y = z. This raises the question of how large can a subset $S \subseteq [n]$ be whilst failing to have this property? (This problem was first posed back in 1977 by Abbott and Wang [1].) Let H be the hypergraph with vertex set [n] in which edges precisely correspond to solutions to x + y = z. (Note H will have edges of size 2 and 3.) Then sets $S \subseteq [n]$ without this property are precisely the union of r disjoint independent sets in H.

In Section 5.2 we state the container theorem for tuples of independent sets in hypergraphs. In Sections 5.3 and 5.4 we give our applications of this container result to enumeration and resilience questions arising in Ramsey theory for graphs and the integers.

5.1.1 Resilience in hypergraphs and the integers

5.1.1.1 Resilience in graphs

The notion of graph resilience has received significant attention in recent years. Roughly speaking, resilience concerns the question of how 'strongly' a graph G satisfies a certain monotone graph property \mathcal{P} . Global resilience concerns how many edges one can delete and still ensure the resulting graph has property \mathcal{P} whilst local resilience considers how many edges one can delete at each vertex whilst ensuring the resulting graph has property \mathcal{P} . More precisely, we define the global resilience of G with respect to \mathcal{P} , $\operatorname{res}(G, \mathcal{P})$, to be the minimum number t such that by deleting t edges from G, one can obtain a set not having \mathcal{P} . Many classical results in extremal combinatorics can be rephrased in terms of resilience. For example, Turán's theorem determines the global resilience of K_n with respect to the property of containing K_r (where r < n) as a subgraph.

The systematic study of graph resilience was initiated in a paper of Sudakov and Vu [115], though such questions had been studied before this. In particular, a key question in the area is to establish the resilience of various properties of the Erdős–Rényi random graph $G_{n,p}$. (Recall that $G_{n,p}$ has vertex set [n] in which each possible edge is present with probability p, independent of all other choices.) The local resilience of $G_{n,p}$ has been investigated, for example, with respect to Hamiltonicity e.g. [115, 77], almost spanning trees [4] and embedding subgraphs of small bandwidth [20]. See [115] and the surveys [27, 114] for further background on the subject. In this chapter we study the global resilience of $G_{n,p}$ with respect to Ramsey properties (in fact, as we explain later, we will consider its hypergraph analogue $G_{n,p}^{(k)}$ for $k \ge 2$). First we will focus on the graph case.

5.1.1.2 Ramsey properties of random graphs

An event occurs in $G_{n,p}$ with high probability (w.h.p.) if its probability tends to 1 as $n \to \infty$. For many properties \mathcal{P} of $G_{n,p}$, the probability that $G_{n,p}$ has the property

exhibits a phase transition, changing from 0 to 1 over a small interval. That is, there is a threshold for \mathcal{P} : a function $p_0 = p_0(n)$ such that $G_{n,p}$ has \mathcal{P} w.h.p. when $p \gg p_0$ (the 1-statement), while $G_{n,p}$ does not have \mathcal{P} w.h.p. when $p \ll p_0$ (the 0-statement). Indeed, Bollobás and Thomason [19] proved that every monotone property \mathcal{P} has a threshold.

Given a graph H, set $d_2(H) := 0$ if e(H) = 0; $d_2(H) := 1/2$ when H is precisely an edge and define $d_2(H) := (e(H) - 1)/(v(H) - 2)$ otherwise. Then define $m_2(H) := \max_{H' \subseteq H} d_2(H')$ to be the 2-density of H. This graph parameter turns out to be very important when determining the threshold for certain properties in $G_{n,p}$ concerning the containment of a small subgraph H, which we explain further below.

Given $\varepsilon > 0$ and a graph H, we say that a graph G is (H, ε) -Turán if every subgraph of G with at least $(1 - \frac{1}{\chi(H) - 1} + \varepsilon)e(G)$ edges contains a copy of H. Note that the Erdős–Stone theorem implies that K_n is (H, ε) -Turán for any fixed H provided n is sufficiently large. To motivate the definition, consider any graph G. Then by considering a random partition of V(G) into $\chi(H) - 1$ parts (and then removing any edge contained within a part) we see that there is a subgraph G' of G that is $(\chi(H) - 1)$ -partite where $e(G') \ge (1 - \frac{1}{\chi(H) - 1})e(G)$. In particular, $H \not\subseteq G'$. Intuitively speaking, this implies that (up to the ε term), (H, ε) -Turán graphs are those graphs that most strongly contain H.

Rephrasing to the language of resilience, we see that if, for any $\varepsilon > 0$, G is (H, ε) -Turán, then $\operatorname{res}(G, G \supseteq H) = (\frac{1}{\chi(H)-1} \pm \varepsilon)e(G)$, and vice versa. (Recall that we write $x = a \pm b$ to say that the value of x is some real number in the interval [a - b, a + b].) The global resilience of $G_{n,p}$ with respect to the Turán problem has been extensively studied. Indeed, a recent trend in combinatorics and probability concerns so-called *sparse random* analogues of extremal theorems (see [27]), and determining when $G_{n,p}$ is (H, ε) -Turán is an example of such a result.

If $p \leq cn^{-1/m_2(H)}$ for some small constant c, then it is not hard to show that w.h.p. $G_{n,p}$ is not (H, ε) -Turán. In [58, 59, 72] it was conjectured that w.h.p. $G_{n,p}$ is (H, ε) -Turán provided that $p \ge Cn^{-1/m_2(H)}$, where C is a (large) constant. After a number of partial results, this conjecture was confirmed by Schacht [108] and (in the case when H is strictly 2-balanced, i.e. $m_2(H') < m_2(H)$ for all $H' \subset H$) by Conlon and Gowers [28].

Theorem 5.1 ([108, 28]). For any graph H with $\Delta(H) \ge 2$ and any $\varepsilon > 0$, there are positive constants c, C such that

$$\lim_{n \to \infty} \mathbb{P}[G_{n,p} \text{ is } (H,\varepsilon) \text{-} \operatorname{Tur}\acute{a}n] = \begin{cases} 0 & \text{ if } p < cn^{-1/m_2(H)}; \\ 1 & \text{ if } p > Cn^{-1/m_2(H)} \end{cases}$$

Given an integer r, an r-colouring of a graph G is a function $\sigma : E(G) \to [r]$. (So this is not necessarily a proper colouring.) We say that G is (H, r)-Ramsey if every r-colouring of G yields a monochromatic copy of H in G. Observe that being (H, 1)-Ramsey is the same as containing H as a subgraph. So the 1-statement of Theorem 5.1 says that, given $\varepsilon > 0$, there exists a positive constant C such that, if $p > Cn^{-1/m_2(H)}$, then

$$\lim_{n \to \infty} \mathbb{P}\left[\frac{\operatorname{res}(G_{n,p}, (H, 1)\text{-Ramsey})}{e(G_{n,p})} = \frac{1}{\chi(H) - 1} \pm \varepsilon\right] = 1.$$
(5.1.1)

The following result of Rödl and Ruciński [90, 91, 92] yields a random version of Ramsey's theorem.

Theorem 5.2 ([90, 91, 92]). Let $r \ge 2$ be a positive integer and let H be a graph that is not a forest consisting of stars and paths of length 3. There are positive constants c, Csuch that

$$\lim_{n \to \infty} \mathbb{P}[G_{n,p} \text{ is } (H,r)\text{-}Ramsey] = \begin{cases} 0 & \text{ if } p < cn^{-1/m_2(H)}; \\ 1 & \text{ if } p > Cn^{-1/m_2(H)}. \end{cases}$$

Thus $n^{-1/m_2(H)}$ is again the threshold for the (H, r)-Ramsey property. Let us provide some intuition as to why. The expected number of copies of H in $G_{n,p}$ is $\Theta(n^{v(H)}p^{e(H)})$, while the expected number of edges in $G_{n,p}$ is $\Theta(pn^2)$. When $p = \Theta(n^{-1/d_2(H)})$, these quantities agree up to a constant. Suppose that H is 2-balanced, i.e. $d_2(H) = m_2(H)$. For small c > 0, when $p < cn^{-1/m_2(H)}$, most copies of H in $G_{n,p}$ contain an edge which appears in no other copy. Thus we can hope to colour these special edges blue and colour the remaining edges red to eliminate all monochromatic copies of H. For large C > 0, most edges lie in many copies of H, so the copies of H are highly overlapping and we cannot avoid monochromatic copies. In general, when H is not necessarily 2-balanced, the threshold is $n^{-1/d_2(H')}$ for the 'densest' subgraph H' of H since, roughly speaking, the appearance of H is governed by the appearance of its densest part.

We remark that Nenadov and Steger [85] recently gave a short proof of Theorem 5.2 using the container method.

5.1.1.3 Asymmetric Ramsey properties in random graphs

It is natural to ask for an *asymmetric* analogue of Theorem 5.2. Now, for graphs H_1, \ldots, H_r , a graph G is (H_1, \ldots, H_r) -Ramsey if for any r-colouring of G there is a copy of H_i in colour i for some $i \in [r]$. (This definition coincides with that of (H, r)-Ramsey when $H_1 = \ldots = H_r = H$.) Kohayakawa and Kreuter [70] conjectured an analogue of Theorem 5.2 in the asymmetric case. To state it, we need to introduce the *asymmetric density* of H_1, H_2 where $m_2(H_1) \ge m_2(H_2)$ via

$$m_2(H_1, H_2) := \max\left\{\frac{e(H_1')}{v(H_1') - 2 + 1/m_2(H_2)} : H_1' \subseteq H_1 \text{ and } e(H_1') \ge 1\right\}.$$
 (5.1.2)

Conjecture 5.3 ([70]). For any graphs H_1, \ldots, H_r with $m_2(H_1) \ge \ldots \ge m_2(H_r) > 1$, there are positive constants c, C > 0 such that

$$\lim_{n \to \infty} \mathbb{P} \left[G_{n,p} \text{ is } (H_1, \dots, H_r) \text{-}Ramsey \right] = \begin{cases} 0 & \text{ if } p < cn^{-1/m_2(H_1, H_2)}; \\ 1 & \text{ if } p > Cn^{-1/m_2(H_1, H_2)} \end{cases}$$

So the conjectured threshold only depends on the 'joint density' of the densest two graphs H_1, H_2 . The intuition for this threshold is discussed in detail e.g. in Section 1.1 in [51]. One can show that $m_2(H_1) \ge m_2(H_1, H_2) \ge m_2(H_2)$ with equality if and only if $m_2(H_1) = m_2(H_2)$. Thus Conjecture 5.3 would generalise Theorem 5.2. Kohayakawa and Kreuter [70] have confirmed Conjecture 5.3 when the H_i are cycles. In [81] it was observed that the approach used by Kohayakawa and Kreuter [70] implies the 1-statement of Conjecture 5.3 holds when H_1 is strictly 2-balanced *provided* the so-called KLR conjecture holds. This latter conjecture was proven by Balogh, Morris and Samotij [8] thereby proving the 1-statement of Conjecture 5.3 holds in this case. The full 1-statement of Conjecture 5.3 has very recently been proven by Mousset, Nenadov and Samotij [83].

5.1.1.4 Ramsey properties of random hypergraphs

Consider now the k-uniform analogue $G_{n,p}^{(k)}$ of $G_{n,p}$ which has vertex set [n] and in which every k-element subset of [n] appears as an edge with probability p, independent of all other choices. Here, we wish to obtain analogues of Theorems 5.1, 5.2 and Conjecture 5.3 by determining the threshold for being (H, ε) -Turán, (H, r)-Ramsey, and more generally being (H_1, \ldots, H_r) -Ramsey. The definitions of (H, r)-Ramsey and (H_1, \ldots, H_r) -Ramsey extend from graphs in the obvious way. Given a k-uniform hypergraph H, let ex(n; H)be the maximum size of an n-vertex H-free hypergraph. A simple averaging argument shows that the limit

$$\pi(H) := \lim_{n \to \infty} \frac{\operatorname{ex}(n; H)}{\binom{n}{k}}$$

exists. Now we say that a k-uniform hypergraph G is (H, ε) -Turán if every subhypergraph of G with at least $(\pi(H) + \varepsilon)e(G)$ edges contains a copy of H. (Since $\pi(H) = 1 - \frac{1}{\chi(H)-1}$ when k = 2, this generalises the definition we gave earlier.) We also need to generalise the notion of 2-density to k-density: Given a k-graph H, define

$$d_k(H) := \begin{cases} 0 & \text{if } e(H) = 0; \\ 1/k & \text{if } v(H) = k \text{ and } e(H) = 1; \\ \frac{e(H) - 1}{v(H) - k} & \text{otherwise,} \end{cases}$$

and let

$$m_k(H) := \max_{H' \subseteq H} d_k(H').$$

The techniques of Conlon–Gowers [28] and of Schacht [108] actually extended to a proof of a version of Theorem 5.1 for hypergraphs:

Theorem 5.4 ([28, 108]). For any k-uniform hypergraph H with maximum vertex degree at least two and any $\varepsilon > 0$, there are positive constants c, C such that

$$\lim_{n \to \infty} \mathbb{P}[G_{n,p}^{(k)} \text{ is } (H,\varepsilon) \text{-} Turán] = \begin{cases} 0 & \text{ if } p < cn^{-1/m_k(H)}; \\ 1 & \text{ if } p > Cn^{-1/m_k(H)} \end{cases}$$

The 1-statement of Theorem 5.2 was generalised to hypergraphs by Friedgut, Rödl and Schacht [42] and by Conlon and Gowers [28], proving a conjecture of Rödl and Ruciński [94]. (The special cases of the complete 3-uniform hypergraph $K_4^{(3)}$ on four vertices and of k-partite k-uniform hypergraphs were already proved in [94], [95] respectively. Also in [85] Nenadov and Steger remark that their proof of the 1-statement of Theorem 5.2 extends to Theorem 5.5.)

Theorem 5.5 ([28, 42]). Let $r, k \ge 2$ be integers and let H be a k-uniform hypergraph with maximum vertex degree at least two. There is a positive constant C such that

$$\lim_{n \to \infty} \mathbb{P}[G_{n,p}^{(k)} \text{ is } (H,r)\text{-}Ramsey] = 1 \quad \text{ if } p > Cn^{-1/m_k(H)}.$$

In [51], sufficient conditions are given for a corresponding 0-statement. However, the authors further show that, for $k \ge 4$, there is a k-uniform hypergraph H such that the threshold for $G_{n,p}^{(k)}$ to be (H,r)-Ramsey is not $n^{-1/m_k(H)}$, and nor does it correspond to the exceptional case in the graph setting of certain forests, where there is a coarse threshold due to the appearance of small subgraphs.

For the asymmetric Ramsey problem, we need to suitably generalise (5.1.2), in the obvious way: for any k-uniform hypergraphs H_1, H_2 with non-empty edge sets and $m_k(H_1) \ge m_k(H_2)$, let

$$m_k(H_1, H_2) := \max\left\{\frac{e(H_1')}{v(H_1') - k + 1/m_k(H_2)} : H_1' \subseteq H_1 \text{ and } e(H_1') \ge 1\right\}$$
(5.1.3)

be the asymmetric k-density of (H_1, H_2) . Again,

$$m_k(H_1) \ge m_k(H_1, H_2) \ge m_k(H_2),$$

so, in particular, $m_k(H_1, H_2) = m_k(H_1)$ if and only if H_1 and H_2 have the same k-density.

Recently, Gugelmann, Nenadov, Person, Steger, Škorić and Thomas [51] generalised the 1-statement of Conjecture 5.3 to k-uniform hypergraphs, in the case when $H'_1 = H_1$ is the unique maximiser in (5.1.3), i.e. H_1 is strictly k-balanced with respect to $m_k(\cdot, H_2)$.

Theorem 5.6 ([51]). For all positive integers r, k with $k \ge 2$ and k-uniform hypergraphs H_1, \ldots, H_r with $m_k(H_1) \ge \ldots \ge m_k(H_r)$ where H_1 is strictly k-balanced with respect to $m_k(\cdot, H_2)$, there exists C > 0 such that

$$\lim_{n \to \infty} \mathbb{P} \left[G_{n,p}^{(k)} \text{ is } (H_1, \dots, H_r) \text{-}Ramsey \right] = 1 \quad \text{if } p > Cn^{-1/m_k(H_1, H_2)}.$$

They further prove a version of Theorem 5.6 with the weaker bound $p > Cn^{-1/m_k(H_1,H_2)} \log n$ when H_1 is not required to be strictly k-balanced with respect to $m_k(\cdot, H_2)$.

5.1.1.5 New resilience result

Our main result here is Theorem 5.7, which generalises, fully and partially, all of the 1-statements of the results discussed in this section, giving a unified setting for both the random Ramsey theorem and the random Turán theorem. Once we have obtained a container theorem for Ramsey graphs (Theorem 5.34), the proof is short (see Section 5.4.6).

For k-uniform hypergraphs H_1, \ldots, H_r and a positive integer n, let $ex^r(n; H_1, \ldots, H_r)$ be the maximum size of an n-vertex k-uniform hypergraph G which is not (H_1, \ldots, H_r) -Ramsey. Define the r-coloured Turán density

$$\pi(H_1, \dots, H_r) := \lim_{n \to \infty} \frac{\exp^r(n; H_1, \dots, H_r)}{\binom{n}{k}}.$$
 (5.1.4)

Observe that $ex^1(n; H) = ex(n; H)$ since a hypergraph is *H*-free if and only if it is not (H, 1)-Ramsey. Note further that $\pi(\cdot, \ldots, \cdot)$ generalises $\pi(\cdot)$. So when k = 2, we have $\pi(H) = 1 - \frac{1}{\chi(H) - 1}$. We will show in Section 5.4.2 that the limit in (5.1.4) does indeed exist, so $\pi(\cdot, \ldots, \cdot)$ is well-defined. Further, crucially for *k*-uniform hypergraphs H_1, \ldots, H_r , there exists an $\varepsilon = \varepsilon(H_1, \ldots, H_r) > 0$ so that $\pi(H_1, \ldots, H_r) < 1 - \varepsilon$ (see (5.4.3) in Section 5.4.2).

Theorem 5.7 (Resilience for random Ramsey). Let $\delta > 0$, let r, k be positive integers with $k \ge 2$ and let H_1, \ldots, H_r be k-uniform hypergraphs each with maximum vertex degree at least two, and such that $m_k(H_1) \ge \ldots \ge m_k(H_r)$. There exists C > 0 such that if $p > Cn^{-1/m_k(H_1)}$, then

$$\lim_{n \to \infty} \mathbb{P}\left[\frac{\operatorname{res}\left(G_{n,p}^{(k)}, (H_1, \dots, H_r) - Ramsey\right)}{e\left(G_{n,p}^{(k)}\right)} = 1 - \pi(H_1, \dots, H_r) \pm \delta\right] = 1.$$

Thus, when $p > Cn^{-1/m_k(H_1)}$, the random hypergraph $G_{n,p}^{(k)}$ is w.h.p such that every subhypergraph G' with at least a $\pi(H_1, \ldots, H_r) + \Omega(1)$ fraction of the edges is (H_1, \ldots, H_r) - Ramsey. Conversely, there is a subgraph of $G_{n,p}^{(k)}$ whose edge density is slightly smaller than this which *does not* have the Ramsey property.

Note that the threshold of $p > Cn^{-1/m_k(H_1)}$ in Theorem 5.7 is tight up to the multiplicative constant C. Indeed, consider the random hypergraph $G_{n,p}^{(k)}$ with $p \ll n^{-1/m_k(H_1)}$. Let $H'_1 \subseteq H_1$ be such that $m_k(H_1) = d_k(H'_1)$. Then the expected number of copies of H'_1 in $G_{n,p}^{(k)}$ is much smaller than the expected number of edges in $G_{n,p}^{(k)}$, so w.h.p. we can delete every copy of H'_1 (and therefore H_1) by removing $o(e(G_{n,p}^{(k)}))$ edges. So the hypergraph Gthat remains has $(1 - o(1))e(G_{n,p}^{(k)})$ edges, and is not (H_1, \ldots, H_r) -Ramsey because we can colour every edge of G with colour 1. Then, since G is H_1 -free, there is no copy of H_i in colour i in G.

Let us describe the importance of Theorem 5.7 (in the case k = 2 and $H_1 = \ldots = H_r = H$) in conjunction with Theorem 5.2. The 0-statement of Theorem 5.2 says that a typical sparse graph, i.e. one with density at most $cn^{2-1/m_2(H)}$, is not (H, r)-Ramsey. On the other hand, by Theorem 5.7, a typical dense graph, i.e. one with density at least $Cn^{2-1/m_2(H)}$, has the Ramsey property in a sense which is as strong as possible with respect to subgraphs: every sufficiently dense subgraph is (H, r)-Ramsey, and this minimum density is the largest we could hope to require.

The relationship between Theorem 5.7 and the previous results stated in this section can be summarised as follows:

- The 1-statement of Theorem 5.1 is recovered when k = 2 and r = 1. This follows from (5.1.1) and the relation between $\pi(H)$ and $\chi(H)$.
- In the case k = 2 and $H_1 = \ldots = H_r = H$, we obtain a stronger statement in place of the 1-statement of Theorem 5.2 as described above.
- Theorem 5.7 proves the 1-statement of Conjecture 5.3 in the case when $m_2(H_1) = m_2(H_2)$ in the same stronger sense as above.

- The 1-statement of Theorem 5.4 is recovered when r = 1.
- Theorem 5.7 implies Theorem 5.5, yielding a resilience version of this result.
- Theorem 5.7 implies a version of Theorem 5.6 when $m_k(H_1) = m_k(H_2)$ but now H_1 is not required to be strictly k-balanced with respect to $m_k(\cdot, H_2)$.

Note that even though Theorem 5.7 implies many of the known results concerning Ramsey properties of random (hyper)graphs, often the resilience random Ramsey problem is different to the random Ramsey problem. In particular, we have determined the threshold for the former problem, whilst we have seen above examples of (hyper)graphs H_1, \ldots, H_r where a lower value of p still ensures that $G_{n,p}^{(k)}$ is w.h.p. (H_1, \ldots, H_r) -Ramsey.

5.1.1.6 Resilience in the integers

An important branch of Ramsey theory concerns partition properties of sets of integers. Schur's classical theorem [110] states that if \mathbb{N} is *r*-coloured there exists a monochromatic solution to x + y = z; later van der Waerden [120] showed that the same hypothesis ensures a monochromatic arithmetic progression of arbitrary length. More generally, Rado's theorem [88] characterises all those systems of homogeneous linear equations \mathcal{L} for which every finite colouring of \mathbb{N} yields a monochromatic solution to \mathcal{L} .

As in the graph case, there has been interest in proving random analogues of such results from arithmetic Ramsey theory.

Before we describe the background of this area we will introduce some notation and definitions. As mentioned in the introduction of this thesis, we will assume A is an $\ell \times k$ integer matrix where $k \ge \ell$ of full rank ℓ and b is an integer vector of dimension ℓ . We will let $\mathcal{L}(A, b)$ denote the associated system of linear equations Ax = b, noting that for brevity we will simply write \mathcal{L} if it is clear from the context which matrix A and vector b it refers to. If A is stated but b is not, then we assume \mathcal{L} refers to the system of linear equations Ax = 0. Let S be a set of integers. Recall that if a vector $x = (x_1, \ldots, x_k) \in S^k$ satisfies Ax = b(i.e. it is a solution to \mathcal{L}) and the x_i are distinct we call x a k-distinct solution to \mathcal{L} in S. Throughout this chapter, with the exception of Sections 5.3.6 and 5.3.7, we consider \mathcal{L}^d -free sets.

We generalise the definition of \mathcal{L}^d -free to r colours in the obvious way: we call a set S of integers (\mathcal{L}^d, r) -free if there exists an r-colouring of S such that it contains no monochromatic k-distinct solution to \mathcal{L} . Otherwise we will call S (\mathcal{L}^d, r) -Rado. In the case when r = 1, we write \mathcal{L}^d -free instead of $(\mathcal{L}^d, 1)$ -free. Define $\mu(n, \mathcal{L}^d, r)$ to be the size of the largest (\mathcal{L}^d, r) -free subset of [n].

Recall from the introduction the definitions of irredundant, partition regular, and m(A). As mentioned above Rado's theorem [88] characterises all partition regular systems of linear equations \mathcal{L} . The study of random versions of Rado's theorem has focused on irredundant partition regular matrices. This is natural since for every redundant $\ell \times k$ matrix A there exists an irredundant $\ell' \times k'$ matrix A' for some $\ell' < \ell$ and k' < k with the same family of solutions (viewed as sets). See [93, Section 1] for a full explanation. Also note that most historical results concern distinct \mathcal{L} -free sets; many methods (including that of this chapter) require the use of a k-uniform hypergraph, so it is easier to work with k-distinct solutions only. However since there are an insignificant number of non k-distinct solutions compared to k-distinct solutions, a removal lemma such as Lemma 2.2 implies that it makes little difference to restrict to only k-distinct solutions.

Another class of matrices that have received attention in relation to this problem are so-called density regular matrices: An irredundant, partition regular matrix A is *density regular* if any subset $F \subseteq \mathbb{N}$ with positive upper density, i.e.,

$$\limsup_{n \to \infty} \frac{|F \cap [n]|}{n} > 0,$$

contains a k-distinct solution to \mathcal{L} .

We now describe some random analogues of results from arithmetic Ramsey theory. Recall that $[n]_p$ denotes a set where each element $a \in [n]$ is included with probability pindependently of all other elements. Rödl and Ruciński [93] showed that for irredundant partition regular matrices A, m(A) is an important parameter for determining whether $[n]_p$ is (\mathcal{L}^d, r) -Rado or (\mathcal{L}^d, r) -free.

Theorem 5.8 ([93]). For all irredundant partition regular full rank matrices A and all positive integers $r \ge 2$, there exists a constant c > 0 such that

$$\lim_{n \to \infty} \mathbb{P}\left[[n]_p \text{ is } (\mathcal{L}^d, r) \text{-} Rado \right] = 0 \quad \text{ if } p < cn^{-1/m(A)}.$$

We remark it is important that $r \ge 2$ in Theorem 5.8. That is, the corresponding statement for r = 1 is not true in general. Roughly speaking, Theorem 5.8 implies that almost all subsets of [n] with significantly fewer than $n^{1-1/m(A)}$ elements are (\mathcal{L}^d, r) -free for any irredundant partition regular matrix A. The following theorem of Friedgut, Rödl and Schacht [42] complements this result, implying that almost all subsets of [n] with significantly more than $n^{1-1/m(A)}$ elements are (\mathcal{L}^d, r) -Rado for any irredundant partition regular matrix A.

Theorem 5.9 ([42]). For all irredundant partition regular full rank matrices A and all positive integers r, there exists a constant C > 0 such that

$$\lim_{n \to \infty} \mathbb{P}\left[[n]_p \text{ is } (\mathcal{L}^d, r) \text{-} Rado \right] = 1 \quad \text{if } p > Cn^{-1/m(A)}.$$

Earlier, Theorem 5.9 was confirmed by Graham, Rödl and Ruciński [46] in the case where \mathcal{L} is x + y = z and r = 2, and then by Rödl and Ruciński [93] in the case when A is density regular.

Together Theorems 5.8 and 5.9 show that the threshold for the property of being (\mathcal{L}^d, r) -Rado is $p = n^{-1/m(A)}$. In light of this, it is interesting to ask if above this threshold the property of being (\mathcal{L}^d, r) -Rado is resilient to the deletion of a significant number of elements. To be precise, given a set S, we define the *resilience of* S with respect to \mathcal{P} , $\operatorname{res}(S, \mathcal{P})$, to be the minimum number t such that by deleting t elements from S, one can obtain a set not having \mathcal{P} . For example, when \mathcal{P} is the property of containing an arithmetic progression of length k, then Szemerédi's theorem can be phrased in terms of resilience; it states that for all $k \geq 3$ and $\varepsilon > 0$, there exists $n_0 > 0$ such that for all integers $n \geq n_0$, we have $\operatorname{res}([n], \mathcal{P}) \geq (1 - \varepsilon)n$.

The following result of Schacht [108] provides a resilience strengthening of Theorem 5.9 in the case of density regular matrices.

Theorem 5.10 ([108]). For all irredundant density regular full rank matrices A, all positive integers r and all $\varepsilon > 0$, there exists a constant C > 0 such that

$$\lim_{n \to \infty} \mathbb{P}\left[\frac{\operatorname{res}([n]_p, (\mathcal{L}^d, r) - Rado)}{|[n]_p|} \ge 1 - \varepsilon\right] = 1 \quad \text{if } p > Cn^{-1/m(A)}.$$

Note that in [108] the result is stated in the r = 1 case only, but the general result follows immediately from this special case.

Our next result gives a resilience strengthening of Theorem 5.9 for all irredundant partition regular matrices.

Theorem 5.11. For all irredundant partition regular full rank matrices A, all positive integers r and all $\delta > 0$, there exists a constant C > 0 such that

$$\lim_{n \to \infty} \mathbb{P}\left[\frac{\operatorname{res}([n]_p, (\mathcal{L}^d, r) - Rado)}{|[n]_p|} = 1 - \frac{\mu(n, \mathcal{L}^d, r)}{n} \pm \delta\right] = 1 \quad \text{if } p > Cn^{-1/m(A)}$$

It is well known that for all irredundant partition regular full rank matrices A and all

positive integers r, there exist $n_0 = n_0(A, r)$, $\eta = \eta(A, r) > 0$, such that for all integers $n \ge n_0$, we have $\mu(n, \mathcal{L}^d, r) \le (1 - \eta)n$. (This follows from a supersaturation lemma of Frankl, Graham and Rödl [41, Theorem 1].) Thus, Theorem 5.11 does imply Theorem 5.9. Further, in the case when A is density regular, [41, Theorem 2] immediately implies that $\mu(n, \mathcal{L}^d, r) = o(n)$ for any fixed $r \in \mathbb{N}$. Thus Theorem 5.11 implies Theorem 5.10. Theorem 5.11 in the case when r = 1 and \mathcal{L} is x + y = z was proved by Schacht [108]. In fact, the method of Schacht can be used to prove the theorem for r = 1 and every irredundant partition regular matrix A.

Intuitively, the reader can interpret Theorem 5.11 as stating that almost all subsets of [n] with significantly more than $n^{1-1/m(A)}$ elements strongly possess the property of being (\mathcal{L}^d, r) -Rado for any irredundant partition regular matrix A. The 'strength' here depends on the parameter $\mu(n, \mathcal{L}^d, r)$. In light of this it is natural to seek good bounds on $\mu(n, \mathcal{L}^d, r)$ (particularly in the cases when $\mu(n, \mathcal{L}^d, r) = \Omega(n)$). In general, not too much is known about this parameter. However, as mentioned earlier, in the case when A = (1, 1, -1) (i.e. \mathcal{L} is x + y = z), this is (by replacing distinct \mathcal{L} -free with strongly \mathcal{L} -free) a 40-year-old problem of Abbott and Wang [1]. In Section 5.3.6 we give an upper bound on $\mu(n, \mathcal{L}^s, r)$ in this case for all $r \in \mathbb{N}$.

Instead of proving Theorem 5.11 directly, in Section 5.3 we will prove a version of the result that holds for a more general class of matrices A, does not assume b = 0, and also deals with the asymmetric case, namely Theorem 5.16. In [112], Spiegel proved the case r = 1 of Theorem 5.16 and used the container method to give an alternative proof of Theorem 5.9.

5.1.2 Enumeration questions for Ramsey problems

A fundamental question in combinatorics is to determine the number of structures with a given property. For example, Erdős, Frankl and Rödl [34] proved that the number of *n*-vertex *H*-free graphs is $2^{\binom{n}{2}(1-\frac{1}{r-1}+o(1))}$ for any graph *H* of chromatic number *r*. Here the lower bound follows by considering all the subgraphs of the (r-1)-partite Turán graph. Given any $k, r, n \in \mathbb{N}$ with $k \ge 2$ and *k*-uniform hypergraphs H_1, \ldots, H_r , define $\operatorname{Ram}(n; H_1, \ldots, H_r)$ to be the collection of all *k*-uniform hypergraphs on vertex set [n] that are (H_1, \ldots, H_r) -Ramsey and $\operatorname{Ram}(n; H_1, \ldots, H_r)$ to be all those *k*-uniform hypergraphs on [n] that are not (H_1, \ldots, H_r) -Ramsey. A natural question is to determine the size of $\operatorname{Ram}(n; H_1, \ldots, H_r)$. Surprisingly, we are unaware of any *explicit* results in this direction for $r \ge 2$. The next application of the container method fully answers this question up to an error term in the exponent.

Theorem 5.12. Let $k, r, n \in \mathbb{N}$ with $k \ge 2$ and H_1, \ldots, H_r be k-uniform hypergraphs. Then

$$|\overline{\operatorname{Ram}}(n; H_1, \dots, H_r)| = 2^{\operatorname{ex}^r(n, H_1, \dots, H_r) + o(n^k)} = 2^{\pi(H_1, \dots, H_r)\binom{n}{k} + o(n^k)}$$

Note that in the case when k = 2 and r = 1, Theorem 5.12 is precisely the above mentioned result of Erdős, Frankl and Rödl [34]. In fact, one can also obtain Theorem 5.12 by using the work from [84], a hypergraph analogue of the result in [34]; see Section 5.4.5 for a proof of this. Similar results were obtained also using containers by Falgas-Ravry, O'Connell and Uzzell in [40], and by Terry in [118] who reproved a result of Ishigami [63].

Our final application of the container method determines, up to an error term in the exponent, the number of (\mathcal{L}^d, r) -free subsets of [n].

Theorem 5.13. Let A be an irredundant partition regular matrix of full rank and let $r \in \mathbb{N}$ be fixed. There are $2^{\mu(n,\mathcal{L}^d,r)+o(n)}$ (\mathcal{L}^d,r) -free subsets of [n].

As an illustration, a result of Hu [61] implies that $\mu(n, \mathcal{L}^d, 2) = 4n/5 + o(n)$ in the case when \mathcal{L} is x + y = z. Thus, Theorem 5.13 tells us all but $2^{(4/5+o(1))n}$ subsets of [n]are $(\mathcal{L}^d, 2)$ -Rado in this case. Related results (in the 1-colour case) were obtained by Green [48] and Saxton and Thomason [107].

5.2 Container results for disjoint independent sets

First recall from Chapter 2 the general hypergraph container theorem of Balogh, Morris and Samotij [8].

Theorem 2.1 ([8]). For every $k \in \mathbb{N}$ and all positive c and ε , there exists a positive constant C such that the following holds. Let \mathcal{H} be a k-uniform hypergraph and let $\mathcal{F} \subseteq$ $\mathcal{P}(V(\mathcal{H}))$ be an increasing family of sets such that $|A| \ge \varepsilon v(\mathcal{H})$ for all $A \in \mathcal{F}$. Suppose that \mathcal{H} is $(\mathcal{F}, \varepsilon)$ -dense and $p \in (0, 1)$ is such that, for every $\ell \in [k]$,

$$\Delta_{\ell}(\mathcal{H}) \leqslant c \cdot p^{\ell-1} \frac{e(\mathcal{H})}{v(\mathcal{H})}.$$

Then there exists a family $S \subseteq \binom{V(\mathcal{H})}{\leqslant C_P \cdot v(\mathcal{H})}$ and functions $f : S \to \overline{\mathcal{F}}$ and $g : \mathcal{I}(\mathcal{H}) \to S$ such that for every $I \in \mathcal{I}(\mathcal{H})$, we have that $g(I) \subseteq I$ and $I \setminus g(I) \subseteq f(g(I))$.

Throughout the chapter, when we consider r-tuples of sets, the r-tuples are always ordered. For two r-tuples of sets (I_1, \ldots, I_r) and (J_1, \ldots, J_r) we write $(I_1, \ldots, I_r) \subseteq$ (J_1, \ldots, J_r) if $I_x \subseteq J_x$ for each $x \in [r]$. We write $(I_1, \ldots, I_r) \cup (J_1, \ldots, J_r) := (I_1 \cup$ $J_1, \ldots, I_r \cup J_r)$.

We write ij to denote the pair $\{i, j\}$. For a hypergraph \mathcal{H} define

$$\mathcal{I}_r(\mathcal{H}) := \left\{ (I_1, \dots, I_r) \in \mathcal{P}(V(\mathcal{H}))^r : I_x \in \mathcal{I}(\mathcal{H}) \text{ and } I_i \cap I_j = \emptyset \text{ for all } x \in [r], ij \in \binom{[r]}{2} \right\}.$$

Whereas Theorem 2.1 provides a set of containers for the independent sets of a hypergraph, the following theorem is an analogous result for the r-tuples of disjoint independent sets of a hypergraph. It is a straightforward consequence of Theorem 2.1.

Theorem 5.14. For every $k, r \in \mathbb{N}$ and all positive c and ε , there exists a positive constant C such that the following holds. Let \mathcal{H} be a k-uniform hypergraph and let

 $\mathcal{F} \subseteq \mathcal{P}(V(\mathcal{H}))$ be an increasing family of sets such that $|A| \ge \varepsilon v(\mathcal{H})$ for all $A \in \mathcal{F}$. Suppose that \mathcal{H} is $(\mathcal{F}, \varepsilon)$ -dense and $p \in (0, 1)$ is such that, for every $\ell \in [k]$,

$$\Delta_{\ell}(\mathcal{H}) \leqslant c \cdot p^{\ell-1} \frac{e(\mathcal{H})}{v(\mathcal{H})}$$

Then there exists a family $S_r \subseteq \mathcal{P}(V(\mathcal{H}))^r$ and functions $f : S_r \to (\overline{\mathcal{F}})^r$ and $g : \mathcal{I}_r(\mathcal{H}) \to S_r$ such that the following conditions hold:

- (i) If $(S_1, \ldots, S_r) \in \mathcal{S}_r$ then $\sum |S_i| \leq Cp \cdot v(\mathcal{H})$;
- (ii) every $S \in \mathcal{S}_r$ satisfies $S \in \mathcal{I}_r(\mathcal{H})$;
- (iii) for every $(I_1, \ldots, I_r) \in \mathcal{I}_r(\mathcal{H})$, we have that $S \subseteq (I_1, \ldots, I_r) \subseteq S \cup f(S)$ where $S := g(I_1, \ldots, I_r).$

Proof. Apply Theorem 2.1 with k, c, ε to obtain a positive constant C_1 . Let $C := rC_1$. We will show that C has the required properties. Let \mathcal{H} be a k-uniform hypergraph which together with a set $\mathcal{F} \subseteq \mathcal{P}(V(\mathcal{H}))$ satisfies the hypotheses of Theorem 5.14. Since \mathcal{H} , \mathcal{F} also satisfy the hypotheses of Theorem 2.1, there exists a family $\mathcal{S} \subseteq \binom{V(\mathcal{H})}{\leq C_1 p \cdot v(\mathcal{H})}$ and functions $f' : \mathcal{S} \to \overline{\mathcal{F}}$ and $g' : \mathcal{I}(\mathcal{H}) \to \mathcal{S}$ such that for every $I \in \mathcal{I}(\mathcal{H})$ we have $g'(I) \subseteq I$ and $I \setminus g'(I) \subseteq f'(g'(I))$. Define

$$\mathcal{S}' := \{ S \in \mathcal{S} : \text{ there exists } I \in \mathcal{I}(\mathcal{H}) \text{ such that } g'(I) = S \},\$$

and

$$\mathcal{S}_r := \left\{ (S_1, \dots, S_r) \in \mathcal{P}(V(\mathcal{H}))^r : S_x \in \mathcal{S}' \text{ and } S_i \cap S_j = \emptyset \text{ for all } x \in [r], ij \in \binom{[r]}{2} \right\}.$$

Let $(S_1, \ldots, S_r) \in \mathcal{S}_r$. First note that

$$\sum_{x \in [r]} |S_x| \leqslant C_1 r \cdot pv(\mathcal{H}) = Cp \cdot v(\mathcal{H}),$$

so (i) holds. Also since $S_x \in \mathcal{S}'$ for all $x \in [r]$, we have $S_x \in \mathcal{I}(\mathcal{H})$ and so by definition of \mathcal{S}_r we have $\mathcal{S}_r \subseteq \mathcal{I}_r(\mathcal{H})$ proving (ii).

Consider any $(S_1, \ldots, S_r) \in S_r$ and any $(I_1, \ldots, I_r) \in \mathcal{I}_r(\mathcal{H})$. Define $f : S_r \to (\overline{\mathcal{F}})^r$ by setting $f(S_1, \ldots, S_r) := (f'(S_1), \ldots, f'(S_r))$ and define $g : \mathcal{I}_r(\mathcal{H}) \to S_r$ by setting $g(I_1, \ldots, I_r) := (g'(I_1), \ldots, g'(I_r)).$

Note that since $f'(S_x) \in \overline{\mathcal{F}}$, $g'(I_x) \in \mathcal{S}'$ and $g'(I_i) \cap g'(I_j) = \emptyset$ for all $x \in [r]$ and $ij \in {[r] \choose 2}$, we do indeed have $(f'(S_1), \ldots, f'(S_r)) \in (\overline{\mathcal{F}})^r$ and $(g'(I_1), \ldots, g'(I_r)) \in \mathcal{S}_r$.

Now for (iii), since $g'(I_x) \subseteq I_x$ and $I_x \setminus g'(I_x) \subseteq f'(g'(I_x))$ for all $x \in [r]$, we have $g(I_1, \ldots, I_r) = (g'(I_1), \ldots, g'(I_r)) \subseteq (I_1, \ldots, I_r)$. Now since $f(g(I_1, \ldots, I_r)) = (f'(g'(I_1)), \ldots, f'(g'(I_r)))$ we also have $(I_1, \ldots, I_r) \subseteq f(g(I_1, \ldots, I_r)) \cup g(I_1, \ldots, I_r)$ as required.

In all of our applications of the container method, we will in fact apply the following asymmetric version of Theorem 5.14. In particular, in the proof of e.g. Theorem 5.7, instead of considering tuples of disjoint independent sets from the *same* hypergraph \mathcal{H} , we are actually concerned with disjoint independent sets from *different* hypergraphs but which have the *same vertex set*: For all $i \in [r]$, let \mathcal{H}_i be a k_i -uniform hypergraph, each on the same vertex set V, and define $\mathcal{I}(\mathcal{H}_1, \ldots, \mathcal{H}_r)$ to be the set of all r-tuples $(I_1, \ldots, I_r) \in \prod_{i \in [r]} \mathcal{I}(\mathcal{H}_i)$ such that $I_i \cap I_j = \emptyset$ for all $1 \leq i < j \leq r$.

We omit the proof of Theorem 5.15 since it follows from Theorem 2.1 as in the proof of Theorem 5.14.

Theorem 5.15. For every $r, k_1, \ldots, k_r \in \mathbb{N}$ with $k_i \ge 2$ for all $i \in [r]$, and all $c, \varepsilon > 0$, there exists a positive constant C such that the following holds. For all $i \in [r]$, let \mathcal{H}_i be a k_i -uniform hypergraph, each on the same vertex set V. For all $i \in [r]$, let $\mathcal{F}_i \subseteq \mathcal{P}(V)$ be an increasing family of sets such that $|A| \ge \varepsilon |V|$ for all $A \in \mathcal{F}_i$. Suppose that each \mathcal{H}_i is $(\mathcal{F}_i, \varepsilon)$ -dense. Further suppose $p \in (0, 1)$ is such that, for every $i \in [r]$ and $\ell \in [k_i]$,

$$\Delta_{\ell}(\mathcal{H}_i) \leqslant c \cdot p^{\ell-1} \frac{e(\mathcal{H}_i)}{|V|}$$

Then there exists a family $S_r \subseteq \mathcal{I}(\mathcal{H}_1, \ldots, \mathcal{H}_r)$ and functions $f : S_r \to \prod_{i \in [r]} \overline{\mathcal{F}_i}$ and $g : \mathcal{I}(\mathcal{H}_1, \ldots, \mathcal{H}_r) \to S_r$ such that the following conditions hold:

- (i) If $(S_1, \ldots, S_r) \in \mathcal{S}_r$ then $\sum |S_i| \leq Cp|V|$;
- (ii) for every $(I_1, \ldots, I_r) \in \mathcal{I}(\mathcal{H}_1, \ldots, \mathcal{H}_r)$, we have that $S \subseteq (I_1, \ldots, I_r) \subseteq S \cup f(S)$ where $S := g(I_1, \ldots, I_r)$.

5.3 Applications of the container method to *r*-tuples of solution-free sets

In this section we will prove Theorems 5.11 and 5.13 by using the container theorem for r-tuples of disjoint independent sets, applied with irredundant partition regular matrices A. Suppose that we have a k-uniform hypergraph \mathcal{H} whose vertex set is a subset of \mathbb{N} and where the edges correspond to the k-distinct solutions of \mathcal{L} . Then in this setting, an (\mathcal{L}^d, r) -free set is precisely an r-tuple of disjoint independent sets in \mathcal{H} .

Theorems 5.11 and 5.13 will be deduced from a container theorem, Theorem 5.21, which in turn follows from Theorem 5.15. Theorem 5.21 actually holds for a class of irredundant matrices of which partition regular matrices are a subclass. Let (*) be the following matrix property:

(*) Under Gaussian elimination A does not have any row which consists of precisely two non-zero rational entries.

For an integer matrix A and integer vector b, call the system of linear equations \mathcal{L} (and the matrix A in the case when b = 0) r-regular if all r-colourings of \mathbb{N} yield a monochromatic solution to Ax = b. Observe that a system of linear equations \mathcal{L} is rregular for all $r \in \mathbb{N}$ if and only if it is partition regular. As outlined in the next subsection, given any $r \ge 2$, all irredundant r-regular matrices A satisfy (*). We will in fact prove stronger versions of Theorems 5.11 and 5.13 that consider irredundant systems of linear equations \mathcal{L} for which the matrix A satisfies (*).

These general results also consider 'asymmetric' Rado properties: Suppose that \mathcal{L}_i is a system of linear equations for each $1 \leq i \leq r$ (and, here and elsewhere, A_i and b_i is the matrix and vector such that $\mathcal{L}_i = \mathcal{L}(A_i, b_i)$). We say a set $X \subseteq \mathbb{N}$ is $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -free if there is an r-colouring of X such that there are no k-distinct solutions to \mathcal{L}_i in X in colour *i* for every $i \in [r]$. Otherwise we say that X is $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -Rado. We denote the size of the largest $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -free subset of [n] by $\mu(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$.

In general it is not known which systems of linear equations $\mathcal{L}_1, \ldots, \mathcal{L}_r$ are such that \mathbb{N} is $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -Rado. However, if each \mathcal{L}_i is an *r*-regular homogenous linear equation, then \mathbb{N} is $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -Rado (see [76, Theorem 9.19]).

We will prove the following strengthenings of Theorems 5.11 and 5.13.

Theorem 5.16. For all positive integers r, all full rank integer matrices A_1, \ldots, A_r of dimension $\ell_i \times k_i$ which satisfy (*) with $m(A_1) \ge \ldots \ge m(A_r)$, all integer vectors b_i of dimension ℓ_i where \mathcal{L}_i is irredundant, and all $\delta > 0$, there exists a constant C > 0 such that

$$\lim_{n \to \infty} \mathbb{P}\left[\frac{\operatorname{res}([n]_p, (\mathcal{L}_1^d, \dots, \mathcal{L}_r^d) - Rado)}{|[n]_p|} = 1 - \frac{\mu(n, \mathcal{L}_1^d, \dots, \mathcal{L}_r^d)}{n} \pm \delta\right] = 1 \quad \text{if } p > Cn^{-1/m(A_1)}.$$

Theorem 5.17. For all positive integers r, all full rank integer matrices A_1, \ldots, A_r of dimension $\ell_i \times k_i$ which satisfy (*), all integer vectors b_i of dimension ℓ_i where \mathcal{L}_i is

irredundant, there are $2^{\mu(n,\mathcal{L}_1^d,\ldots,\mathcal{L}_r^d)+o(n)}$ $(\mathcal{L}_1^d,\ldots,\mathcal{L}_r^d)$ -free subsets of [n].

Given a system of linear equations \mathcal{L} , recall that a strongly \mathcal{L} -free subset of [n] is a subset that contains no solution to \mathcal{L} of any kind. Although this is not quite the same definition as distinct \mathcal{L} -free, we remark that Theorem 5.17 implies a result of Green [48, Theorem 9.3] in the case where $k \ge 3$, on the number of strongly \mathcal{L} -free subsets of [n] for homogeneous linear equations \mathcal{L} .

As mentioned in the introduction of this chapter, Spiegel [112] independently proved the case r = 1 of Theorem 5.16. (Note in [112] this result is mentioned in terms of *abundant* matrices A. That is every $\ell \times (k-2)$ submatrix of A has rank ℓ . But this is clearly equivalent to (*) in the case of irredundant full rank matrices.)

5.3.1 Matrices which satisfy (*)

First we prove that irredundant partition regular matrices are a strict subclass of irredundant matrices which satisfy (*).

Suppose that an irredundant matrix A does not satisfy (*). Then for all solutions (x_1, \ldots, x_k) to \mathcal{L} , there exists a pair $ij \in {\binom{[k]}{2}}$ and non-zero rationals α, β such that $\alpha x_i = \beta x_j$. If $\alpha = \beta$ then no solution to \mathcal{L} is k-distinct and so A is redundant, a contradiction. Otherwise, without loss of generality, assume that $\alpha > \beta$, and devise the following 2-colouring of \mathbb{N} : greedily colour the numbers $\{1, 2, 3, \ldots\}$ so that when colouring x, we always give it a different colour to $\beta x/\alpha$ (if $\beta x/\alpha \in \mathbb{N}$). Such a colouring ensures that no solution to \mathcal{L} is monochromatic, and so A is not partition regular.

Note that the converse is not true. An $\ell \times k$ matrix with columns $a^{(1)}, \ldots, a^{(k)}$ satisfies the *columns property* if there is a partition of [k], say $[k] = D_1 \cup \cdots \cup D_t$ such that

$$\sum_{i \in D_1} a^{(i)} = 0$$

and for every $r \in [t]$ we have

$$\sum_{i \in D_r} a^{(i)} \in \langle a^{(j)} : j \in D_1 \cup \dots \cup D_{r-1} \rangle$$

Rado's theorem [88] states that a matrix is partition regular if and only if it satisfies the columns property. Now, for example $A := \begin{pmatrix} 2 & 2 & -1 \end{pmatrix}$ clearly satisfies (*), and additionally does not have the columns property, so is not partition regular.

The argument above actually implies that if an irredundant matrix A is 2-regular, then it satisfies (*). So in the symmetric case, Theorems 5.16 and 5.17 consider *all* pairs (\mathcal{L}, r) such that \mathcal{L} is an irredundant *r*-regular system of linear equations and $r \ge 2$.

5.3.2 Useful matrix lemmas

Before we can prove our container result (Theorem 5.21), we require some matrix lemmas. Note that all of these lemmas hold for irredundant matrices which satisfy (*). As a consequence, Theorem 5.8 was actually implicitly proven for irredundant matrices which satisfy (*), since in [93] the only necessity of the matrix being partition regular was so that the results stated below could be applied.

Recall the definition of m(A) given by (1.3.2). Parts (i) and (ii) of the following proposition were verified for irredundant partition regular matrices by Rödl and Ruciński (see Proposition 2.2 in [93]). In fact their result easily extends to matrices which satisfy (*). We give the full proof for completeness, and add further facts ((iii)–(v)) which will be useful in the proof of Theorem 5.21.

Proposition 5.18. Let A be an $\ell \times k$ irredundant matrix of full rank ℓ which satisfies (*). Then for every $W \subseteq [k]$, the following hold.

- (i) If |W| = 1, then rank $(A_{\overline{W}}) = \ell$.
- (ii) If $|W| \ge 2$, then $\ell \operatorname{rank}(A_{\overline{W}}) + 2 \le |W|$.

(iii) If $|W| \ge 2$, then

$$|W| + \operatorname{rank}(A_{\overline{W}}) \ge \ell + 1 + \frac{|W| - 1}{m(A)}.$$

Furthermore,

- (iv) $k \ge \ell + 2;$
- (v) m(A) > 1.

Proof. For (i), suppose that $\operatorname{rank}(A_{\overline{W}}) = \ell - 1$ for some $W \subseteq [k]$ with |W| = 1. Since $A_{\overline{W}}$ is an $\ell \times (k-1)$ matrix of rank $\ell - 1$, under Gaussian elimination it must contain a row of zeroes. Hence A under Gaussian elimination contains a row with at most one rational non-zero entry. If there is one, then there are no positive solutions to \mathcal{L} , which contradicts A being irredundant. If there are none, then A does not have rank ℓ , also a contradiction.

For (ii) proceed by induction on |W|. Assume first that there is a $W \subseteq [k]$ with |W| = 2, such that $\operatorname{rank}(A_{\overline{W}}) < \ell$. Using a similar argument to (i), under Gaussian elimination A contains a row with at most two rational non-zero entries. If there are two rational non-zero entries this contradicts A satisfying (*). Otherwise we again get a contradiction to either A being irredundant or of rank ℓ . Assume now that $|W| \ge 3$ and that the statement holds for |W| - 1. The rank of a matrix drops by at most one when a column is deleted, hence the required inequality follows by induction.

For (iii), note that for $|W| \ge 2$, by definition we have $m(A) \ge (|W| - 1)/(|W| - 1 + \operatorname{rank}(A_{\overline{W}}) - \ell)$. This can be rearranged to give the required inequality. For (iv), suppose that $k \le \ell + 1$. Then under Gaussian elimination A must have a row with at most two rational non-zero entries. This leads to the same contradiction as the base case of the induction for (ii). For (v), take W = [k]. Then by definition $m(A) \ge (k-1)/(k-\ell-1) > 1$, where the second inequality follows since the denominator is positive by (iv).

We will require Lemma 2.3. Finally we need the following well known result (and a simple corollary of it), which gives a useful upper bound on the number of solutions to a system of linear equations. Note that in this lemma only, we *do not* assume A to be necessarily of full rank (as we will apply the result directly to matrices formed by deleting columns from our original matrix of full rank).

Lemma 5.19. For an $\ell \times k$ matrix A not necessarily of full rank, an ℓ -dimensional integer vector b and a set $X \subseteq [n]$, the system Ax = b has at most $|X|^{k-\operatorname{rank}(A)}$ solutions in X.

Proof. Let $m := \operatorname{rank}(A)$. Proceed by induction on k. If k = 1, then m = 1 or m = 0. If m = 1, then there is a unique solution to Ax = b, so Ax = b has at most $1 = |X|^{k-m}$ solution in X. If m = 0, then A is the zero matrix, and so each element in X could be a solution (if b = 0), but then Ax = b has at most $|X| = |X|^{k-m}$ solutions in X.

For the inductive step, pick any $c \in [k]$ and fix some $x_c \in X$. Form a new system of linear equations A'x' = b', where A' is formed from $A := (a_{ij})$ by removing the c^{th} column, and $b' := (b'_1, \ldots, b'_\ell)$ is formed from $b := (b_1, \ldots, b_\ell) \in \mathbb{Z}^\ell$ by setting $b'_r := b_r - a_{rc}x_c$. (So $x := (x_1, \ldots, x_k)$ is a solution to Ax = b if and only if $x' := (x_1, \ldots, x_{c-1}, x_{c+1}, \ldots, x_k)$ is a solution to A'x' = b'.)

Note that A' must have rank m or m-1. If A' has rank m then by the induction hypothesis A'x' = b' has at most $|X|^{(k-1)-\operatorname{rank}(A')} = |X|^{k-m-1}$ solutions in X. Since there are |X| valid choices for x_c , there are at most $|X|^{k-m}$ solutions to Ax = b in X, as required.

Now suppose A' has rank m - 1. Then under Gaussian elimination, A' and A have $\ell - m + 1$, and respectively $\ell - m$ rows consisting entirely of zeroes, and in particular, A has a row with precisely one non-zero entry which is in the c^{th} column. Hence there is at most one value x_c can take in any solution $x = (x_1, \ldots, x_k)$ to Ax = b. So for this choice of x_c , Ax = b and A'x' = b' have precisely the same number of solutions in X. Since A' is an $\ell \times (k-1)$ matrix of rank m-1, the induction hypothesis implies that there are
$|X|^{(k-1)-(m-1)} = |X|^{k-\operatorname{rank}(A)}$ solutions to A'x' = b' and thus Ax = b, as desired.

Corollary 5.20. Consider an $\ell \times k$ matrix A of rank ℓ , a set $X \subseteq [n]$ and an integer $1 \leq t \leq k$. Fix distinct $y_1, \ldots, y_t \in X$ and consider any $W = \{s_1, \ldots, s_t\} \subseteq [k]$. The system Ax = b has at most $|X|^{k-t-\operatorname{rank}(A_{\overline{W}})}$ solutions (x_1, \ldots, x_k) in X for which $x_{s_j} = y_j$ for each $j \in [t]$. Moreover, if the system Ax = b is irredundant and A satisfies (*) and t = 1, then the system Ax = b has at most $|X|^{k-\ell-1}$ solutions (x_1, \ldots, x_k) in X for which $x_{s_1} = y_1$.

Proof. Write $A =: (a_{ij})$. Consider the system of linear equations $A_{\overline{W}}x' = b'$ where, for each $r \in [\ell]$, the r^{th} term in b' is

$$b'_r := b_r - \sum_{s_j \in W} a_{rs_j} y_j.$$

Now by Lemma 5.19 the system of linear equations $A_{\overline{W}}x' = b'$ has at most $|X|^{k-t-\operatorname{rank}(A_{\overline{W}})}$ solutions in X. The first part of the corollary then follows since all solutions (x_1, \ldots, x_k) to Ax = b with $x_{s_j} = y_j$ for each $j \in [t]$, rise from a solution x' to $A_{\overline{W}}x' = b'$. For the second part, if Ax = b is irredundant and A satisfies (*) and t = 1, then by Proposition 5.18(i), we have $\operatorname{rank}(A_{\overline{W}}) = \ell$ and so the result follows.

5.3.3 A container theorem for tuples of \mathcal{L}^d -free sets

Recall that an \mathcal{L}^d -free set is simply an $(\mathcal{L}^d, 1)$ -free set. Let $\mathcal{I}(n, \mathcal{L}_1^d, \dots, \mathcal{L}_r^d)$ denote the set of all ordered *r*-tuples $(X_1, \dots, X_r) \in \mathcal{P}([n])^r$ so that each X_i is \mathcal{L}_i^d -free and $X_i \cap X_j = \emptyset$ for all distinct $i, j \in [r]$. Note that any $(\mathcal{L}_1^d, \dots, \mathcal{L}_r^d)$ -free subset X of [n] has a partition X_1, \dots, X_r so that $(X_1, \dots, X_r) \in \mathcal{I}(n, \mathcal{L}_1^d, \dots, \mathcal{L}_r^d)$. We now prove a container theorem for the elements of $\mathcal{I}(n, \mathcal{L}_1^d, \dots, \mathcal{L}_r^d)$. **Theorem 5.21.** Let $r \in \mathbb{N}$ and $0 < \delta < 1$. For each $i \in [r]$ let A_i be an integer matrix of dimension $\ell_i \times k_i$ which satisfies (*), let b_i be an integer vector of dimension ℓ_i and suppose \mathcal{L}_i is irredundant. Suppose that $m(A_1) \ge \ldots \ge m(A_r)$. Then there exists D > 0such that the following holds. For all $n \in \mathbb{N}$, there is a collection $\mathcal{S}_r \subseteq \mathcal{P}([n])^r$ and a function $f : \mathcal{S}_r \to \mathcal{P}([n])^r$ such that:

- (i) For all $(I_1, \ldots, I_r) \in \mathcal{I}(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$, there exists $S \in \mathcal{S}_r$ such that $S \subseteq (I_1, \ldots, I_r) \subseteq f(S)$.
- (*ii*) If $(S_1, ..., S_r) \in S_r$ then $\sum_{i \in [r]} |S_i| \leq Dn^{\frac{m(A_1)-1}{m(A_1)}}$.
- (iii) Every $S \in \mathcal{S}_r$ satisfies $S \in \mathcal{I}(n, \mathcal{L}_1^d, \dots, \mathcal{L}_r^d)$.
- (iv) Given any $S = (S_1, \ldots, S_r) \in \mathcal{S}_r$, write $f(S) =: (f(S_1), \ldots, f(S_r))$. Then
 - (a) for each $1 \leq i \leq r$, $f(S_i)$ contains at most $\delta n^{k_i \ell_i} k_i$ -distinct solutions to \mathcal{L}_i ; and

$$(b) \mid \bigcup_{i \in [r]} f(S_i) \mid \leq \mu(n, \mathcal{L}_1^d, \dots, \mathcal{L}_r^d) + \delta n$$

We emphasise that (iv)(b) does not necessarily guarantee $\sum_{i \in [r]} |f(S_i)| \leq \mu(n, \mathcal{L}_1^d, \dots, \mathcal{L}_r^d) + \delta n$. Rather it ensures at most $\mu(n, \mathcal{L}_1^d, \dots, \mathcal{L}_r^d) + \delta n$ elements of [n] appear in at least one of the co-ordinates of f(S). This property is crucial for our applications.

Proof. First note that since each of the systems of linear equations \mathcal{L}_i are irredundant, a result of Kusch, Rué, Spiegel and Szabó [75] implies that there exists a constant d > 0such that, for each $i \in [r]$, there are at least $dn^{k_i - \ell_i} k_i$ -distinct solutions to \mathcal{L}_i in [n]. (This is a generalisation of a result by Janson and Ruciński [65].)

Note that it suffices to prove the theorem in the case when $0 < \delta < d$. So let $0 < \delta < d$ and $r \in \mathbb{N}$ be given and apply Lemma 2.3 to obtain $n_0, \varepsilon > 0$. Without loss of generality we may assume $\varepsilon \leq \delta$. Define $k := \max k_i$ and let

$$\varepsilon' := \frac{\varepsilon}{2}$$
 and $c := \frac{k!}{\varepsilon'}$.

Apply Theorem 5.15 with parameters $r, k_1, \ldots, k_r, c, \varepsilon'$ playing the roles of $r, k_1, \ldots, k_r, c, \varepsilon$ respectively to obtain $D_1 > 0$. Increase n_0 if necessary so that

$$0 < \frac{1}{n_0} \ll \frac{1}{D_1}, \frac{1}{k_1}, \dots, \frac{1}{k_r}, \frac{1}{r}, \varepsilon, \delta.$$

For $n < n_0$, set S_r to be $\mathcal{I}(n, \mathcal{L}_1^d, \dots, \mathcal{L}_r^d)$; set f to be the identity function and choose D_2 to be large. By setting D to be the maximum of D_1 and D_2 , it remains to prove the result for integers $n \ge n_0$. So now fix $n \ge n_0$.

For each $i \in [r]$ let $\mathcal{H}_{n,i}$ be the hypergraph with $V(\mathcal{H}_{n,i}) := [n]$ and an edge set which consists of all k_i -distinct solutions to \mathcal{L}_i in [n]. Observe that $\mathcal{H}_{n,i}$ is k_i -uniform and an independent set in $\mathcal{H}_{n,i}$ is an \mathcal{L}_i^d -free set.

For each $i \in [r]$ we define $\mathcal{F}_{n,i} := \{F \subseteq V(\mathcal{H}_{n,i}) : e(\mathcal{H}_{n,i}[F]) \ge \varepsilon' e(\mathcal{H}_{n,i})\}$. Note that since $\varepsilon' < d$, we have

$$\varepsilon' n^{k_i - \ell_i} \leqslant e(\mathcal{H}_{n,i}). \tag{5.3.1}$$

We claim that $\mathcal{H}_{n,i}$ and $\mathcal{F}_{n,i}$ satisfy the hypotheses of Theorem 5.15 with parameters chosen as above with

$$p = p(n) := n^{-1/m(A_1)}.$$

Clearly $\mathcal{F}_{n,i}$ is increasing and $\mathcal{H}_{n,i}$ is $(\mathcal{F}_{n,i},\varepsilon')$ -dense. By Lemma 5.19, a set $F \subseteq V(\mathcal{H}_{n,i})$ contains at most $|F|^{k_i-\ell_i}$ solutions to \mathcal{L}_i (so $e(\mathcal{H}_{n,i}[F]) \leq |F|^{k_i-\ell_i}$). Hence for all

 $F \in \mathcal{F}_{n,i}$, we have

$$|F| \ge e(\mathcal{H}_{n,i}[F])^{\frac{1}{k_i - \ell_i}} \ge (\varepsilon' e(\mathcal{H}_{n,i}))^{\frac{1}{k_i - \ell_i}} \stackrel{(5.3.1)}{\ge} ((\varepsilon')^2 n^{k_i - \ell_i})^{\frac{1}{k_i - \ell_i}} \ge \varepsilon' n^{k_i - \ell_i}$$

where the last inequality follows by Proposition 5.18(iv).

For each $j \in [k_i]$, we wish to bound the number of hyperedges containing some $\{y_1, \ldots, y_j\} \subseteq V(\mathcal{H}_{n,i})$. Suppose (x_1, \ldots, x_{k_i}) is a k_i -distinct solution to \mathcal{L}_i so that $\{y_1, \ldots, y_j\} \subseteq \{x_1, \ldots, x_{k_i}\}$. There are $k_i!/(k_i - j)!$ choices for picking the j roles the y_i play in (x_1, \ldots, x_{k_i}) . Let W be one such choice for the set of indices of the x_a used by $\{y_1, \ldots, y_j\}$. In this case, Corollary 5.20 implies there are at most $n^{k_i - j - \operatorname{rank}((A_i)_{\overline{W}})}$ such solutions to \mathcal{L}_i , and if j = 1, there are at most $n^{k_i - \ell_i - 1}$ such solutions. So for j = 1 this yields

$$\deg_{\mathcal{H}_{n,i}}(y_1) \leqslant k_i n^{k_i - \ell_i - 1} \overset{(5.3.1)}{\leqslant} \frac{k_i}{\varepsilon'} \frac{e(\mathcal{H}_{n,i})}{v(\mathcal{H}_{n,i})} \leqslant c \frac{e(\mathcal{H}_{n,i})}{v(\mathcal{H}_{n,i})}$$

For $j \ge 2$, by Proposition 5.18(iii) we have $k_i - j - \operatorname{rank}((A_i)_{\overline{W}}) \le k_i - \ell_i - 1 - (j-1)/m(A_i)$. Also $m(A_1) \ge m(A_i)$ for all $i \in [r]$ and hence we have

$$\deg_{\mathcal{H}_{n,i}}(\{y_1,\ldots,y_j\}) \leqslant k_i! n^{k_i - \ell_i - 1 - \frac{j-1}{m(A_i)}} \leqslant k_i! n^{k_i - \ell_i - 1 - \frac{j-1}{m(A_1)}}$$
$$\leqslant \frac{k_i!}{\varepsilon'} p^{j-1} \frac{e(\mathcal{H}_{n,i})}{v(\mathcal{H}_{n,i})} \leqslant c p^{j-1} \frac{e(\mathcal{H}_{n,i})}{v(\mathcal{H}_{n,i})}.$$

Since $\{y_1, \ldots, y_j\}$ was arbitrary, we therefore have $\Delta_j(\mathcal{H}_{n,i}) \leq cp^{j-1}e(\mathcal{H}_{n,i})/v(\mathcal{H}_{n,i})$, as required. We have therefore shown that $\mathcal{H}_{n,i}$ and $\mathcal{F}_{n,i}$ satisfy the hypotheses of Theorem 5.15 for all $i \in [r]$.

Then Theorem 5.15 implies that there exists a family $S_r \subseteq \prod_{i \in [r]} \mathcal{P}(V(\mathcal{H}_{n,i})) = \mathcal{P}([n])^r$ and functions $f' : S_r \to \prod_{i \in [r]} \overline{\mathcal{F}_{n,i}}$ and $g : \mathcal{I}(\mathcal{H}_{n,1}, \ldots, \mathcal{H}_{n,r}) \to S_r$ such that the following conditions hold:

- (a) If $(S_1, \ldots, S_r) \in \mathcal{S}_r$ then $\sum_{i \in [r]} |S_i| \leq D_1 pn$;
- (b) every $S \in \mathcal{S}_r$ satisfies $S \in \mathcal{I}(\mathcal{H}_{n,1}, \ldots, \mathcal{H}_{n,r})$;
- (c) for every $(I_1, \ldots, I_r) \in \mathcal{I}(\mathcal{H}_{n,1}, \ldots, \mathcal{H}_{n,r})$, we have that $S \subseteq (I_1, \ldots, I_r) \subseteq S \cup f'(S)$, where $S := g(I_1, \ldots, I_r)$.

Note that $\mathcal{I}(\mathcal{H}_{n,1},\ldots,\mathcal{H}_{n,r}) = \mathcal{I}(n,\mathcal{L}_1^d,\ldots,\mathcal{L}_r^d)$. For each $S \in \mathcal{S}_r$, define

$$f(S) := S \cup f'(S).$$

So $f: \mathcal{S}_r \to \mathcal{P}([n])^r$. Thus, (a)–(c) immediately imply that (i)–(iii) hold.

Given any $S = (S_1, \ldots, S_r) \in S_r$ write $f(S) =: (f(S_1), \ldots, f(S_r))$ and $f'(S) =: (f'(S_1), \ldots, f'(S_r))$. (Note the slight abuse of the use of the f and f' notation here.) By definition of $\mathcal{F}_{n,i}$ any $F \in \overline{\mathcal{F}_{n,i}}$ contains at most $\varepsilon' n^{k_i - \ell_i} k_i$ -distinct solutions to \mathcal{L}_i . By Corollary 5.20, the number of k_i -distinct solutions to \mathcal{L}_i in [n] that use at least one element from S_i is at most $k_i n^{k_i - \ell_i - 1} |S_i|$. Further,

$$k_i n^{k_i - \ell_i - 1} |S_i| \leqslant k_i D_1 p n^{k_i - \ell_i} \leqslant \varepsilon' n^{k_i - \ell_i}.$$

Here, the first inequality holds by (a), and the second since $p = n^{-1/m(A_1)}$ and $m(A_1) > 0$ by Proposition 5.18(v). Thus, in total $f(S_i) = S_i \cup f'(S_i)$ contains at most $2\varepsilon' n^{k_i - \ell_i} \leq \delta n^{k_i - \ell_i} k_i$ -distinct solutions to \mathcal{L}_i , so (iv)(a) holds.

In fact, the argument above implies that there is an r-colouring of the set $\bigcup_{i \in [r]} f(S_i)$ so that there are at most $2\varepsilon' n^{k_i - \ell_i} = \varepsilon n^{k_i - \ell_i} k_i$ -distinct solutions to \mathcal{L}_i in colour *i*, in $\bigcup_{i \in [r]} f(S_i)$. Hence, Lemma 2.3 ensures (iv)(b), as desired.

5.3.4 The number of $(\mathcal{L}_1^d, \dots, \mathcal{L}_r^d)$ -free subsets of [n]

Our first application of Theorem 5.21 yields an enumeration result (Theorem 5.17) for the number of $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -free subsets of [n].

Proof of Theorem 5.17. By definition of $\mu(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ there are at least $2^{\mu(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)}$ $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -free subsets of [n]. So it suffices to prove the upper bound.

For this, note that we may assume n is sufficiently large. Let $0 < \delta < 1$ be arbitrary and let D > 0 be obtained from Theorem 5.21 applied to A_1, \ldots, A_r with parameter δ . We obtain a collection S_r and function f as in Theorem 5.21. Consider any $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ free subset X of [n]. Note that X has a partition X_1, \ldots, X_r so that $(X_1, \ldots, X_r) \in$ $\mathcal{I}(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$. So by Theorem 5.21(i) this means there is some $S = (S_1, \ldots, S_r) \in S_r$ so that $X \subseteq \bigcup_{i \in [r]} f(S_i)$.

Further, given any $S = (S_1, \ldots, S_r) \in \mathcal{S}_r$, we have that $|\bigcup_{i \in [r]} f(S_i)| \leq \mu(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d) + \delta n$. Thus, each such $\bigcup_{i \in [r]} f(S_i)$ contains at most $2^{\mu(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d) + \delta n}$ $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -free subsets of [n]. Note that, by Theorem 5.21(ii),

$$|\mathcal{S}_r| \leqslant \left(\sum_{s=0}^{Dn \frac{m(A_1)-1}{m(A_1)}} \binom{n}{s}\right)^r < 2^{\delta n},$$

where the last inequality holds since n is sufficiently large.

Altogether, this implies that the number of $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -free subsets of [n] is at most

$$2^{\delta n} \times 2^{\mu(n,\mathcal{L}_1^d,\dots,\mathcal{L}_r^d)+\delta n} = 2^{\mu(n,\mathcal{L}_1^d,\dots,\mathcal{L}_r^d)+2\delta n}$$

Since the choice of $0 < \delta < 1$ was arbitrary this proves the theorem.

5.3.5 The resilience of being $(\mathcal{L}_1^d, \dots, \mathcal{L}_r^d)$ -Rado

Recall that the resilience of S with respect to \mathcal{P} , $\operatorname{res}(S, \mathcal{P})$, is the minimum number t such that by deleting t elements from S, one can obtain a set not having \mathcal{P} . In this section we will determine $\operatorname{res}([n]_p, (\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -Rado) for irredundant systems of linear equations $\mathcal{L}_1, \ldots, \mathcal{L}_r$ for which matrices A_1, \ldots, A_r which satisfy (*). We now use Theorem 5.21 to deduce Theorem 5.16.

Proof of Theorem 5.16. Let $0 < \delta < 1$, $r \in \mathbb{N}$ and $\mathcal{L}_1, \ldots, \mathcal{L}_r$ be the systems of linear equations as in the statement of the theorem. Given n, if $p > n^{-1/m(A_1)}$ then since $m(A_1) > 1$ by Proposition 5.18(v), Proposition 1.2(ii) implies that, w.h.p.,

$$|[n]_p| = \left(1 \pm \frac{\delta}{4}\right) pn.$$
(5.3.2)

We first show that

$$\lim_{n \to \infty} \mathbb{P}\left[\frac{\operatorname{res}([n]_p, (\mathcal{L}_1^d, \dots, \mathcal{L}_r^d) \operatorname{-Rado})}{|[n]_p|} \leqslant 1 - \frac{\mu(n, \mathcal{L}_1^d, \dots, \mathcal{L}_r^d)}{n} + \delta\right] = 1 \quad \text{if } p > n^{-1/m(A_1)}.$$

For this, we must show that the probability of the event that there exists a set $S \subseteq [n]_p$ such that $|S| \ge (\mu(n, \mathcal{L}_1^d, \dots, \mathcal{L}_r^d)/n - \delta)|[n]_p|$ and S is $(\mathcal{L}_1^d, \dots, \mathcal{L}_r^d)$ -free, tends to one as n tends to infinity. This indeed follows: Let T be an $(\mathcal{L}_1^d, \dots, \mathcal{L}_r^d)$ -free subset of [n] of maximum size $\mu(n, \mathcal{L}_1^d, \dots, \mathcal{L}_r^d)$. Then, by Proposition 1.2(ii), w.h.p. we have $|T \cap [n]_p| = (\mu(n, \mathcal{L}_1^d, \dots, \mathcal{L}_r^d)/n \pm \delta)|[n]_p|$, and $T \cap [n]_p$ is $(\mathcal{L}_1^d, \dots, \mathcal{L}_r^d)$ -free, as required.

For the remainder of the proof, we will focus on the lower bound, namely that there exists C > 0 such that whenever $p > Cn^{-1/m(A_1)}$,

$$\mathbb{P}\left[\operatorname{res}([n]_p, (\mathcal{L}_1^d, \dots, \mathcal{L}_r^d) \operatorname{-Rado}) \geqslant \left(1 - \frac{\mu(n, \mathcal{L}_1^d, \dots, \mathcal{L}_r^d)}{n} - \delta\right) |[n]_p|\right] \to 1 \quad \text{as } n \to \infty.$$
(5.3.3)

Suppose *n* is sufficiently large. Apply Theorem 5.21 with parameters $r, \delta/8, \mathcal{L}_1, \ldots, \mathcal{L}_r$ to obtain D > 0, a collection $S_r \subseteq \mathcal{P}([n])^r$ and a function *f* satisfying (i)–(iv). Now choose *C* such that $0 < 1/C \ll 1/D, \delta, 1/r$. Let $p \ge Cn^{-1/m(A_1)}$.

Since (5.3.2) holds with high probability, to prove (5.3.3) holds it suffices to show that the probability $[n]_p$ contains an $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -free subset of size at least $(\frac{\mu(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)}{n} + \delta/2)np$ tends to zero as n tends to infinity.

Suppose that $[n]_p$ does contain an $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -free subset I of size at least $(\frac{\mu(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)}{n} + \delta/2)np$. Note that I has a partition I_1, \ldots, I_r so that $(I_1, \ldots, I_r) \in \mathcal{I}(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$. Further, there is some $S = (S_1, \ldots, S_r) \in S_r$ such that $S \subseteq (I_1, \ldots, I_r) \subseteq f(S)$. Thus, $[n]_p$ must contain $\cup_{i \in [r]} S_i$ as well as at least $(\frac{\mu(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)}{n} + \delta/4)np$ elements from $(\bigcup_{i \in [r]} f(S_i)) \setminus (\bigcup_{i \in [r]} S_i)$. (Note here we are using that $|\bigcup_{i \in [r]} S_i| \leq \delta np/4$, which holds by Theorem 5.21(ii) and since $0 < 1/C \ll 1/D, \delta$.) Writing $s := |\bigcup_{i \in [r]} S_i|$, the probability $[n]_p$ contains $\bigcup_{i \in [r]} S_i$ is p^s . Note that $|(\bigcup_{i \in [r]} f(S_i)) \setminus (\bigcup_{i \in [r]} S_i)| \leq \mu(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d) + \delta n/8$ by Theorem 5.21(iv)(b). So by Proposition 1.2(i), the probability $[n]_p$ contains at least $(\frac{\mu(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)}{n} + \delta/4)np$ elements from $(\bigcup_{i \in [r]} f(S_i)) \setminus (\bigcup_{i \in [r]} S_i)$, is at most $\exp(-\delta^2 np/256)$.

Write $N := n^{(m(A_1)-1)/m(A_1)}$ and $\gamma := \delta^2/256$. Given some $0 \leq s \leq DN$, there are at most $r^s \binom{n}{s}$ elements $(S_1, \ldots, S_r) \in \mathcal{S}_r$ such that $|\bigcup_{i \in [r]} S_i| = s$. Indeed, this follows since there are r^s ways to partition a set of size s into r classes. Thus, the probability $[n]_p$ does contain an $(\mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$ -free subset I of size at least $(\frac{\mu(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)}{n} + \delta/2)np$ is at most

$$\begin{split} \sum_{s=0}^{DN} r^s \binom{n}{s} \cdot p^s \cdot e^{-\gamma n p} &\leqslant (DN+1)(rp)^{DN} \binom{n}{DN} e^{-\gamma n p} \leqslant (DN+1) \left(\frac{repn}{DN}\right)^{DN} e^{-\gamma n p} \\ &\leqslant (DN+1) \left(\frac{reC}{D}\right)^{DN} e^{-\gamma CN} \leqslant e^{\gamma CN/2} e^{-\gamma CN} = e^{-\gamma CN/2} \end{split}$$

which tends to zero as n tends to infinity. This completes the proof.

5.3.6 The size of the largest (\mathcal{L}^s, r) -free set

Both as a natural question in itself, and in light of Theorems 5.16 and 5.17, it is of interest to obtain good bounds on $\mu(n, \mathcal{L}_1^d, \ldots, \mathcal{L}_r^d)$. For the rest of this section consider the symmetric homogeneous case $(A := A_1 = \cdots = A_r \text{ and } b = 0)$ and assume that A is a $1 \times k$ matrix, i.e. we are interested in solutions to a linear equation $a_1x_1 + \cdots + a_kx_k = 0$. Recall $S \subseteq [n]$ is *strongly* (\mathcal{L}, r) -*free* if there exists an r-colouring of S which contains no monochromatic solutions to \mathcal{L} of any type (that is, solutions are not required to be k-distinct). Note that for any density regular matrix A, (x, \ldots, x) is a solution to \mathcal{L} for all $x \in [n]$ (as observed by Frankl, Graham and Rödl [41, Fact 4]) and so we have $\mu(n, \mathcal{L}^s, r) = 0$. (Note that this result implies that all density regular $1 \times k$ matrices give rise to an equation \mathcal{L} which is translation-invariant.) In fact, if A is any $1 \times k$ irredundant integer matrix, then for all $\varepsilon > 0$ there exists an $n_0 > 0$ such that for all integers $n \ge n_0$ we have

$$\mu(n, \mathcal{L}^s, r) \leqslant \mu(n, \mathcal{L}^d, r) \leqslant \mu(n, \mathcal{L}^s, r) + \varepsilon n.$$

This follows from Lemma 2.2 since such \mathcal{L} have $o(n^{k-\ell})$ non-k-distinct solutions in [n](i.e. a solution (x_1, \ldots, x_k) where there is an $i \neq j$ such that $x_i = x_j$).

Consequently it is equally interesting to study $\mu(n, \mathcal{L}^s, r)$ in the case when $\mu(n, \mathcal{L}^d, r) = \Omega(n)$. In the case of sum-free sets (where \mathcal{L} is x + y = z), the study of $\mu(n, \mathcal{L}^s, r)$ is a classical problem of Abbott and Wang [1]. (Note that the only difference between $\mu(n, \mathcal{L}^d, r)$ and $\mu(n, \mathcal{L}^s, r)$ in this case is that $\mu(n, \mathcal{L}^d, r)$ allows non-distinct sums x + x = z whereas $\mu(n, \mathcal{L}^s, r)$ does not.) Let $\mu(n, r) := \mu(n, \mathcal{L}^s, r)$ where \mathcal{L} is x + y = z. An easy proof shows that $\mu(n, 1) = \lceil n/2 \rceil$.

The following definitions help motivate the study of $\mu(n, r)$ for $r \ge 2$. Let f(r) denote the largest positive integer m for which there exists a partition of [m] into r sum-free sets, and let h(r) denote the largest positive integer m for which there exists a partition of [m] into r sets which are sum-free modulo m + 1.

Abbott and Wang [1] conjectured that h(r) = f(r), and showed that $\mu(n, r) \ge n - \lfloor n/(h(r) + 1) \rfloor$. They also proved the following upper bound.

Theorem 5.22 ([1]). We have $\mu(n,r) \leq n - \lfloor cn/((f(r)+1)\log(f(r)+1)) \rfloor$ where $c := e^{-\gamma} \approx 0.56$ (γ denotes the Euler-Mascheroni constant).

We provide an alternate upper bound, which is a modification of Hu's [61] proof that $\mu(n,2) = n - \lfloor \frac{n}{5} \rfloor$. (To see why this is a lower bound, consider the set $\{x \in [n] : x \equiv 1 \text{ or } 4 \mod 5\} \cup \{y \in [n] : y \equiv 2 \text{ or } 3 \mod 5\}$.) First we need the following fact. Given $x \in [n]$ and $T \subseteq [n]$, write $x + T := \{x + y : y \in T\}$. Given $S, T \subseteq [n]$, say that T is a difference set of S if there exists $x \in S$ such that $x + T \subseteq S$.

Fact 5.23. Let $n \in \mathbb{N}$ and $S, T, T' \subseteq [n]$.

- (i) If T is a difference set of a sum-free set S, then $S \cap T = \emptyset$.
- (ii) If T' is a difference set of T, and T is a difference set of S, then T' is a difference set of S.

Proof. If there exists $x \in S$ such that $x + T \subseteq S$ and moreover there exists $y \in S \cap T$, then $x + y \in S$, proving (i). For (ii), suppose that there is $x' \in T$ and $x \in S$ such that $x' + T' \subseteq T$ and $x + T \subseteq S$. Then $x + x' + T' \subseteq S$ and $x + x' \in x + T \subseteq S$, proving (ii).

Theorem 5.24. We have $\mu(n,r) \leq n - \lfloor \frac{n}{|r|e|} \rfloor$.

Note that Theorem 5.24 does indeed recover Hu's bound [61] for the case r = 2.

Proof. Fix $n, r \in \mathbb{N}$. Let $\ell(0) := 1$. For all integers $i \ge 1$, define

$$\ell(i) := i! \left(1 + \sum_{t \in [i]} \frac{1}{t!} \right) = \lfloor i! e \rfloor.$$

Note that $\ell(i) = i\ell(i-1) + 1$ for all $i \ge 1$. Choose the unique $q \in \mathbb{N} \cup \{0\}$ and $0 \le k \le \ell(r) - 1$ such that $n = \ell(r)q + k$. Consider any partition $S_1 \cup \cdots \cup S_r \cup R = [n]$, where each S_i is sum-free. We wish to show that $|R| \ge q$, since then $\mu(\ell(r)q + k, r) \le (\ell(r) - 1)q + k$ and so $\mu(n, r) \le n - \lfloor n/\ell(r) \rfloor$.

Suppose not. We will obtain integers $\{j_1, \ldots, j_r\} = [r]$ and subsets D_0, D_1, \ldots, D_r of [n] such that the following properties hold for all $0 \leq i \leq r$.

$$P_1(i) |D_i| \ge \ell(r-i)q;$$

 $P_2(i)$ D_i is a difference set of S_{j_t} for all $t \in [i]$;

$$P_3(i) \ D_i \cap S_{j_t} = \emptyset \text{ for all } t \in [i].$$

Let $D_0 := [n]$. Then $P_1(0)$ holds by definition, and $P_2(0)$ and $P_3(0)$ are vacuous. Suppose, for some $0 \leq i < r$, we have obtained distinct $\{j_1, \ldots, j_i\} \subseteq [r]$ and D_0, D_1, \ldots, D_i such that $P_1(t)-P_3(t)$ hold for all $t \in [i]$.

Suppose that $|D_i \cap \bigcup_{t \in [r] \setminus \{j_1, \dots, j_i\}} S_t| \leq (\ell(r-i) - 1)q$. Then we have that

$$|D_i \cap R| \stackrel{P_3(i)}{\geqslant} |D_i| - (\ell(r-i) - 1)q \stackrel{P_1(i)}{\geqslant} q,$$

a contradiction. So by averaging, there exists $j_{i+1} \in [r] \setminus \{j_1, \ldots, j_i\}$ such that

$$|D_i \cap S_{j_{i+1}}| \ge \left\lceil \frac{(\ell(r-i)-1)q+1}{r-i} \right\rceil = \ell(r-i-1)q+1$$

Thus we can write $D_i \cap S_{j_{i+1}} \supseteq \{s_{i,0} < \ldots < s_{i,\ell(r-i-1)q}\}$. Let $D_{i+1} := \{s_{i,x} - s_{i,0} : x \in [\ell(r-i-1)q]\}$. We claim that $P_1(i+1) - P_3(i+1)$ hold. Property $P_1(i+1)$ is clear by definition. For $P_2(i+1)$, note that D_{i+1} is a difference set of both D_i and $S_{j_{i+1}}$. Then Fact 5.23(ii) and $P_2(i)$ imply that additionally D_{i+1} is a difference set of S_{j_t} for all $t \in [i]$. Fact 5.23(i) implies that $D_{i+1} \cap S_{j_t} = \emptyset$ for all $t \in [i+1]$, proving $P_3(i+1)$.

Thus we obtain D_r satisfying $P_1(r) - P_3(r)$. By $P_1(r)$ and $P_3(r)$ we have that $|D_r| \ge \ell(0)q = q$ and $D_r \subseteq R$, a contradiction.

5.3.7 Two-coloured analogue of the Cameron–Erdős conjecture

We conclude the section with a problem which, since the paper [55] corresponding to this chapter was first submitted, has been (essentially) solved by Tran [119]. Recall Hu [61] showed that $\mu(n,2) = n - \lfloor \frac{n}{5} \rfloor$. So in the case when \mathcal{L} is x + y = z, Theorem 5.17 implies that there are $2^{4n/5+o(n)}$ (\mathcal{L}^d , 2)-free subsets of [n]. By considering (\mathcal{L}^s , 2)-free subsets of [n] instead, the error term in the exponent here can be replaced by a constant.

Theorem 5.25 ([119]). Let \mathcal{L} denote x + y = z. There are $\Theta(2^{4n/5})$ ($\mathcal{L}^s, 2$)-free subsets of [n].

Note that Theorem 5.25 can be viewed as a 2-coloured analogue of the Cameron–Erdős conjecture [23] which was famously resolved by Green [47] and independently Sapozhenko [104].

5.4 Applications of the container method to graph Ramsey theory

In this section we answer some questions in hypergraph Ramsey theory, introduced in Sections 5.1.1 and 5.1.2. How many *n*-vertex hypergraphs are not Ramsey, and what does a typical such hypergraph look like? How dense must the Erdős-Rényi random hypergraph be to have the Ramsey property with high probability, and above this threshold, how strongly does it possess the Ramsey property?

Our main results here are applications of the asymmetric container theorem (Theorem 5.15). For arbitrary k-uniform hypergraphs H_1, \ldots, H_r , we first prove Theorem 5.34, a container theorem for non- (H_1, \ldots, H_r) -Ramsey k-uniform hypergraphs. To see how one might prove such a theorem, observe that, if \mathcal{H}_i is the hypergraph of copies of H_i on *n* vertices (i.e. vertices correspond to *k*-subsets of [n], and edges correspond to copies of $E(H_i)$), then every non- (H_1, \ldots, H_r) -Ramsey *k*-uniform hypergraph *G* corresponds to a set in $\mathcal{I}(\mathcal{H}_1, \ldots, \mathcal{H}_r)$. We then use Theorem 5.34 to:

- count the number of k-uniform hypergraphs on n vertices which are not (H₁,..., H_r)-Ramsey (Theorem 5.12);
- (2) determine the global resilience of $G_{n,p}^{(k)}$ with respect to the property of being (H_1, \ldots, H_r) -Ramsey (Theorem 5.7). That is, we show that there is a constant C such that whenever $p \ge Cn^{-1/m_k(H_1)}$, we obtain a function t(n,p) such that, with high probability, any subhypergraph $G \subseteq G_{n,p}^{(k)}$ with $e(G) > t + \Omega(pn^k)$ is (H_1, \ldots, H_r) -Ramsey. Further, there is some $G' \subseteq G_{n,p}^{(k)}$ with $e(G') > t - o(pn^k)$ which is not (H_1, \ldots, H_r) -Ramsey.
- (3) As a corollary of (2), we see that, whenever $p \ge Cn^{-1/m_k(H_1)}$, the random hypergraph $G_{n,p}^{(k)}$ is (H_1, \ldots, H_r) -Ramsey with high probability.

Notice that each of the statements (1)–(3) involve a common parameter: the maximum size ex^r $(n; H_1, \ldots, H_r)$ of an *n*-vertex *k*-uniform hypergraph which is not (H_1, \ldots, H_r) -Ramsey. For this reason, we generalise the classical supersaturation result of Erdős and Simonovits [36] to show that any *n*-vertex *k*-uniform hypergraph *G* with at least ex^r $(n; H_1, \ldots, H_r)$ + $\Omega(n^k)$ edges is somehow 'strongly' (H_1, \ldots, H_r) -Ramsey. In the graph case, an old result of Burr, Erdős and Lovász [21] allows us to quite accurately determine ex^r $(n; H_1, \ldots, H_r)$.

5.4.1 Definitions and notation

In this section, $k \ge 2$ is an integer and we use *k*-graph as shorthand for *k*-uniform hypergraph. Recall from Section 5.1.1 that, given $r \in \mathbb{N}$ and a *k*-graph *G*, an *r*-colouring is a function $\sigma : E(G) \to [r]$. Given *k*-graphs H_1, \ldots, H_r , we say that σ is (H_1, \ldots, H_r) - free if $\sigma^{-1}(i)$ is H_i -free for all $i \in [r]$. Then G is (H_1, \ldots, H_r) -Ramsey if it has no (H_1, \ldots, H_r) -free r-colouring.

Given an integer $\ell \ge k$, denote by $K_{\ell}^{(k)}$ the complete k-graph on ℓ vertices. A k-graph H is k-partite if the vertices of H can be k-coloured so that each edge contains one vertex of each colour. Given a k-graph S, recall the definitions

$$d_k(S) := \begin{cases} 0 & \text{if } e(S) = 0; \\ 1/k & \text{if } v(S) = k \text{ and } e(S) = 1; \\ \frac{e(S)-1}{v(S)-k} & \text{otherwise} \end{cases}$$

and

$$m_k(S) := \max_{S' \subseteq S} d_k(S').$$

5.4.2 The maximum density of a hypergraph which is not Ramsey

Given integers $n \ge k$ and a k-graph H, we denote by ex(n; H) the maximum size of an *n*-vertex *H*-free k-graph. Define the *Turán density* $\pi(H)$ of *H* by

$$\pi(H) := \lim_{n \to \infty} \frac{\operatorname{ex}(n; H)}{\binom{n}{k}}$$
(5.4.1)

(which exists by a simple averaging argument, see [66]). The so-called supersaturation phenomenon discovered by Erdős and Simonovits [36] asserts that any sufficiently large hypergraph with density greater than $\pi(H)$ contains not just one copy of H, but in fact a positive fraction of v(H)-sized sets span a copy of H.

Theorem 5.26 ([36]). For all $k \in \mathbb{N}$; $\delta > 0$ and all k-graphs H, there exist $n_0, \varepsilon > 0$ such that for all integers $n \ge n_0$, every n-vertex k-graph G with $e(G) \ge (\pi(H) + \delta) \binom{n}{k}$ contains at least $\varepsilon \binom{n}{v(H)}$ copies of H.

When k = 2, the Erdős–Stone–Simonovits theorem [37] says that for all graphs H, the value of $\pi(H)$ is determined by the chromatic number $\chi(H)$ of H, via

$$\pi(H) = 1 - \frac{1}{\chi(H) - 1}.$$
(5.4.2)

For $k \ge 3$, the value of $\pi(H)$ is only known for a small family of k-graphs H. It remains an open problem to even determine the Turán density of $K_4^{(3)}$, the smallest non-trivial complete 3-graph (the widely-believed conjectured value is $\frac{5}{9}$). For more background on this, the so-called *hypergraph Turán problem*, the interested reader should consult the excellent survey of Keevash [67].

In this section, we generalise Theorem 5.26 from H-free hypergraphs to non- (H_1, \ldots, H_r) -Ramsey hypergraphs (note that a hypergraph is H-free if and only if it is not (H)-Ramsey). Given $\varepsilon > 0$, we say that an n-vertex k-graph G is ε -strongly (H_1, \ldots, H_r) -Ramsey if for all r-colourings σ of G there exists an $i \in [r]$ such that the number of copies of H_i in $\sigma^{-1}(i)$ is more than $\varepsilon {n \choose v(H_i)}$.

Using a well-known averaging argument of Katona, Nemetz and Simonovits [66], we can show that $\binom{n}{k}^{-1} \exp^r(n; H_1, \ldots, H_r)$ converges as n tends to infinity. Indeed, let G be an n-vertex non- (H_1, \ldots, H_r) -Ramsey graph with $e(G) = \exp^r(n; H_1, \ldots, H_r)$. The average density of an (n-1)-vertex induced subgraph of G is precisely

$$\binom{n}{n-1}^{-1} \sum_{U \subseteq V(G): |U|=n-1} \frac{e(G[U])}{\binom{n-1}{k}} = (n-k)^{-1} \cdot \binom{n}{k}^{-1} \sum_{U \subseteq V(G): |U|=n-1} e(G[U]) = \binom{n}{k}^{-1} e(G).$$

But the left-hand side is at most $\binom{n-1}{k}^{-1} \cdot ex^r(n-1; H_1, \ldots, H_r)$, otherwise G would contain an (n-1)-vertex subgraph which is (H_1, \ldots, H_r) -Ramsey, violating the choice of

G. We have shown that

$$\frac{\mathrm{ex}^r(n; H_1, \dots, H_r)}{\binom{n}{k}}$$

is a non-increasing function of n (which is bounded below, by 0), and so this function has a limit. Therefore we may define the *r*-coloured Turán density $\pi(H_1, \ldots, H_r)$ of (H_1, \ldots, H_r) by

$$\pi(H_1,\ldots,H_r) := \lim_{n \to \infty} rac{\mathrm{ex}^r(n;H_1,\ldots,H_r)}{\binom{n}{k}}.$$

As for $k \ge 3$, the problem of determining $\pi(H)$ is still out of reach, we certainly cannot evaluate $\pi(H_1, \ldots, H_r)$ in general. However, any non- (H_1, \ldots, H_r) -Ramsey graph is $K_s^{(k)}$ free, where $s := R(H_1, \ldots, H_r)$ is the smallest integer m such that $K_m^{(k)}$ is (H_1, \ldots, H_r) -Ramsey. Thus

$$\pi(H_1,\ldots,H_r) \leqslant \pi(K_s^{(k)}),\tag{5.4.3}$$

which is at most $1 - {\binom{s-1}{k-1}}^{-1}$ (de Caen [29]). An interesting question is for which H_1, \ldots, H_r the inequality in (5.4.3) is tight. We discuss the case k = 2 in detail in Section 5.4.3.

We now state the main result of this subsection, which generalises Theorem 5.26 to $r \ge 1$.

Theorem 5.27. For all $\delta > 0$, integers $r \ge 1$ and $k \ge 2$, and k-graphs H_1, \ldots, H_r , there exist $n_0, \varepsilon > 0$ such that for all integers $n \ge n_0$, every n-vertex k-graph G with $e(G) \ge (\pi(H_1, \ldots, H_r) + \delta) \binom{n}{k}$ is ε -strongly (H_1, \ldots, H_r) -Ramsey.

Proof. Let $\delta > 0$ and let r, k be positive integers with $k \ge 2$. By the definition of $\pi(\cdot)$, there exists $m_0 > 0$ such that for all integers $m \ge m_0$,

$$\operatorname{ex}^{r}(m; H_{1}, \ldots, H_{r}) < \left(\pi(H_{1}, \ldots, H_{r}) + \frac{\delta}{2}\right) \binom{m}{k}.$$

Fix an integer $m \ge m_0$. Without loss of generality, we may assume that $m \ge v(H_i)$ for

all $i \in [r]$. Choose $\varepsilon > 0$ to be such that

$$\varepsilon \leqslant \frac{\delta}{2r} \binom{m}{v(H_i)}^{-1}$$

for all $i \in [r]$. Let *n* be an integer which is sufficiently large compared to *m*, and let *G* be a *k*-graph on *n* vertices with $e(G) = (\pi(H_1, \ldots, H_r) + \delta) \binom{n}{k}$. We need to show that, for every *r*-colouring σ of *G*, there is $i \in [r]$ such that $\sigma^{-1}(i)$ contains at least $\varepsilon \binom{n}{v(H_i)}$ copies of H_i ; so fix an arbitrary σ .

Define \mathcal{M} to be the set of $M \in \binom{V(G)}{m}$ such that $e(G[M]) \ge (\pi(H_1, \dots, H_r) + \frac{\delta}{2})\binom{m}{k}$. Then

$$\sum_{U \subseteq V(G): |U|=m} e(G[U]) \leq |\mathcal{M}| \binom{m}{k} + \left(\binom{n}{m} - |\mathcal{M}|\right) \left(\pi(H_1, \dots, H_r) + \frac{\delta}{2}\right) \binom{m}{k}$$

But for every $e \in E(G)$, there are exactly $\binom{n-k}{m-k}$ sets $U \subseteq V(G)$ with |U| = m such that $e \in E(G[U])$. Thus also

$$\sum_{U \subseteq V(G):|U|=m} e(G[U]) \ge {\binom{n-k}{m-k}} (\pi(H_1, \dots, H_r) + \delta) {\binom{n}{k}}$$
$$= (\pi(H_1, \dots, H_r) + \delta) {\binom{n}{m}} {\binom{m}{k}},$$

and so, rearranging, we have $|\mathcal{M}| \ge \delta\binom{n}{m}/2$. By the choice of m, for every $M \in \mathcal{M}$, there exists $i = i(M) \in [r]$ such that $\sigma^{-1}(i)$ contains a copy of H_i with vertices in M. Choose $\mathcal{M}' \subseteq \mathcal{M}$ such that the i(M') are equal for all $M' \in \mathcal{M}'$ and $|\mathcal{M}'| \ge |\mathcal{M}|/r$. Without loss of generality let us assume that i(M') = 1 for all $M' \in \mathcal{M}'$. So for each $M' \in \mathcal{M}'$, there is a copy of $H_1 \subseteq G[M']$ which is monochromatic with colour 1 under σ . Each such copy has vertex set contained in at most $\binom{n-v(H_1)}{m-v(H_1)}$ sets $M' \in \mathcal{M}'$. Thus the number of such monochromatic copies of H_1 in G is at least

$$\frac{\frac{\delta}{2} \cdot \binom{n}{m}}{r\binom{n-v(H_1)}{m-v(H_1)}} = \frac{\delta}{2r} \cdot \binom{m}{v(H_1)}^{-1} \cdot \binom{n}{v(H_1)} \ge \varepsilon \binom{n}{v(H_1)}.$$

So G is ε -strongly (H_1, \ldots, H_r) -Ramsey, as required.

5.4.3 The special case of graphs: maximum size and typical structure

The intimate connection between forbidden subgraphs and chromatic number when k = 2allows us to make some further remarks here. (This section is separate from the remainder of the chapter and the results stated here will not be required later on.)

5.4.3.1 The maximum number of edges in a graph which is not Ramsey

Given $s, n \in \mathbb{N}$, let $T_s(n)$ denote the *s*-partite Turán (2-)graph on *n* vertices; that is, the vertex set of $T_s(n)$ has a partition into *s* parts V_1, \ldots, V_s such that $||V_i| - |V_j|| \leq 1$ for all $i, j \in [s]$; and xy is an edge of $T_s(n)$ if and only if there are $ij \in {[s] \choose 2}$ such that $x \in V_i$ and $y \in V_j$. Write $t_s(n) := e(T_s(n))$.

We need to define two notions of Ramsey number.

Given an integer $r \ge 1$ and families $\mathcal{H}_1, \ldots, \mathcal{H}_r$ of graphs, the Ramsey number $R(\mathcal{H}_1, \ldots, \mathcal{H}_r)$ is the least m such that any r-colouring of K_m contains an i-coloured copy of H_j for some $i \in [r]$ and some $H_j \in \mathcal{H}_i$. If $\mathcal{H}_i = \{K_{\ell_i}\}$ for all $i \in [r]$ then we instead write $R(\ell_1, \ldots, \ell_r)$, and simply $R^r(\ell)$ in the case when $\ell_1 = \ldots = \ell_r =: \ell$. Given graphs H_1, \ldots, H_r , the chromatic Ramsey number $R_{\chi}(H_1, \ldots, H_r)$ is the least m for which there exists an (H_1, \ldots, H_r) -Ramsey graph with chromatic number m.

Trivially, for any k-graph H, we have that $R_{\chi}(H) = \chi(H)$. If H_1, \ldots, H_r are graphs,

then

$$t_{R_{\chi}(H_1,\dots,H_r)-1}(n) \leq \exp^r(n; H_1,\dots,H_r) \leq t_{R_{\chi}(H_1,\dots,H_r)-1}(n) + o(n^2).$$
 (5.4.4)

Thus

$$\pi(H_1, \dots, H_r) = 1 - \frac{1}{R_{\chi}(H_1, \dots, H_r) - 1} = \pi \left(K_{R_{\chi}(H_1, \dots, H_r)} \right).$$
(5.4.5)

The first inequality in (5.4.4) follows by definition of $ex^r(n; H_1, \ldots, H_r)$; the second from (5.4.2) applied with a graph H which is (H_1, \ldots, H_r) -Ramsey and has $\chi(H) = R_{\chi}(H_1, \ldots, H_r)$. Clearly, then, $\pi(H_1, \ldots, H_r) = \pi(J_1, \ldots, J_r)$ if and only if $R_{\chi}(J_1, \ldots, J_r) = R_{\chi}(H_1, \ldots, H_r)$. So, in the graph case, the inequality (5.4.3) is tight when the Ramsey number and chromatic Ramsey number coincide.

As noted by Bialostocki, Caro and Roditty [16], one can determine $ex^r(n; H_1, \ldots, H_r)$ exactly in the case when H_1, \ldots, H_r are cliques of equal size.

Theorem 5.28 ([16]). For all positive integers $\ell, n \ge 3$ and $r \ge 1$, we have $ex^r(n; K_\ell, \ldots, K_\ell) = t_{R^r(\ell)-1}(n)$.

Thus in this case (5.4.3) is tight. The chromatic Ramsey number was introduced by Burr, Erdős and Lovász [21] who showed that, in principle, one can determine R_{χ} given the usual Ramsey number R. A graph homomorphism from a graph H to a graph K is a function $\phi : V(H) \to V(K)$ such that $\phi(x)\phi(y) \in E(K)$ whenever $xy \in E(H)$. Let $\operatorname{Hom}(H)$ denote the set of all graphs K such that there exists a graph homomorphism ϕ for which $K = \phi(H)$. Since there exists a homomorphism from H into K_{ℓ} if and only if $\chi(H) \leq \ell$, we also have that $R(\operatorname{Hom}(H)) = \chi(H)$. Thus $R(\operatorname{Hom}(H)) = R_{\chi}(H)$. In fact this relationship extends to all $r \geq 1$.

Lemma 5.29 ([21, 26, 78]). For all integers $r \in \mathbb{N}$ and graphs H_1, \ldots, H_r ,

$$R_{\chi}(H_1,\ldots,H_r) = R(\operatorname{Hom}(H_1),\ldots,\operatorname{Hom}(H_r)).$$

Moreover, for all integers $\ell_1, \ldots, \ell_r \ge 3$, we have that

$$R_{\chi}(K_{\ell_1},\ldots,K_{\ell_r})=R(\ell_1,\ldots,\ell_r)$$

The second statement is a corollary of the first since $\text{Hom}(K_{\ell}) = \{K_{\ell}\}$. Another observation (see [21]) is that for all $\ell \in \mathbb{N}$, the chromatic Ramsey number $R_{\chi}(C_{2\ell+1}, C_{2\ell+1})$ is equal to 5 if $\ell = 2$, and equal to 6 otherwise.

The first inequality in (5.4.4) is not always tight, for example when H is the disjoint union of two copies of some graph G. Indeed, $\operatorname{Hom}(H) \supseteq \operatorname{Hom}(G)$ and so $R_{\chi}(H, \ldots, H) =$ $R_{\chi}(G, \ldots, G)$. Let F be an n-vertex graph with $e(F) = \exp^{r}(n; G, \ldots, G)$ which is not (G, r)-Ramsey. Obtain a graph T by adding an edge e to F. Then there exists an rcolouring of T in which every monochromatic copy of G contains e (the monochromatic-G-free colouring of F, with e arbitrarily coloured). Hence T is not (H, r)-Ramsey and so

$$\operatorname{ex}^{r}(n; H, \dots, H) > \operatorname{ex}^{r}(n; G, \dots, G) \ge t_{R_{\chi}(G, \dots, G)}(n) = t_{R_{\chi}(H, \dots, H)}(n).$$

We say that a graph H is *(weakly) colour-critical* if there exists $e \in E(H)$ for which $\chi(H-e) < \chi(H)$. Complete graphs and odd cycles are examples of colour-critical graphs. The following conjecture would generalise Theorem 5.28 to provide a large class of graphs where the first inequality in (5.4.4) is tight.

Conjecture 5.30. Let r be a positive integer and H a colour-critical graph. Then, whenever n is sufficiently large,

$$ex^{r}(n; H, ..., H) = t_{R_{\chi}(H,...,H)-1}(n).$$

If true, this conjecture would also generalise a well-known result of Simonovits [111] which extends Turán's theorem to colour-critical graphs. It would also determine $ex^r(n; H, ..., H)$ explicitly whenever H is an odd cycle.

5.4.3.2 The typical structure of non-Ramsey graphs

There has been much interest in determining the *typical* structure of an *H*-free graph. For example, Kolaitis, Prömel and Rothschild [73] proved that almost all K_r -free graphs are (r-1)-partite. It turns out that one can easily obtain a result on the typical structure of non-Ramsey graphs from a result of Prömel and Steger [87].

Given two families $\mathcal{A}(n), \mathcal{B}(n)$ of *n*-vertex graphs such that $\mathcal{B}(n) \subseteq \mathcal{A}(n)$, we say that almost all *n*-vertex graphs $G \in \mathcal{A}(n)$ are in $\mathcal{B}(n)$ if

$$\lim_{n \to \infty} \frac{|\mathcal{A}(n)|}{|\mathcal{B}(n)|} = 1.$$

The next result of Prömel and Steger [87] immediately tells us the typical structure of non-Ramsey graphs in certain cases.

Theorem 5.31 ([87]). For every graph H, the following holds. Almost all H-free graphs are $(\chi(H) - 1)$ -partite if and only if H is colour-critical.

Corollary 5.32. For all integers r and graphs H_1, \ldots, H_r , if there exists an (H_1, \ldots, H_r) -Ramsey graph H such that $\chi(H) = R_{\chi}(H_1, \ldots, H_r)$ and H is colour-critical, then almost every non- (H_1, \ldots, H_r) -Ramsey graph is $(R_{\chi}(H_1, \ldots, H_r) - 1)$ -partite.

Proof. The result follows since every non- (H_1, \ldots, H_r) -Ramsey graph G is H-free, and every $(R_{\chi}(H_1, \ldots, H_r) - 1)$ -partite graph is non- (H_1, \ldots, H_r) -Ramsey.

In particular, if in Corollary 5.32, each H_i is a clique, say $H_i = K_{\ell_i}$, then by Lemma 5.29 we can take $H := K_{R(\ell_1,...,\ell_r)}$. So, for example, almost every non- $(K_3, 2)$ -Ramsey graph is 5-partite.

5.4.4 A container theorem for Ramsey hypergraphs

Recall that $\overline{\operatorname{Ram}}(n; H_1, \ldots, H_r)$ is the set of *n*-vertex *k*-graphs which are not (H_1, \ldots, H_r) -Ramsey and $\operatorname{Ram}(H_1, \ldots, H_r)$ is the set of (H_1, \ldots, H_r) -Ramsey *k*-graphs (on any number of vertices). Recall further that an *H*-free *k*-graph is precisely a non-(H, 1)-Ramsey graph. Write $\mathcal{G}_k(n)$ for the set of all *k*-graphs on vertex set [n]. Let $\mathcal{I}_r(n; H_1, \ldots, H_r)$ denote the set of all ordered *r*-tuples $(G_1, \ldots, G_r) \in (\mathcal{G}_k(n))^r$ of *k*-graphs such that each G_i is H_i -free and $E(G_i) \cap E(G_j) = \emptyset$ for all distinct $i, j \in [r]$. Note that for any $G \in \overline{\operatorname{Ram}}(n; H_1, \ldots, H_r)$, there exist pairwise edge-disjoint *k*-graphs G_1, \ldots, G_r such that $\bigcup_{i \in [r]} G_i = G$ and $(G_1, \ldots, G_r) \in \mathcal{I}_r(n; H_1, \ldots, H_r)$. Given an integer $k \ge 2$, a *k*-graph *H* and positive integer *n*, the hypergraph \mathcal{H} of copies of *H* in $K_n^{(k)}$ has vertex set $V(\mathcal{H}) := {n \choose k}$, and $E \subseteq {V(\mathcal{H}) \choose e(H)}$ is an edge of \mathcal{H} if and only if *E* is isomorphic to E(H).

In this subsection, we prove a container theorem for elements in $\mathcal{I}_r(n; H_1, \ldots, H_r)$. To do so, we will apply Theorem 5.15 to hypergraphs $\mathcal{H}_1, \ldots, \mathcal{H}_r$, where \mathcal{H}_i is the hypergraph of copies of H_i In \mathcal{H}_i , an independent set corresponds to an H_i -free k-graph.

We will need the following simple proposition from [8].

Proposition 5.33 ([8], Proposition 7.3). Let H be a k-graph. Then there exists c > 0 such that, for all positive integers n, the following holds. Let \mathcal{H} be the e(H)-uniform hypergraph of copies of H in $K_n^{(k)}$. Then, letting $p = n^{-1/m_k(H)}$,

$$\Delta_{\ell}(\mathcal{H}) \leqslant c \cdot p^{\ell-1} \frac{e(\mathcal{H})}{v(\mathcal{H})},$$

for every $\ell \in [e(H)]$.

We can now prove our container theorem for elements in $\mathcal{I}_r(n; H_1, \ldots, H_r)$.

Theorem 5.34. Let $r, k \in \mathbb{N}$ with $k \ge 2$ and $\delta > 0$. Let H_1, \ldots, H_r be k-graphs such that $m_k(H_1) \ge \ldots \ge m_k(H_r)$ and $\Delta_1(H_i) \ge 2$ for all $i \in [r]$. Then there exists D > 0 such

that the following holds. For all $n \in \mathbb{N}$, there is a collection $S_r \subseteq (\mathcal{G}_k(n))^r$ and a function $f: S_r \to (\mathcal{G}_k(n))^r$ such that:

- (i) For all $(I_1, \ldots, I_r) \in \mathcal{I}_r(n; H_1, \ldots, H_r)$, there exists $S \in \mathcal{S}_r$ such that $S \subseteq (I_1, \ldots, I_r) \subseteq f(S)$.
- (ii) If $(S_1, \ldots, S_r) \in \mathcal{S}_r$ then $\sum_{i \in [r]} e(S_i) \leq Dn^{k-1/m_k(H_1)}$.
- (iii) Every $S \in \mathcal{S}_r$ satisfies $S \in \mathcal{I}_r(n; H_1, \ldots, H_r)$.
- (iv) Given any $S = (S_1, \ldots, S_r) \in \mathcal{S}_r$, write $f(S) =: (f(S_1), \ldots, f(S_r))$. Then
 - (a) $\bigcup_{i \in [r]} f(S_i)$ is not δ -strongly (H_1, \dots, H_r) -Ramsey; and (b) $e\left(\bigcup_{i \in [r]} f(S_i)\right) \leq \exp^r(n; H_1, \dots, H_r) + \delta\binom{n}{k}.$

Note that if H is a k-graph with $\Delta_1(H) = 1$, then H is a matching, i.e. a set of vertex-disjoint edges.

Proof. We will identify any hypergraph which has vertex set [n] with its edge set. We may further assume that there are no isolated vertices in H_i for any $i \in [r]$.

Apply Proposition 5.33 with input hypergraphs H_1, \ldots, H_r to obtain c > 0 such that its conclusion holds with H_i playing the role of H, for all $i \in [r]$. Let $\delta > 0$, $r \in \mathbb{N}$ and $k \ge 2$ be given and apply Theorem 5.27 (with $\delta/2$ playing the role of δ) to obtain $n_0, \varepsilon > 0$. Without loss of generality we may assume $\varepsilon \le \delta < 1$. For each $i \in [r]$, let $v_i := v(H_i)$ and $m_i := e(H_i)$ for all $i \in [r]$. Set $v := \max_{i \in [r]} v_i$; $m := \max_{i \in [r]} m_i$;

$$\varepsilon' := \frac{\varepsilon}{2 \cdot v!}; \text{ and } \varepsilon'' := \frac{\varepsilon'}{\binom{v}{k} \cdot v!}$$

Apply Theorem 5.15 with parameters $r, m_1, \ldots, m_r, c, \varepsilon''$ playing the roles of $r, k_1, \ldots, k_r, c, \varepsilon$ respectively to obtain $D_1 > 0$. Increase n_0 if necessary so that $0 < 1/n_0 \ll 1/D_1, 1/k, 1/r, \varepsilon, \delta$. If $n < n_0$, then set S_r to be $\mathcal{I}_r(n; H_1, \ldots, H_r)$; set f to be the identity function and choose D_2 to be large. By setting D to be the maximum of D_1 and D_2 , it remains to prove the result for integers $n \ge n_0$. So now fix $n \ge n_0$.

Let $\mathcal{H}_{n,i}$ be the hypergraph of copies of H_i in $K_n^{(k)}$. That is, $V(\mathcal{H}_{n,i}) := {\binom{[n]}{k}}$ and for each m_i -subset E of ${\binom{[n]}{k}}$, put $E \in E(\mathcal{H}_{n,i})$ if and only if E is isomorphic to a copy of H_i . By definition, $\mathcal{H}_{n,i}$ is an m_i -uniform hypergraph and an independent set in $\mathcal{H}_{n,i}$ corresponds to an H_i -free k-graph with vertex set [n]. Since H_i is a k-graph with no isolated vertices,

$$e(\mathcal{H}_{n,i}) = \frac{v_i!}{|\operatorname{Aut}(H_i)|} \binom{n}{v_i}$$
(5.4.6)

where $\operatorname{Aut}(H_i)$ is the automorphism group of H_i . For all $i \in [r]$, let

$$\mathcal{F}_{n,i} := \left\{ A \subseteq \binom{[n]}{k} : e(\mathcal{H}_{n,i}[A]) \geqslant \varepsilon' e(\mathcal{H}_{n,i}) \right\}.$$

We claim that $\mathcal{H}_{n,1}, \ldots, \mathcal{H}_{n,r}$ and $\mathcal{F}_{n,1}, \ldots, \mathcal{F}_{n,r}$ satisfy the hypotheses of Theorem 5.15 with the parameters chosen as above and with

$$p = p(n) := n^{-1/m_k(H_1)}$$

Clearly each family $\mathcal{F}_{n,i}$ is increasing, and $\mathcal{H}_{n,i}$ is $(\mathcal{F}_{n,i}, \varepsilon')$ -dense. Next, we show that $|A| \ge \varepsilon''\binom{n}{k}$ for all $A \in \mathcal{F}_{n,i}$. In any k-graph on n vertices, there are at most $v_i!\binom{n-k}{v_i-k}$ copies of H_i that contain some fixed set $\{x_1, \ldots, x_k\}$ of vertices. Therefore, for every $e \in \binom{[n]}{k}$, the number of $E \in E(\mathcal{H}_{n,i})$ containing e is at most

$$v_i! \binom{n-k}{v_i-k}.$$
(5.4.7)

Thus every $A \in \mathcal{F}_{n,i}$ satisfies

$$|A| \ge \frac{e(\mathcal{H}_{n,i}[A])}{v_i!\binom{n-k}{v_i-k}} \stackrel{(5.4.6)}{\ge} \frac{\varepsilon' v_i!\binom{n}{v_i}}{v_i!\binom{n-k}{v_i-k}|\operatorname{Aut}(H_i)|} = \frac{\varepsilon'}{\binom{v}{k}|\operatorname{Aut}(H_i)|}\binom{n}{k} \ge \varepsilon''\binom{n}{k},$$

where, in the final inequality, we used the fact that $|\operatorname{Aut}(H_i)| \leq v_i!$. Note that $\varepsilon'' < \varepsilon'$. So $\mathcal{H}_{n,i}$ is $(\mathcal{F}_{n,i}, \varepsilon'')$ -dense and $|A| \geq \varepsilon'' \binom{n}{k}$ for all $A \in \mathcal{F}_{n,i}$.

Certainly $p \ge n^{-1/m_k(H_j)}$ for all $j \in [r]$. By the choice of c, we then have

$$\Delta_{\ell}(\mathcal{H}_{n,i}) \leqslant c \cdot p^{\ell-1} \frac{e(\mathcal{H}_{n,i})}{\binom{n}{k}}$$

for all $i \in [r]$ and $\ell \in [m_i]$. We have shown that $\mathcal{H}_{n,i}$ and $\mathcal{F}_{n,i}$ satisfy the hypotheses of Theorem 5.15 for all $i \in [r]$.

Then Theorem 5.15 implies that there exists a family $S_r \subseteq \mathcal{I}(\mathcal{H}_{n,1},\ldots,\mathcal{H}_{n,r})$ and functions $f': S_r \to \prod_{i \in [r]} \overline{\mathcal{F}_{n,i}}$ and $g: \mathcal{I}(\mathcal{H}_{n,1},\ldots,\mathcal{H}_{n,r}) \to S_r$ such that the following conditions hold:

- (a) If $(S_1, \ldots, S_r) \in \mathcal{S}_r$ then $\sum |S_i| \leq D_1 p\binom{n}{k}$;
- (b) every $S \in \mathcal{S}_r$ satisfies $S \in \mathcal{I}(\mathcal{H}_{n,1}, \ldots, \mathcal{H}_{n,r})$;
- (c) for every $(I_1, \ldots, I_r) \in \mathcal{I}(\mathcal{H}_{n,1}, \ldots, \mathcal{H}_{n,r})$, we have that $S \subseteq (I_1, \ldots, I_r) \subseteq S \cup f'(S)$, where $S := g(I_1, \ldots, I_r)$.

Note that $(G_1, \ldots, G_r) \in \mathcal{I}(\mathcal{H}_{n,1}, \ldots, \mathcal{H}_{n,r})$ if and only if $(G_1, \ldots, G_r) \in \mathcal{I}_r(n; H_1, \ldots, H_r)$ (where we recall the identification of graphs and edge sets). For each $S \in \mathcal{S}_r$, define

$$f(S) := S \cup f'(S).$$

So $f: \mathcal{S}_r \to \mathcal{P}(\binom{[n]}{k})^r$. (Note that under the correspondence of graphs and edge sets we can view $\mathcal{P}(\binom{[n]}{k})^r = (\mathcal{G}_k(n))^r$.) Thus (a)–(c) immediately imply that (i) and (iii) hold,

and additionally for any $(S_1, \ldots, S_r) \in \mathcal{S}_r$ we have

$$\sum_{i \in [r]} e(S_i) \leqslant D_1 p\binom{n}{k} \leqslant D_1 n^{-1/m_k(H_1)} \cdot \frac{n^k}{k!} < D_1 n^{k-1/m_k(H_1)}$$

yielding (ii).

Given any $S = (S_1, \ldots, S_r) \in S_r$ write $f(S) =: (f(S_1), \ldots, f(S_r))$ and $f'(S) =: (f'(S_1), \ldots, f'(S_r))$. Let $G := \bigcup_{i \in [r]} f(S_i)$; so G is a k-graph with vertex set [n]. To prove (iv)(a), we need to exhibit an r-colouring σ of G with the property that $\sigma^{-1}(i)$ contains less than $\varepsilon {n \choose v_i}$ copies of H_i for all $i \in [r]$. Indeed, consider the r-colouring σ of G defined by setting $\sigma(e) = i$ when i is the least integer such that $e \in f(S_i)$. Then the subgraph of G coloured i is $\sigma^{-1}(i) \subseteq f(S_i) = S_i \cup f'(S_i)$. Since S_i is an independent set in $\mathcal{H}_{n,i}$, we have that S_i is H_i -free. Every copy of H_i in $\sigma^{-1}(i)$ either contains at least one edge in S_i , or has every edge contained in $f'(S_i)$. Note that $m_k(H_i) \leq m$. By (5.4.7), the number of copies of H_i in G containing at least one edge in S_i is at most

$$e(S_i) \cdot v_i! \binom{n-k}{v_i-k} \leqslant D_1 n^{k-1/m_k(H_1)} \cdot v_i! (n-k)^{v_i-k} \leqslant D_1 v_i! \cdot n^{v_i-\frac{1}{m}} < \frac{\varepsilon}{2} \binom{n}{v_i}.$$

For each $i \in [r]$ we have that $f'(S_i) \in \overline{\mathcal{F}_{n,i}}$, and so $e(\mathcal{H}_{n,i}[f'(S_i)]) < \varepsilon' e(\mathcal{H}_{n,i})$. That is, the number of copies of H_i in $f'(S_i)$ is less than

$$\varepsilon' \cdot \frac{v_i!}{|\operatorname{Aut}(H_i)|} \binom{n}{v_i} \leqslant \frac{\varepsilon}{2} \binom{n}{v_i}.$$

Thus, in total $f(S_i) = S_i \cup f'(S_i)$ contains at most $\varepsilon \binom{n}{v_i}$ copies of H_i , so G is not ε -strongly (H_1, \ldots, H_r) -Ramsey. Since $\varepsilon \leq \delta$, this immediately implies (iv)(a), and (iv)(b) follows from Theorem 5.27, our choice of parameters, and since n is sufficiently large.

As in Theorem 2.1, we will call the elements $S \in S_r$ fingerprints, and each $\bigcup_{i \in [r]} f(S_i)$ with $(S_1, \ldots, S_r) \in S_r$ is a *container*.

5.4.5 The number of hypergraphs which are not Ramsey

Our first application of Theorem 5.34 is an enumeration result for non- (H_1, \ldots, H_r) -Ramsey hypergraphs (Theorem 5.12), which asymptotically determines the logarithm of $|\overline{\text{Ram}}(n; H_1, \ldots, H_r)|$.

Proof of Theorem 5.12. Let $0 < \delta < 1$ be arbitrary, and let $n \in \mathbb{N}$ be sufficiently large. Clearly, $|\overline{\text{Ram}}(n; H_1, \ldots, H_r)| \ge 2^{\exp^r(n; H_1, \ldots, H_r)}$ since no subhypergraph of an *n*-vertex non- (H_1, \ldots, H_r) -Ramsey *k*-graph with $\exp^r(n; H_1, \ldots, H_r)$ edges is (H_1, \ldots, H_r) -Ramsey.

For the upper bound, suppose first that $\Delta_1(H_i) \ge 2$ for all $i \in [r]$. Let D > 0be obtained from Theorem 5.34 applied to H_1, \ldots, H_r with parameter δ . We obtain a collection S_r and a function f as in Theorem 5.34. Consider any $G \in \overline{\text{Ram}}(n; H_1, \ldots, H_r)$. Note that there are pairwise edge-disjoint k-graphs G_1, \ldots, G_r such that $\bigcup_{i \in [r]} G_i = G$ and $(G_1, \ldots, G_r) \in \mathcal{I}_r(n; H_1, \ldots, H_r)$. So by Theorem 5.34(i) this means there is some $S = (S_1, \ldots, S_r) \in S_r$ so that $G \subseteq \bigcup_{i \in [r]} f(S_i)$. Further, given any $S = (S_1, \ldots, S_r) \in S_r$, we have

$$e\left(\bigcup_{i\in[r]}f(S_i)\right)\leqslant \operatorname{ex}^r(n;H_1,\ldots,H_r)+\delta\binom{n}{k}.$$

Thus, each such $\bigcup_{i \in [r]} f(S_i)$ contains at most $2^{\exp^r(n;H_1,\ldots,H_r)+\delta\binom{n}{k}} k$ -graphs in $\overline{\operatorname{Ram}}(n;H_1,\ldots,H_r)$. Note that, by Theorem 5.34(ii),

$$|\mathcal{S}_r| \leqslant \left(\sum_{s=0}^{Dn^{k-1/m_k(H_1)}} \binom{\binom{n}{k}}{s}\right)^r < 2^{\delta\binom{n}{k}},$$

where the last inequality holds since n is sufficiently large. Altogether, this implies

$$|\overline{\text{Ram}}(n; H_1, \dots, H_r)| \leqslant 2^{\delta\binom{n}{k}} \times 2^{\exp^r(n; H_1, \dots, H_r) + \delta\binom{n}{k}} = 2^{\exp^r(n; H_1, \dots, H_r) + 2\delta\binom{n}{k}}.$$
 (5.4.8)

Since the choice of $0 < \delta < 1$ was arbitrary, this proves the theorem in the case when

 $\Delta_1(H_i) \ge 2$ for all $i \in [r]$.

Suppose now that, say, $\Delta_1(H_1) = 1$. Then H_1 is a matching. Certainly every non-(H_2, \ldots, H_r)-Ramsey k-graph is non-(H_1, \ldots, H_r)-Ramsey. Let $H \in \overline{\text{Ram}}(n; H_1, \ldots, H_r)$. Then there exists an r-colouring σ of H such that $\sigma^{-1}(i)$ is H_i -free for all $i \in [r]$. Thus His the union of pairwise edge-disjoint k-graphs $J \in \overline{\text{Ram}}(n; H_2, \ldots, H_r)$ and $J' := \sigma^{-1}(1)$. But J' is H_1 -free and hence does not contain a matching of size $\lfloor v(H_1)/2 \rfloor =: h$. A result of Erdős [33] (used here in a weaker form) implies that, for sufficiently large n,

$$e(J') \leqslant (h-1)\binom{n-1}{k-1}$$

Thus, for large n,

$$|\overline{\operatorname{Ram}}(n; H_1, \dots, H_r)| \leq \sum_{J \in \overline{\operatorname{Ram}}(n; H_2, \dots, H_r)} \sum_{e(J')=0}^{(h-1)\binom{n-1}{k-1}} \binom{\binom{n}{k}}{e(J')}$$
$$= |\overline{\operatorname{Ram}}(n; H_2, \dots, H_r)| \sum_{e(J')=0}^{\frac{k(h-1)}{n}\binom{n}{k}} \binom{\binom{n}{k}}{e(J')}$$
$$\leq |\overline{\operatorname{Ram}}(n; H_2, \dots, H_r)| \cdot 2^{\delta\binom{n}{k}}.$$

Iterating this argument, using (5.4.8) and the fact that $0 < \delta < 1$ was arbitrary, we obtain the required upper bound in the general case.

In fact Theorem 5.12 can be recovered in a different way, which, to the best of our knowledge, has not been explicitly stated elsewhere. Let \mathcal{F} be a (possibly infinite) family of k-graphs, and let $Forb(n; \mathcal{F})$ be the set of n-vertex k-graphs which contain no copy of any $F \in \mathcal{F}$ as a subhypergraph. The following result of Nagle, Rödl and Schacht [84] asymptotically determines the logarithm of $|Forb(n; \mathcal{F})|$. (This generalises the corresponding result of Erdős, Frankl and Rödl [34] for graphs.) Let

$$ex(n; \mathcal{F}) := max\{e(H) : H \in Forb(n; \mathcal{F})\}.$$

(So when $\mathcal{F} = \{F\}$ contains a single k-graph, we have $ex(n; \{F\}) = ex(n; F)$.)

Theorem 5.35 (Theorem 2.3, [84]). Let $k \ge 2$ be a positive integer and \mathcal{F} be a (possibly infinite) family of k-graphs. Then, for all $n \in \mathbb{N}$,

$$|\operatorname{Forb}(n;\mathcal{F})| = 2^{\operatorname{ex}(n;\mathcal{F})+o(n^k)}.$$

Since $G \in \text{Ram}(n; H_1, \ldots, H_r)$ if and only if G is an *n*-vertex k-graph without a copy of any $F \in \text{Ram}(H_1, \ldots, H_r)$ as a subhypergraph, Theorem 5.35 immediately implies Theorem 5.12.

We remark that the proof of Nagle, Rödl and Schacht [84] uses hypergraph regularity. Our proof of Theorem 5.12 has the advantage that it is is regularity-free.

5.4.6 The resilience of being (H_1, \ldots, H_r) -Ramsey

Recall that $G_{n,p}^{(k)}$ has vertex set [n], where each edge lies in $\binom{[n]}{k}$ and appears with probability p, independently of all other edges. In this section we apply Theorem 5.34 to prove Theorem 5.7, which determines $\operatorname{res}(G_{n,p}^{(k)}, (H_1, \ldots, H_r)$ -Ramsey) for given fixed k-graphs H_1, \ldots, H_r . Explicitly, $\operatorname{res}(G_{n,p}^{(k)}, (H_1, \ldots, H_r)$ -Ramsey) is the minimum integer t such that one can remove t edges from $G_{n,p}^{(k)}$ to obtain a k-graph H which has an (H_1, \ldots, H_r) -free r-colouring.

Observe that Theorem 5.7 together with (5.4.3) immediately implies the following corollary.

Corollary 5.36 (Random Ramsey for hypergraphs). For all positive integers r, kwith $k \ge 2$ and k-graphs H_1, \ldots, H_r with $m_k(H_1) \ge \ldots \ge m_k(H_r)$ and $\Delta_1(H_i) \ge 2$ for all $i \in [r]$, there exists C > 0 such that

$$\lim_{n \to \infty} \mathbb{P} \left[G_{n,p}^{(k)} \text{ is } (H_1, \dots, H_r) \text{-}Ramsey \right] = 1 \quad \text{if } p > Cn^{-1/m_k(H_1)}.$$

In the case when $m_k(H_1) = m_k(H_2)$, Corollary 5.36 generalises Theorem 5.6 since we do not require H_1 to be strictly k-balanced. Further, Corollary 5.36 resolves (the 1-statement part) of Conjecture 5.3 in the case when $m_2(H_1) = m_2(H_2)$.

Proof of Theorem 5.7. Let $0 < \delta < 1$ be arbitrary, $r, k \in \mathbb{N}$ with $k \ge 2$, and let H_1, \ldots, H_r be k-graphs as in the statement of the theorem. Given $n \in \mathbb{N}$, if $p > n^{-1/m_k(H_1)}$, then $p > n^{-(k-1)}$ since $\Delta_1(H_1) \ge 2$. Proposition 1.2(ii) implies that, w.h.p.,

$$e(G_{n,p}^{(k)}) = \left(1 \pm \frac{\delta}{4}\right) p\binom{n}{k}.$$
(5.4.9)

For brevity, write $\pi := \pi(H_1, \ldots, H_r)$. We will first prove the upper bound

$$\lim_{n \to \infty} \mathbb{P}\left[\operatorname{res}(G_{n,p}^{(k)}, (H_1, \dots, H_r) - \operatorname{Ramsey}) \leqslant (1 - \pi + \delta) e(G_{n,p}^{(k)}) \right] = 1 \quad \text{if} \quad p > n^{-1/m_k(H_1)}.$$

For this, we must show that the probability of the event that there exists an *n*-vertex *k*-graph $G \subseteq G_{n,p}^{(k)}$ such that $e(G) \ge (\pi - \delta)e(G_{n,p}^{(k)})$ and $G \in \overline{\text{Ram}}(n; H_1, \ldots, H_r)$, tends to one as *n* tends to infinity. This indeed follows: Let *n* be sufficiently large so that $ex^r(n; H_1, \ldots, H_r) \ge (\pi - \delta/2) {n \choose k}$. Let G^* be an *n*-vertex non- (H_1, \ldots, H_r) -Ramsey *k*graph with $e(G^*) = ex^r(n; H_1, \ldots, H_r)$. Then, by Proposition 1.2(ii), w.h.p. we have $e(G^* \cap G_{n,p}^{(k)}) = (\pi \pm \delta)e(G_{n,p}^{(k)})$, and $G^* \cap G_{n,p}^{(k)} \in \overline{\text{Ram}}(n; H_1, \ldots, H_r)$, as required.

For the remainder of the proof, we will focus on the lower bound, namely that there exists C > 0 such that whenever $p > Cn^{-1/m_k(H_1)}$,

$$\mathbb{P}\left[\operatorname{res}(G_{n,p}^{(k)}, (H_1, \dots, H_r) - \operatorname{Ramsey}) \ge (1 - \pi - \delta)e(G_{n,p}^{(k)})\right] \to 1 \quad \text{as} \quad n \to \infty.$$
(5.4.10)

Suppose *n* is sufficiently large. Apply Theorem 5.34 with parameters $r, k, \delta/16, (H_1, \ldots, H_r)$ to obtain D > 0 and for each $n \in \mathbb{N}$, a collection S_r and a function f satisfying (i)–(iv). Now choose C such that $0 < 1/C \ll 1/D, \delta, 1/k, 1/r$. Let $p \ge Cn^{-1/m_k(H_1)}$.

Since (5.4.9) holds with high probability, to prove (5.4.10) holds it suffices to show that the probability $G_{n,p}^{(k)}$ contains a non- (H_1, \ldots, H_r) -Ramsey k-graph with at least $(\pi + \delta/2)p\binom{n}{k}$ edges tends to zero as n tends to infinity.

Suppose that $G_{n,p}^{(k)}$ does contain a non- (H_1, \ldots, H_r) -Ramsey k-graph I with at least $(\pi + \delta/2)p\binom{n}{k}$ edges. Then there exist pairwise edge-disjoint k-graphs I_1, \ldots, I_r such that $\bigcup_{i \in [r]} I_i = I$ and $(I_1, \ldots, I_r) \in \mathcal{I}_r(n; H_1, \ldots, H_r)$. Further, there is some $S = (S_1, \ldots, S_r) \in \mathcal{S}_r$ such that $S \subseteq (I_1, \ldots, I_r) \subseteq f(S)$. Thus, $G_{n,p}^{(k)}$ must contain (the edges of) $\bigcup_{i \in [r]} S_i$ as well as at least $(\pi + \delta/4)p\binom{n}{k}$ edges from $(\bigcup_{i \in [r]} f(S_i)) \setminus (\bigcup_{i \in [r]} S_i)$. (Note here we are using that $e(\bigcup_{i \in [r]} S_i) \leq \delta p\binom{n}{k}/4$, which holds by Theorem 5.34(ii) and since $0 < 1/C \ll 1/D, 1/k, \delta$.) Writing $s := e(\bigcup_{i \in [r]} S_i)$, the probability $G_{n,p}^{(k)}$ contains $\bigcup_{i \in [r]} S_i$ is p^s . Note that $e((\bigcup_{i \in [r]} f(S_i)) \setminus (\bigcup_{i \in [r]} S_i)) \leq (\pi + \delta/8)\binom{n}{k}$ by Theorem 5.34(iv)(b) and since n is sufficiently large. So by Proposition 1.2(i), the probability $G_{n,p}^{(k)}$ contains at least $(\pi + \delta/4)p\binom{n}{k}$ edges from $(\bigcup_{i \in [r]} f(S_i)) \setminus (\bigcup_{i \in [r]} S_i)$ is at most $\exp(-\delta^2 p\binom{n}{k}/256) \leq \exp(-\delta^2 pn^k/256k^k)$.

Write $N := n^{k-1/m_k(H_1)}$ and $\gamma := \delta^2/256k^k$. Given some integer $0 \leq s \leq DN$, there are at most $r^s\binom{\binom{n}{k}}{s}$ elements $(S_1, \ldots, S_r) \in S_r$ such that $e(\bigcup_{i \in [r]} S_i) = s$. Indeed, this follows since there are r^s ways to partition a set of size s into r classes. Thus, the probability that $G_{n,p}^{(k)}$ does contain a non- (H_1, \ldots, H_r) -Ramsey k-graph I with at least $(\pi + \delta/2)p\binom{n}{k}$ edges is at most

$$\begin{split} \sum_{s=0}^{DN} r^s \binom{\binom{n}{k}}{s} \cdot p^s \cdot e^{-\gamma n^k p} &\leq (DN+1)(rp)^{DN} \binom{\binom{n}{k}}{DN} e^{-\gamma n^k p} \\ &\leq (DN+1) \left(\frac{re^{k+1}pn^k}{DNk^k}\right)^{DN} e^{-\gamma n^k p} \end{split}$$

$$\leq (DN+1) \left(\frac{re^{k+1}C}{Dk^k}\right)^{DN} e^{-\gamma CN} \leq e^{\gamma CN/2} e^{-\gamma CN} = e^{-\gamma CN/2},$$

which tends to zero as \boldsymbol{n} tends to infinity. This completes the proof.

CHAPTER 6

THE MAKER-BREAKER RADO GAME ON A RANDOM SET OF INTEGERS

6.1 Introduction

Given a finite set X and a family of subsets of X, $\mathcal{F} \subseteq \mathcal{P}(X)$, we define the Maker-Breaker game on (X, \mathcal{F}) to be the game where Maker and Breaker take turns to select a previously unchosen element $x \in X$, and at the conclusion of the game, if Maker has claimed all of the elements of some $F \in \mathcal{F}$, then Maker wins. Otherwise Breaker has claimed at least one element x in every set $F \subseteq \mathcal{F}$, and Breaker wins. The set X is known as the *board*, and the family \mathcal{F} as the *winning sets*. If Maker has a strategy so that no matter how Breaker plays, Maker can always win, then we call the game *Maker's win*. If Maker has no such strategy, then since there is no draw scenario, the game is *Breaker's win*.

Maker-Breaker games first stemmed from a seminal paper by Erdős and Selfridge [35], where they proved their famous criterion which gives a general winning strategy for Breaker. Some well-known examples of Maker-Breaker games are where the board Xis the edge set of a complete graph K_n , and the winning sets \mathcal{F} are all sets of edges which correspond to a perfect matching; a Hamilton cycle; or a fixed subgraph H. All of these games turn out to be Maker's win if n is sufficiently large, therefore an adjustment to the game is required if we wish to make the problem of determining whose win the game is more interesting. This leads to the following two variations of Maker-Breaker board games, which have each received significant attention.

- Biased board games. Maker claims one element of the board per turn, whereas Breaker claims b elements per turn, for some fixed $b \in \mathbb{N}$. We call the game the (1:b) game on (X, \mathcal{F}) . Maker-Breaker games are 'bias-monotone' (see e.g. [52]). This means that there exists a threshold bias b_0 such that the (1:b) game on (X, \mathcal{F}) is Maker's win if and only if $b < b_0$.
- Random board games. For a fixed probability p = p(n) and game (X, F), let X_p be obtained by including each element x ∈ X with probability p independently of all other elements, and let F_p := {F ∈ F : x ∈ X_p for all x ∈ F}. We then consider the game on the random board (X_p, F_p), noting that it is a probability space of games. By the monotonicity of the game (X, F) being Maker's win, the existence of a threshold function follows from [19]. That is, there exists a threshold probability p₀ = p₀(n) such that

$$\lim_{n \to \infty} \mathbb{P}[\text{The game on } (X_p, \mathcal{F}_p) \text{ is Maker's win}] = \begin{cases} 1 & \text{if } p/p_0 \to \infty \text{ as } n \to \infty; \\ 0 & \text{if } p/p_0 \to 0 \text{ as } n \to \infty. \end{cases}$$

The interesting problem now is to determine the threshold bias and threshold probability for various Maker-Breaker games. For examples and further history of combinatorial board games, see e.g. [12, 52].

Random board games were first introduced by Stojaković and Szabó [113], who considered games played on a random subset of the edges of a complete graph. Note that this precisely corresponds to the edges of the Erdős-Rényi random graph $G_{n,p}$. Here, we focus on the game where Maker's aim is to obtain a solution to a system of linear equations within a random set of integers. To be precise, in our Maker-Breaker game, the board will be a random set of integers $[n]_p$ (recall this is obtained by including each element of [n] with probability p independently of all other elements). The winning sets are all sets of size k which correspond to a k-distinct solution (i.e. $x = (x_1, \ldots, x_k)$ has each x_i distinct) to a system of linear equations Ax = b, where A is a fixed integer-valued matrix of dimension $\ell \times k$ and b is a fixed integer-valued vector of dimension ℓ . We call such a game played on a set of integers X the (A, b)-game on X, or the \mathcal{L} -game on X(recalling the use of \mathcal{L} to represent the system of linear equations Ax = b used in previous chapters). The class of all \mathcal{L} -games are known as *Rado games* (introduced in [75]), due to the intimate link with Rado's partition theorem, which will be discussed shortly. As in the majority of the last chapter, we only care about k-distinct solutions (and therefore distinct \mathcal{L} -free sets).

Maker-Breaker games in this setting were first considered by Beck [13], who studied the van der Waerden game. Here, Maker's aim is to obtain a k-term arithmetic progression $a, a + r, \ldots, a + (k - 1)r$ for some $a, r \in \mathbb{N}$ and fixed $k \in \mathbb{N}$. Note that the set of k-term arithmetic progressions in [n] exactly coincides with the set of k-distinct solutions to Ax = 0 in [n] where A is the $(k - 2) \times k$ matrix given by

$$\begin{pmatrix} 1 & -2 & 1 & 0 & \cdots & 0 & 0 & 0 \\ 0 & 1 & -2 & 1 & \cdots & 0 & 0 & 0 \\ & & & \ddots & & & \\ 0 & 0 & 0 & 0 & \cdots & 1 & -2 & 1 \end{pmatrix}$$

Beck determined that the smallest $n \in \mathbb{N}$ such that the (A, 0)-game on [n] is Maker's win is $n = 2^{k(1+o(1))}$. Here we consider a generalisation of the van der Waerden game using the following definitions which we recall from the previous chapter. Let A be a fixed integer-valued matrix of dimension $\ell \times k$ and b a fixed integer-valued vector of dimension ℓ . We call \mathcal{L} (and the matrix A in the case where b = 0) *irredundant* if there exists a k-distinct solution to Ax = b in \mathbb{N} , and *partition regular* if for any finite colouring of \mathbb{N} , there is always a monochromatic solution (of any kind) to Ax = b.

A cornerstone result in the area of Ramsey theory for integers is Rado's theorem [88], which characterises all partition regular systems of linear equations \mathcal{L} . In [60], Hindman and Leader extended Rado's theorem to characterise all systems \mathcal{L} for which given any finite colouring of N, there is always a monochromatic k-distinct solution to Ax = b (in particular, if b = 0 then A must be irredundant and partition regular). Hindman and Leader's result implies that given such a system \mathcal{L} , if n is sufficiently large then however we 2-colour [n], there exists a monochromatic k-distinct solution to Ax = b. So in order for Breaker to win the \mathcal{L} -game on [n], he must himself obtain a k-distinct solution. But then by the classic strategy-stealing argument, Maker can claim this solution for herself. Thus this game is an (easy) win for Maker. Therefore it is interesting to consider biased and random versions of the \mathcal{L} -game on [n]. In a very recent paper of Kusch, Rué, Spiegel and Szabó [75], the biased version is considered. In this chapter, we consider the random version.

In fact (as in [75]), we consider a wider class of systems of linear equations \mathcal{L} . Recall the definition of (*), the matrix property we introduced in the previous chapter:

(*) Under Gaussian elimination A does not have any row which consists of precisely two non-zero rational entries.

In [75] the term *abundant* is used, which, recall from the previous chapter, is equivalent to (*) in the case of irredundant full rank matrices. Recall that in Section 5.3.1 it is proven that irredundant partition regular matrices are a strict subclass of irredundant matrices
which satisfy (*).

Recall the definition of m(A) (see (1.3.2)). The biased game result of [75] is the following.

Theorem 6.1 ([75], **Theorem 1.3 and Proposition 1.4**). Let A be a fixed integervalued matrix of dimension $\ell \times k$ and b a fixed integer-valued vector of dimension ℓ . Given the system of linear equations \mathcal{L} and the matrix A are both irredundant, then we have the following:

- (i) If A satisfies (*) then the threshold bias for the (A, b)-game on [n] is $\Theta(n^{1/m(A)})$;
- (ii) If A does not satisfy (*) then the (1:2) (A,b)-game on [n] is Breaker's win.

In this chapter we mainly focus on the case when A satisfies (*), though the case where A does not satisfy (*) does feature in our first result and also Section 6.4.

Our first result gives the threshold for the random \mathcal{L} -game whenever \mathcal{L} is a single linear equation.

Theorem 6.2. Let A be a fixed non-zero-integer-valued matrix of dimension $1 \times k$ and b a fixed integer (i.e. Ax = b corresponds to a single linear equation $a_1x_1 + \cdots + a_kx_k = b$ with the a_i non-zero integers).

- (i) If the system of linear equations *L* is irredundant and A is irredundant and satisfies
 (*), then the *L*-game on [n]_p has a threshold probability of Θ(n^{-k-2}/k-1);
- (ii) If the system of linear equations L is irredundant and A is irredundant and does not satisfy (*), then the L-game on [n]_p is Maker's win if p ≫ n^{-1/3} and Breaker's win if p ≪ n^{-1/3};
- (iii) If the system of linear equations \mathcal{L} is irredundant and A is not irredundant, then

- (a) the \mathcal{L} -game on $[n]_p$ is Breaker's win w.h.p. for any p = o(1) if the coefficients a_i are all positive or all negative;
- (b) the \mathcal{L} -game on $[n]_p$ is Maker's win if $p \gg n^{-1/3}$ and Breaker's win if $p \ll n^{-1/3}$ otherwise;
- (iv) If the system of linear equations L is not irredundant, then the L-game on [n] is
 (trivially) Breaker's win.

Note that most 'interesting' equations lies in the class of equations given by (i). In particular it includes several natural equations that have been extensively studied, e.g. x + y = z, x + y = z + t and x + y = 2z. In the \mathcal{L} -games corresponding to these equations, Breaker's aim is to restrict Maker's set to being a *sum-free set*, a *Sidon set* and a *progression-free set* respectively. The remaining classes of equations given by (ii)–(iv) are all in some sense 'trivial'; the proofs of these statements appear in Section 6.4.3.

In fact Theorem 6.2(i) will follow immediately from a much more general theorem. First we need some more definitions. We say that an $\ell \times k$ matrix A of full rank ℓ is strictly balanced if, for every $W \subseteq [k]$ for which $2 \leq |W| < k$, the inequality

$$\frac{|W|-1}{|W|-1+\operatorname{rank}(A_{\overline{W}})-\ell} < \frac{k-1}{k-1-\ell}$$

holds. In particular note that if A is strictly balanced then $m(A) = \frac{k-1}{k-1-\ell}$ (though the converse is not true). Given an irredundant matrix A which satisfies (*), we define the associated matrix B(A) to be a strictly balanced, irredundant matrix of full rank which satisfies (*), which is found by using elementary row operations on A then deleting some rows and columns, and satisfies m(B(A)) = m(A). The fact that such a matrix exists is not entirely obvious, and is essentially proven in [93]. We provide further details in Section 6.3. Also note that if A itself is strictly balanced then we simply have B(A) = A.

Recall $\mu(n, \mathcal{L}^d)$ denotes the size of the largest subset of [n] which does not contain a *k*-distinct solution to Ax = b. The main result of this chapter is the following.

Theorem 6.3. Let A be a fixed integer-valued matrix of dimension $\ell' \times k'$ and b a fixed integer-valued vector of dimension ℓ' . Given the system of linear equations \mathcal{L} is irredundant and A is irredundant and satisfies (*) we have the following:

(i) Let $\varepsilon > 0$. There exists a positive constant C such that if $p > Cn^{-1/m(A)}$, then for any $R \subseteq [n]_p$ satisfying $|R| \leq (1 - \frac{\mu(n, \mathcal{L}^d)}{n} - \varepsilon)np$, we have

$$\lim_{n\to\infty} \mathbb{P}\left(Maker \text{ wins the } \mathcal{L}\text{-}game \text{ on } [n]_p \setminus R\right) = 1.$$

(ii) Suppose the associated matrix B(A) is an $\ell \times k$ matrix of full rank ℓ , where ℓ divides k-1. There exists a positive constant c such that if $p < cn^{-1/m(A)}$ then

$$\lim_{n \to \infty} \mathbb{P}\left(Breaker \text{ wins the } \mathcal{L}\text{-}game \text{ on } [n]_p\right) = 1.$$

First note that it follows from a supersaturation result (Theorem 6.5) that for all pairs (A, b) as stated in Theorem 6.3, there exist $n_0 = n_0(A, b)$, $\delta = \delta(A, b) > 0$, such that for all integers $n \ge n_0$ we have $\mu(n, \mathcal{L}^d) \le (1-\delta)n$. Thus in particular Theorem 6.3(i) implies that there exists a positive constant C such that if $p > Cn^{-1/m(A)}$, then Maker wins the \mathcal{L} -game on $[n]_p$ w.h.p. Also, note that if A is a $1 \times k$ matrix with non-zero entries, then it is strictly balanced, and so B(A) = A. Thus A is a matrix which satisfies the hypothesis of Theorem 6.3(ii). Theorem 6.2(i) follows immediately from these two comments.

Another example of a class of pairs (A, b) for which Theorem 6.3 gives the threshold probability up to a constant factor are all irredundant systems of linear equations \mathcal{L} for which A is irredundant, has no columns consisting entirely of zeroes, satisfies (*) and is of dimension $2 \times k'$ for some odd k'. This follows since by construction either B(A) = Aor B(A) is a $1 \times k$ matrix for some k < k'. Either way, B(A) then satisfies the hypothesis of Theorem 6.3(ii).

For the Maker's win statement, the fact that we can delete a certain fraction of elements from $[n]_p$ and still have Maker's win w.h.p. means we have a resilience theorem. Note that in our result, the property is the game being Maker's win w.h.p., and the resilience is best possible in terms of the bound on the size of the set R: Indeed, since the largest subset of [n] with no k-distinct solutions to Ax = b has size $\mu(n, \mathcal{L}^d)$, w.h.p. $[n]_p$ contains a subset S of size $p(\mu(n, \mathcal{L}^d) - \varepsilon n)$ with no k-distinct solutions to Ax = b. Thus we can remove $(1 - \frac{\mu(n, \mathcal{L}^d)}{n} + \varepsilon)np$ elements from $[n]_p$ to obtain S (noting that a game on S is trivially Breaker's win).

It is very interesting to note the parallels between our theorem and the following random Rado theorems and the resilience theorem, which appeared in the previous chapter.

Theorem 5.8 ([93]). For all irredundant partition regular full rank matrices A and all positive integers $r \ge 2$, there exists a constant c > 0 such that

$$\lim_{n \to \infty} \mathbb{P}\left[[n]_p \text{ is } (\mathcal{L}^d, r) \text{-} Rado \right] = 0 \quad \text{ if } p < cn^{-1/m(A)}.$$

Theorem 5.9 ([42]). For all irredundant partition regular full rank matrices A and all positive integers r, there exists a constant C > 0 such that

$$\lim_{n \to \infty} \mathbb{P}\left[[n]_p \text{ is } (\mathcal{L}^d, r) \text{-} Rado \right] = 1 \quad \text{if } p > Cn^{-1/m(A)}.$$

Theorem 5.11. For all irredundant partition regular full rank matrices A, all positive integers r and all $\delta > 0$, there exists a constant C > 0 such that

$$\lim_{n \to \infty} \mathbb{P}\left[\frac{\operatorname{res}([n]_p, (\mathcal{L}^d, r) - Rado)}{|[n]_p|} = 1 - \frac{\mu(n, \mathcal{L}^d, r)}{n} \pm \delta\right] = 1 \quad \text{if } p > Cn^{-1/m(A)}$$

Theorem 5.9 implies that by again using strategy-stealing, we could obtain a proof for the non-resilient version of Theorem 6.3(i) for irredundant partition regular matrices. However our method as already noted achieves the best resilience possible, and further it extends to all irredundant matrices which satisfy (*) (even those for which there exists a 2-colouring of \mathbb{N} with no monochromatic k-distinct solutions to Ax = b). Our proof also gives an explicit strategy.

The proof of Theorem 6.3(i) closely follows the method of Theorem 16 in [86]. Here, Nenadov, Steger and Stojaković consider a similar problem: the *H*-game is where the board is the edges of a complete graph, and the winning sets are sets of edges which correspond to a copy of a fixed subgraph H. This game and its related Ramsey problems resemble the \mathcal{L} -game as follows: Set $d_2(H) := 0$ if e(H) = 0, $d_2(H) := 1/2$ if e(H) = 1, and $d_2(H) := (e(H) - 1)/(v(H) - 2)$ otherwise. Then define the 2-density of H to be $m_2(H) := \max_{H' \subseteq H} d_2(H')$. For most graphs H, the graph analogues of Theorems 5.8, 5.9 and 5.11 (the random Ramsey theorem and resilient subgraphs theorem, see Theorems 5.2 and 5.1) have a threshold of $\Theta(n^{-1/m_2(H)})$. Bednarska and Łuczak [14] showed that the threshold bias for the *H*-game is $\Theta(n^{1/m_2(H)})$. Thus both the *H*-games and \mathcal{L} -games (in most cases) have a threshold bias which is the inverse of the threshold for the random (respective) Ramsey/Rado theorem and the resilience theorems. Kusch, Rué, Spiegel and Szabó [75] in fact show that there is an intimate link between resilience and the threshold bias, which explains the parameters of m(A) and $m_2(H)$ appearing for both. They refer to this phenomenon as the *probabilistic Turán intuition* for biased Maker-Breaker games; see Section 6.4 of [75] for more details.

An analogous definition of strictly balanced exists for graphs. In [86], Nenadov, Steger and Stojaković show that the threshold probability for the random *H*-game is $\Theta(n^{-1/m_2(H)})$ when *H* is strictly balanced (Theorem 2 in [86]). However there are a class of graphs which have a threshold probability different to that of the random Ramsey/resilient subgraph theorem and the inverse of the threshold bias (Theorem 4 in [86]). Indeed, this is one of the main motivations for studying the random \mathcal{L} -game: For the proof of Theorem 6.3(ii), we build upon the method used by Rödl and Ruciński [93] to prove Theorem 5.8. Although our Breaker win statement is 'incomplete', its proof does seem to indicate that the threshold probability for the random \mathcal{L} -game (for any system of linear equations \mathcal{L} which is irredundant and A irredundant and satisfying (*)) should be the same as the random Rado threshold. That is, we hope that there is no need for the assumption that ℓ divides k - 1 in Theorem 6.3(ii). Also note that if we could prove our Breaker win statement for all strictly balanced matrices, then the full result would follow (see Proposition 6.8 and the paragraph following it). So interestingly in this sense, the random \mathcal{L} -game does not resemble the random H-game.

We prove the two parts of Theorem 6.3 in Sections 6.2 and 6.3 respectively, before finishing by proving Theorem 6.2(ii)–(iv) along with making some further remarks in Section 6.4.

6.2 Proof of Maker's win in Theorem 6.3

First we list a few results which are required for the proof. We will use the following simplification of Theorem 5.21. First recall $\mathcal{I}(n, \mathcal{L}^d)$ denote all sets from $\mathcal{P}([n])$ which contain no k-distinct solutions to Ax = b.

Theorem 6.4. Let $0 < \delta < 1$. Let A be a fixed integer-valued matrix of dimension $\ell \times k$ and b a fixed integer-valued vector of dimension ℓ . Suppose the system of linear equations \mathcal{L} is irredundant and A is irredundant and satisfies (*). Then there exists D > 0 such that the following holds. For all $n \in \mathbb{N}$, there is a collection $\mathcal{S} \subseteq \mathcal{P}([n])$ and a function $f : \mathcal{S} \to \mathcal{P}([n])$ such that:

(i) For all $I \in \mathcal{I}(n, \mathcal{L}^d)$, there exists $S \in \mathcal{S}$ such that $S \subseteq I \subseteq f(S)$.

Additionally, every $S \in \mathcal{S}$ satisfies

(*ii*) $|S| \leq Dn^{\frac{m(A)-1}{m(A)}}$;

(*iii*)
$$S \in \mathcal{I}(n, \mathcal{L}^d)$$
;

- (iv) f(S) contains at most $\delta n^{k-\ell}$ k-distinct solutions to Ax = b; and
- (v) $|f(S)| \leq \mu(n, \mathcal{L}^d) + \delta n.$

An upper bound on the size of the largest subset of [n] containing no k-distinct solutions to Ax = b is also required. The following is a consequence of Lemma 4.1 in [75] and Lemma 2.2.

Theorem 6.5. Let A be a fixed integer-valued matrix of dimension $\ell \times k$ and b a fixed integer-valued vector of dimension ℓ . Given the system of linear equations \mathcal{L} is irredundant and A is irredundant and satisfies (*), then there exist $n_0 \in \mathbb{N}$ and $\delta > 0$ such that for all integers $n \ge n_0$ we have $\mu(n, \mathcal{L}^d) \le (1 - \delta)n$.

Finally, we require the Erdős-Selfridge Criterion, commonly used to prove a Breaker strategy result, which we mentioned in the introduction. (Note that we do mean Breaker here; in our proof, we create an auxiliary game where the original Maker needs to play the role of Breaker!)

Theorem 6.6 ([35]). Let X be a set and let \mathcal{F} be a family of subsets of X. Then if Breaker has the first move in the game, and

$$\sum_{A \in \mathcal{F}} 2^{-|A|} < 1,$$

then Breaker has a winning strategy for the Maker-Breaker game (X, \mathcal{F}) .

Proof of Theorem 6.3(i). Apply Theorem 6.5 with parameters A, b to obtain $\varepsilon' > 0$ such that $\mu(n, \mathcal{L}^d) \leq (1 - \varepsilon')n$ for sufficiently large n. Let $\varepsilon > 0$ noting that without loss of generality we can assume $\varepsilon \ll \varepsilon'$. Suppose *n* is sufficiently large. Apply Theorem 6.4 with parameters $\varepsilon/4, A, b$ to obtain D > 0, a collection $S \subseteq \mathcal{P}([n])$ and a function fsatisfying Theorem 6.4(i)–(v). Fix $\delta \ll \varepsilon$ and choose *C* such that $0 \leq 1/C \ll 1/D, \delta, \varepsilon$. Let $p > Cn^{-1/m(A)}$. Note that m(A) > 1 (see Proposition 5.18(v)) and thus pn tends to infinity as *n* tends to infinity. Let *R* be as in the statement of the theorem and set $X := [n]_p \setminus R$.

Maker's aim is to claim a k-distinct solution to Ax = b within X, and Breaker's aim is to prevent this. If Maker loses, then her set $M \subseteq X$ does not contain a k-distinct solution to Ax = b. Hence $M \in \mathcal{I}(n, \mathcal{L}^d)$ and so there exists $S \in \mathcal{S}$ such that $S \subseteq M \subseteq f(S)$ and $S \subseteq X$. Given $S \in \mathcal{S}$ note that if Maker claims one element from $X \setminus f(S)$ then $M \not\subseteq f(S)$; hence consider the auxiliary game (X, \mathcal{F}) where

$$\mathcal{F} := \{ X \setminus f(S) : S \in \mathcal{S} \text{ and } S \subseteq X \}.$$

Maker can ensure that she wins the \mathcal{L} -game on X by picking at least one element from each set in \mathcal{F} , that is, she wins the auxiliary game as Breaker. We now make the following claim about the auxiliary game.

Claim 6.7. (i) For all $S \in S$ such that $S \subseteq X$, we have $|X \setminus f(S)| \ge \varepsilon np/2$ w.h.p.

(ii) We have $|\mathcal{F}| \leq 2^{\varepsilon np/4} w.h.p.$

Assuming the claim holds, it now easily follows that

$$\sum_{F \in \mathcal{F}} 2^{-|F|} \leqslant 2^{\varepsilon n p/4} \cdot 2^{-\varepsilon n p/2} < 1,$$

that is, the hypothesis of Theorem 6.6 holds for the game (X, \mathcal{F}) . Thus Maker wins the game as Breaker in the auxiliary game, and thus wins the original game (the \mathcal{L} -game on X). Since this happens w.h.p., it remains to prove the claim.

Proof of Claim 6.7. First we shall count $|\mathcal{F}|$. We wish to count the number of $S \in \mathcal{S}$ such that $S \subseteq X$. Recall that every $S \in \mathcal{S}$ satisfies $|S| \leq Dn^{1-1/m(A)} \leq Dpn/C$ and there are at most $\binom{n}{s}$ sets $S \in \mathcal{S}$ of size s. Thus we have

$$\mathbb{E}[|\mathcal{F}|] \leqslant \sum_{S \in \mathcal{S}} \mathbb{P}[S \subseteq [n]_p] \leqslant \sum_{S \in \mathcal{S}} p^{|S|} \leqslant \sum_{s=0}^{Dpn/C} \binom{n}{s} p^s \leqslant (Dpn/C+1) \binom{n}{Dpn/C} p^{Dpn/C}$$

$$(6.2.1)$$

$$\leqslant (Dpn/C+1) \left(\frac{Ce}{D}\right)^{Dpn/C} \leqslant e^{\delta n p} \leqslant 2^{\varepsilon n p/8},$$

where the last two inequalities follows by our choice of C and since $\delta \ll \varepsilon$ respectively. Thus by Proposition 1.1 we have

$$\mathbb{P}[|\mathcal{F}| \ge 2^{\varepsilon n p/4}] \le 2^{-\varepsilon n p/8},$$

which tends to zero as n tends to infinity, proving (ii).

Now for (i), observe that if we show that the probability that there exists $S \in \mathcal{S}$ such that $S \subseteq X$ and $|X \setminus f(S)| \leq \varepsilon np/2$ tends to zero as n tends to infinity, we will be done. First observe by Theorem 6.4 that for all $S \in \mathcal{S}$ we have $|f(S)| \leq \mu(n, \mathcal{L}^d) + \varepsilon n/4$ and so $|[n] \setminus f(S)| \geq n - \mu(n, \mathcal{L}^d) - \varepsilon n/4$. (In particular note $|[n] \setminus f(S)| = \Omega(n)$.) Let $\gamma := \varepsilon/(4 - 4\mu(n, \mathcal{L}^d)/n - \varepsilon)$ and $Y := [n]_p \setminus f(S)$ (noting $\gamma > 0$ since $\varepsilon \ll \varepsilon'$). By Proposition 1.2(ii) we have

$$\mathbb{P}\left[|([n] \setminus f(S)) \cap [n]_p| < \left(1 - \frac{\mu(n, \mathcal{L}^d)}{n} - \frac{\varepsilon}{2}\right)np\right] \leqslant \mathbb{P}\left[|Y| < (1 - \gamma)\mathbb{E}[|Y|]\right]$$
$$\leqslant 2e^{-\mathbb{E}[|Y|]\gamma^2/3} \leqslant e^{-2\delta np},$$

where the last inequality follows since $\delta \ll \varepsilon \ll \varepsilon'$. Note that since $|R| \leq (1 - \frac{\mu(n, \mathcal{L}^d)}{n} - \varepsilon)np$

and $X \setminus f(S) = Y \setminus R$ we have

$$\mathbb{P}[|X \setminus f(S)| < \varepsilon np/2] \leqslant e^{-2\delta np}, \tag{6.2.2}$$

for all $S \in \mathcal{S}$. Also since $S \subseteq f(S)$, the events $S \subseteq [n]_p$ and $|X \setminus f(S)|$ being small are independent. Thus

$$\mathbb{P}[\text{There exists } S \in \mathcal{S} \text{ such that } S \subseteq [n]_p \text{ and } |X \setminus f(S)| < \varepsilon np/2]$$

$$\leqslant \sum_{S \in \mathcal{S}} \mathbb{P}[S \subseteq [n]_p \text{ and } |X \setminus f(S)| < \varepsilon np/2]$$

$$\leqslant \sum_{S \in \mathcal{S}} \left(\mathbb{P}[S \subseteq [n]_p] \cdot \mathbb{P}[|X \setminus f(S)| < \varepsilon np/2]\right) \stackrel{(6.2.2)}{\leqslant} e^{-2\delta np} \sum_{S \in \mathcal{S}} \mathbb{P}[S \subseteq [n]_p]$$

$$\stackrel{(6.2.1)}{\leqslant} e^{-2\delta np} \cdot e^{\delta np} = e^{-\delta np},$$

which tends to zero as n tends to infinity, as required.

6.3 Proof of Breaker's win in Theorem 6.3

The proof will follow a similar tactic to that used by Rödl and Ruciński [93] for their proof of Theorem 5.8. Recall that the goal of Rödl and Ruciński was to show that, given an irredundant partition regular matrix A, an integer $r \ge 2$, and an upper bound on the probability p, then w.h.p. there exists an r-colouring of $[n]_p$ such that there are no monochromatic k-distinct solutions to Ax = 0. The proof consisted of three parts:

(P1) A reduction of the problem. It is shown that it suffices to prove the result for the associated matrix B(A). The problem is then rephrased to one about an associated hypergraph.

- (P2) A deterministic lemma. It is shown that if all r-colourings of $[n]_p$ contain a monochromatic k-distinct solution to Ax = 0, then the associated hypergraph must contain a certain kind of connected subhypergraph.
- (P3) A probabilistic lemma. It is shown that if $p < cn^{-1/m(A)}$, then w.h.p. the associated hypergraph does not contain the kind of subhypergraph given by the deterministic lemma.

Recall that our aim is to show that under the hypothesis of Theorem 6.3(ii), w.h.p. Breaker wins the \mathcal{L} -game on $[n]_p$. Our proof consists of the same three general parts, with appropriate amendments to the lemmas.

- (Q1) A reduction of the problem. As (P1) above.
- (Q2) **Two deterministic lemmas.** These together show that if Maker wins the \mathcal{L} -game on $[n]_p$, then the associated hypergraph must contain a certain kind of connected subhypergraph.
- (Q3) A probabilistic lemma. It is shown that if B(A) is an $\ell \times k$ matrix of full rank ℓ which satisfies ℓ divides k 1, and $p < cn^{-1/m(A)}$, then w.h.p. the associated hypergraph does not contain the kind of subhypergraph given by the deterministic lemmas.

We will of course make this more rigorous as we get to each part of the proof. We will shortly compare the differences between Rödl and Ruciński's proof and ours, but in order to do this it first makes sense to reduce each problem to one about an associated hypergraph.

(Q1) A reduction of the problem. First we show that in order for Breaker to win the (A, b)-game on any set of integers X, its suffices to show that Breaker wins the

(B, b')-game on X, for some matrix B := B(A) and vector b' := b'(A, b). For a vector $x = (x_1, \ldots, x_k)$ and a non-empty set $W \subseteq [k]$, let $x_W := (x_i)_{i \in W}$.

Proposition 6.8 ([75], Corollary 4.3 and Lemma 4.2). Let A be a fixed integer-valued matrix of dimension $\ell \times k$ and b a fixed integer-valued vector of dimension ℓ . Suppose the system of linear equations \mathcal{L} is irredundant and A is irredundant and satisfies (*). There exists a non-empty set $W \subseteq [k]$, a matrix B of full row-rank which is irredundant, with column-set W, satisfies (*), and is strictly balanced, and a vector b' for which the pair (B,b') is irredundant, such that if Ax = b, then $Bx_W = b'$.

Note that the homogeneous case for where b, b' are zero vectors is implicitly stated in [93]. We call the pair (B, b') above the associated pair of (A, b), and call B = B(A) the associated matrix of A. The consequence for us of Proposition 6.8 is that if Maker wins the (A, b)-game, then Maker also wins the (B, b')-game (since a solution to Ax = b always gives rise to a solution to Bx' = b'). Thus in order to prove Breaker wins the (A, b)-game, it suffices to prove that Breaker wins the (B, b')-game.

With any game (X, \mathcal{F}) there exists an associated hypergraph $H(X, \mathcal{F})$ with vertex set X and edge set \mathcal{F} . Write $H(X, B, b') := H(X, \mathcal{F})$ to represent the hypergraph where X is a set of integers, and \mathcal{F} is the set of all k-distinct solutions to Bx' = b' (assuming that B is an $\ell \times k$ matrix). Thus we may think of the game as one where Maker and Breaker take turns claiming a vertex of the k-uniform hypergraph $H([n]_p, B, b')$ and Maker's aim is to obtain an edge of $H([n]_p, B, b')$, and Breaker's aim is to prevent this. For the remainder of the proof we will assume that we have fixed A and b (and therefore B and b'), and set $H := H([n]_p, B, b')$. We will assume that B is an $\ell \times k$ matrix, and note that since B satisfies property (*), by Proposition 5.18(iv) we have $k \ge 3$.

Before we continue with the proof, we give a quick impression of the task at hand by considering the following example. Let \mathcal{L} be x + y = z and consider the hypergraph H drawn in Figure 6.1, where $a_i \in [n]_p$ for $i \in [15]$. Suppose Maker and Breaker are



Figure 6.1: Example H: different colours are used to represent each edge.

playing the \mathcal{L} -game on H, (i.e. Maker must claim all of the vertices of an edge of H). Maker has a strategy to win here. First she picks a_1 , then Breaker must pick from one of the two cycles; without loss of generality he picks a number from $\{a_9, \ldots, a_{15}\}$. Now Maker picks a_3 , enforcing Breaker to pick a_2 . Next, Maker picks a_5 enforcing Breaker to pick a_4 . Finally Maker picks a_7 ensuring that in her next turn she can win, since Breaker cannot select both of a_6 and a_8 , meaning that Maker will obtain the sum $\{a_1, a_8, a_7\}$ or $\{a_5, a_6, a_7\}$. Since Maker has a winning strategy here, this hypergraph H should be in the list of connected subhypergraphs in (Q2). We now give a sketch of why with high probability we can expect the associated hypergraph not to contain H.

We need to select numbers from $[n]_p$ to represent each of the a_i . We go through each edge in turn. For the first edge $\{a_1, a_2, a_3\}$ first pick a_1 and a_2 arbitrarily from $[n]_p$. Then note a_3 must form a solution to x + y = z with a_1 and a_2 so there are at most 3 choices for it (corresponding to which role of $x, y, z \, a_3$ plays). Thus there are $O(n^2p^3)$ choices for this first edge (since each number is included with probability p). For the next edge $\{a_3, a_4, a_5\}$, note that a_3 has already been chosen. One can arbitrarily pick a_4 , then there are at most 3 choices for a_5 , so there are $O(np^2)$ choices for the second edge. Continuing in this manner, there is a total of $O(n^7p^{15})$ choices for H. Now observe that for x + y = z, the threshold is $n^{-1/2}$, so the expected number of choices for H is o(1) as $n \to \infty$. Thus with high probability H does not appear in the associated hypergraph.

The overall aim of (Q2) and (Q3) is to ensure that any hypergraph for which Maker could win on does not appear in the associated hypergraph with high probability. As a comparison to the Rödl and Ruciński's proof of Theorem 5.8, note that if Breaker can win the \mathcal{L} -game on the associated hypergraph, then certainly there exists a 2-colouring of the vertices of the associated hypergraph such that there are no monochromatic edges. Thus finding such a colouring should be an easier task than finding a strategy for Breaker. Indeed, the deterministic lemma (P2) uses the fact that if a hypergraph has the property that however its vertices are 2-coloured there is always a monochromatic edge, then the hypergraph has chromatic number at least 3. Using this property it is then fairly easy to show that the hypergraph must contain one of a small list of certain connected subhypergraphs. Since the task of creating a Breaker strategy is harder, more work is required to do the analogous step $(\mathbf{Q2})$; in particular the list of certain subhypergraphs in $(\mathbf{P2})$ is a proper subset of the list of certain subhypergraphs for (Q2). To further demonstrate this point, note that the example in Figure 6.1 is a hypergraph for which there exists a 2-colouring such that there are no monochromatic edges (simply colour a_1, a_5, a_{12} red and the rest blue), while simultaneously being a win for Maker.

Hypergraph notation. We now introduce some notation which will be required for the deterministic and probabilistic lemmas. For the rest of Section 6.3, we assume that $k \ge 3$. For a k-uniform hypergraph H with edge set E := E(H) and vertex set V := V(H), let an *edge order* be an enumeration of the edges E. For a given edge order of E and edge $e \in E$, call a vertex $v \in e$ new in e if v did not appear in any edge which came before e in the edge order. Otherwise call v old in e. We call an edge e good if it has precisely one old vertex, bad if it has between two and k - 1 old vertices, and k-bad if it has k old vertices. Note that we always consider edges to be good, bad, or k-bad with respect to a given edge order; similarly whether a vertex is new or old in a given edge also depends on the given edge order. So throughout we will make it clear which edge order we are referring to. Note that given an edge order, a vertex will always be new in precisely one edge (and old in every other edge it appears in). For ease of notation we may sometimes identify a hypergraph with an edge order of its edges, e.g. if we have $P := e_0, \ldots, e_t$, then we consider the hypergraph P to have $E(P) := \{e_0, \ldots, e_t\}$ and $V(P) := \{x \in e, e \in E(P)\}.$

Let e_0, \ldots, e_t be an edge order. We call the edge order allowed if for all $i \in [t]$, e_i is good, bad or k-bad (that is, there is no edge e_i with $i \ge 1$ such that e_i is vertex-disjoint from all the edges e_0, \ldots, e_{i-1}). We call it valid if for all $i \in [t]$, e_i is good or bad. It is simple if for all $i \in [t]$, e_i is good. For a subset of edges e_{f_1}, \ldots, e_{f_u} of e_0, \ldots, e_t we do not assume $f_1 \le \ldots \le f_u$ unless otherwise stated. For two vertex-disjoint sets $X_1, X_2 \subseteq V(H)$, we define a minimal path from X_1 to X_2 in $\{e_0, \ldots, e_t\}$ to be a subset of edges e_{f_1}, \ldots, e_{f_u} of e_0, \ldots, e_t such that

- (i) we have $X_1 \cap e_{f_1} \neq \emptyset$, and $x \notin e_{f_a}$ for any $a \ge 2$ and $x \in X_1$;
- (ii) we have $X_2 \cap e_{f_u} \neq \emptyset$, and $x \notin e_{f_a}$ for any $a \leqslant u 1$ and $x \in X_2$;
- (iii) for all $i, j \in [u]$ with i < j we have $|e_{f_i} \cap e_{f_j}| \ge 1$ if i = j 1 and $|e_{f_i} \cap e_{f_j}| = 0$ otherwise.

We now give names to a variety of k-uniform hypergraphs which will appear in our deterministic and probabilistic lemmas. Suppose that e_{f_1}, \ldots, e_{f_u} for some $u \in \mathbb{N}$ is a valid edge order, where $|e_{f_i} \cap e_{f_{i+1}}| \ge 1$ for all $i \in [u-1]$. We call e_{f_1}, \ldots, e_{f_u} :

• An overlapping pair, if u = 2 and $2 \leq |e_{f_1} \cap e_{f_2}| \leq k - 1$;

• A loose cycle, if $u \ge 3$, and for all $i, j \in [u]$ with i < j we have

$$|e_{f_i} \cap e_{f_j}| = \begin{cases} 1 \text{ if } i = j - 1, \text{ or } i = 1 \text{ and } j = u; \\ 0 \text{ otherwise;} \end{cases}$$

• A loose path, if for all $i, j \in [u]$ with i < j we have

$$|e_{f_i} \cap e_{f_j}| = \begin{cases} 1 \text{ if } i = j - 1; \\ 0 \text{ otherwise;} \end{cases}$$

- A spoiled cycle, if $P_1 := e_{f_1}, e_{f_2}$ forms an overlapping pair, $P_2 := e_{f_3}, \ldots, e_{f_u}$ forms a loose path, and P_1 and P_2 are vertex-disjoint except for two vertices $x \neq y$, where we have $x = (e_{f_2} \setminus e_{f_1}) \cap (e_{f_3} \setminus e_{f_z})$ (where z = 4 if $u \ge 4$, and z = 1 otherwise) and $y = (e_{f_1} \setminus e_{f_2}) \cap (e_{f_u} \setminus e_{f_{u-1}})$;
- A double loose cycle, if for some $v \leq u 2$, $P_1 := e_{f_1}, \ldots, e_{f_v}$ forms a loose cycle, $P_2 := e_{f_{v+1}}, \ldots, e_{f_u}$ forms a loose path, and P_1 and P_2 are vertex-disjoint except for two vertices $x \neq y$, where we have $x = (e_{f_{v+1}} \setminus e_{f_{v+2}}) \cap e_{f_v}$ and $y = (e_{f_u} \setminus e_{f_{u-1}}) \cap e_{f_a}$ for some $a \in [v]$;
- A double overlapping pair, if u = 4, e_{f_1} , e_{f_2} and e_{f_3} , e_{f_4} each form overlapping pairs, which are vertex-disjoint except for two vertices $x \neq y$, where we have $x = (e_{f_1} \setminus e_{f_2}) \cap (e_{f_4} \setminus e_{f_3})$ and $y = (e_{f_2} \setminus e_{f_1}) \cap (e_{f_3} \setminus e_{f_4})$, and $|e_{f_3} \cap e_{f_4}| \leq k - 2$;
- An overlapping pair with handle/loose cycle with handle, if $e_{f_1}, \ldots, e_{f_{u-1}}$ forms an overlapping pair/loose cycle and e_{f_u} is bad in the edge order e_{f_1}, \ldots, e_{f_u} ;
- An overlapping pair/loose cycle to overlapping pair/loose cycle, if for some $w \leq v < u$, $P_1 := e_{f_1}, \ldots, e_{f_w}$ forms an overlapping pair or loose cycle, $P_2 := e_{f_{w+1}}, \ldots, e_{f_v}$

forms a loose path and $P_3 := e_{f_{v+1}}, \ldots, e_{f_u}$ forms an overlapping pair or loose cycle; moreover if w = v then $|V(P_1) \cap V(P_3)| = 1$; otherwise $|V(P_1) \cap V(P_2)| = 1$, $V(P_1) \cap V(P_3) = \emptyset$, $|V(P_2) \cap V(P_3)| = 1$, and additionally if $w \leq v - 2$, then $e_{f_{w+2}} \cap V(P_1) = \emptyset$ and $e_{f_{v-1}} \cap V(P_3) = \emptyset$.

Note that since we identify hypergraphs with one of their allowed edge orders, a hypergraph may fit the description of more than one of the above (e.g. a hypergraph could be both a spoiled cycle and an overlapping pair with handle, see Figure 6.2). We define a *bicycle* to be a hypergraph which is one of:

- a spoiled cycle;
- a double overlapping pair;
- a double loose cycle;
- an overlapping pair with handle;
- a loose cycle with handle;
- an overlapping pair to overlapping pair;
- an overlapping pair to loose cycle;
- a loose cycle to overlapping pair;
- a loose cycle to loose cycle.

Suppose that e_{f_1}, \ldots, e_{f_u} for some $u \in \mathbb{N}$ is an allowed edge order. We call e_{f_1}, \ldots, e_{f_u} :

• A Pasch configuration if k = 3, u = 4, there are six vertices within the four edges, and each of these appear in precisely two of the edges (one vertex for each of the six pairs of edges); see Figure 6.3;





An overlapping pair to loose cycle

Figure 6.2: Examples of our hypergraphs: different colours are used to represent each edge.

- A *k*-uniform loose *u*-star, if the edges are completely disjoint except for all intersecting in one 'central vertex';
- A (k, u/2, 2)-star, if u is even and the edges form two k-uniform loose (u/2)-stars S_1 and S_2 , and there is a bijection f between the edges of S_1 to those of S_2 such that e and f(e) share all their vertices except for the two central vertices, for each edge e in S_1 (so a (k, u, 2)-star has one more vertex than a k-uniform loose u-star, but has twice as many edges);
- A (k, u, a)-link, if there are k + a vertices within the u edges, and given any $i, j \in [u]$ with i < j we have $|e_{f_i} \cup e_{f_j}| = k + a$ (i.e. any pair of the edges contain all k + avertices between them).

Observe that a (k, u, a)-link with $u \ge 3$ must have $a \le \lfloor k/(u-1) \rfloor$, meanwhile a Pasch configuration is unique up to isomorphism.

Given S is any of our defined hypergraphs, we say that e_t completes S if the edge order e_0, \ldots, e_{t-1} does not contain a copy of S, whereas e_0, \ldots, e_t does.

Note that all bicycles have a valid edge order which contains at least two bad edges;



Figure 6.3: Examples of our hypergraphs: different colours are used to represent each edge.

we will in fact show that any hypergraph which has a valid edge order with at least two bad edges must contain a bicycle (see Claim 6.12). Meanwhile Pasch configurations, (k, u, 2)-stars with $u \ge 2$ and (k, u, a)-links with $u \ge 3$ and $a \le \lfloor k/(u-1) \rfloor$ all have the property that any allowed edge order contains at least one k-bad edge, and also precisely one bad edge; in particular these hypergraphs do not contain a bicycle. The roles of bicycles are crucial in our proof. The deterministic lemmas will imply that Breaker has a winning strategy for the game played on any component of H which does not contain a bicycle. The probabilistic lemma will show that w.h.p. H does not contain any bicycles.

(Q2) Two deterministic lemmas. Recall that we wish to show that if Maker wins the game on the associated hypergraph H, then H must contain a particular subhypergraph. (In particular, this subhypergraph will be a bicycle.) This section contains two deterministic lemmas, which together prove that the contrapositive statement holds; that is, if H does not contain a bicycle, then Breaker has a strategy to win the game on H.

Lemma 6.9. Let H' be a connected component of H and suppose H' does not contain a

bicycle. Then H' has an edge order e_0, \ldots, e_t with the property that there exists $a \in [0, t]$ such that e_i is good for all $i \in [a + 1, t]$, and also precisely one of the following holds:

(*i*)
$$a = 0$$
;

- (ii) $a \ge 2$ and e_0, \ldots, e_a forms a loose cycle;
- (iii) a = 1 and e_0, e_1 forms an overlapping pair;
- (iv) a = 3, k = 3 and e_0, \ldots, e_3 forms a Pasch configuration;
- (v) $a \ge 3$ is odd and e_0, \ldots, e_a forms a (k, (a+1)/2, 2)-star;
- (vi) $a \ge 2$, and e_0, \ldots, e_a forms a (k, (a+1), d)-link, where $d \le \lfloor k/a \rfloor$.

Lemma 6.10. Let H' be a component of H which is as described in Lemma 6.9. Breaker has a strategy for winning the Maker-Breaker game played on H'.

Note that by Breaker always choosing a vertex from the same component as Maker if he can, these results imply that if H does not contain a bicycle, then Breaker can win the game played on H, and therefore the (B, b')-game on $[n]_p$.

We now prove four claims; the proof of Lemma 6.9 will follow easily from the statements of these claims.

Claim 6.11. Suppose that $E_1 := e_0, \ldots, e_t$ is an allowed edge order of the edges of a connected hypergraph J, for which e_i for some $i \in [t]$ is the first bad or k-bad edge. Then we have the following:

- (i) Either there exists $j \in [0, i 1]$ such that e_j, e_i forms an overlapping pair, or e_i completes a loose cycle.
- (ii) Suppose that S is a connected subhypergraph of J with s edges, which contains the overlapping pair or loose cycle guaranteed by (i). Then there exists an allowed edge

order E_2 of E(J) which starts with the overlapping pair or loose cycle, followed by the rest of the edges of S, followed by any remaining edges of J.

(iii) If E_1 is valid and if S in (ii) is a loose cycle or overlapping pair, then it is possible to construct E_2 in (ii) so that additionally it is valid.

Proof. For (i), if there exists $j \in [0, i - 1]$ such that $|e_j \cap e_i| \ge 2$ then e_j, e_i forms an overlapping pair and we are done. So suppose:

(A1) For all $j \in [0, i-1]$ we have $|e_j \cap e_i| \leq 1$.

Also note that since e_i is the first bad or k-bad edge in E_1 , we have:

(A2) For all $j \in [1, i-1]$, e_j is good in E_1 .

Since e_i is bad or k-bad, it has $q \ge 2$ old vertices in the edge order E_1 . Label these as x_1, \ldots, x_q and consider a minimal path $P_1 := e_{f_1}, \ldots, e_{f_u}$ in $\{e_0, \ldots, e_{i-1}\}$ from $X_1 := \{x_1\}$ to $X_2 := \{x_2, \ldots, x_q\}$. By definition of P_1 , (A1) and (A2) we have

•
$$|e_{f_j} \cap e_i| = \begin{cases} 1 \text{ if } j = 1 \text{ or } j = u; \\ 0 \text{ otherwise;} \end{cases}$$

• $u \ge 2;$

• P_1 is a loose path.

It follows from these three facts that $e_{f_1}, \ldots, e_{f_u}, e_i$ forms a loose cycle. By (A2) e_0, \ldots, e_{i-1} clearly does not contain a loose cycle, and hence e_i completes a loose cycle in e_0, \ldots, e_i .

For (ii), such an allowed edge order E_2 exists since both S and J are connected; simply pick the overlapping pair or loose cycle first, then pick the remaining edges of S in any way so that each edge has non-empty intersection with the set of all previously chosen edges. Then pick the remaining edges of J in the same way. For (iii), suppose E_1 is valid and S is an overlapping pair or loose cycle. Then by the definition of E_1 , there are k new vertices in e_0 , k-1 new vertices in e_a for each $a \in [i-1]$ and at least one new vertex in e_i . Thus the hypergraph $J' := e_0, \ldots, e_i$ satisfies

(A3)
$$|V(J')| \ge k + (k-1)(i-1) + 1.$$

If there was an allowed edge order E_3 of E(J') which contained a k-bad edge, we would have $|V(J')| \leq k + (k-1)(i-1)$ since E_3 has one initial edge, at most i-1 good or bad edges, and at least one k-bad edge. However this violates (A3) and thus:

(A4) All allowed edge orders of E(J') are valid.

Now consider the edge order E_2 which starts with the loose cycle or overlapping pair, followed by the rest of the edges in $\{e_0, \ldots, e_i\}$ chosen so that each edge has non-empty intersection with the set of all previously chosen edges, then followed by e_{i+1}, \ldots, e_t (in this order). First note that for any e_a with $a \in [i + 1, t]$, the set of all previously chosen edges is $\{e_0, \ldots, e_{a-1}\}$ in both edge orders E_1 and E_2 . Thus if e_a is good or bad in E_1 , then it is also good or bad in E_2 . Finally note that by (A4), the first i + 1 edges in E_2 form a valid edge order of E(J'). We conclude that E_2 is the valid edge order of E(J)required.

Claim 6.12. A hypergraph J does not contain a bicycle if and only if any valid edge order of the edges of any connected subhypergraph J' of J has at most one bad edge.

Proof. First note that if J contains a bicycle, then by considering the edge order e_{f_1}, \ldots, e_{f_u} given in the definitions of each of the hypergraphs which the bicycle could be, we see immediately that J contains a connected subhypergraph which has a valid edge order with at least two bad edges.

Now we must show that if there exists a connected subhypergraph J' of J and a valid edge order of E(J') with at least two bad edges, then J contains a bicycle.

So let J' be such a hypergraph, and let $E_1 := e_0, \ldots, e_t$ be the valid edge order of E(J') with at least two bad edges. By using all three parts of Claim 6.11, we may assume without loss of generality that there exists $i \in [t]$ such that we have precisely one of the following:

- (B1) i = 1 and e_0, e_1 forms an overlapping pair;
- (B2) e_0, \ldots, e_i forms a loose cycle (with edges ordered cyclically).

Let $P_1 := e_0, \ldots, e_i$ and let $j \in [i+1, t]$ be such that e_j is the next bad edge in E_1 after e_i . We have precisely one of the following:

- (B3) $2 \leq |e_j \cap V(P_1)| \leq k-1;$
- (B4) There exists $a \in [i + 1, j 1]$ such that $|e_j \cap e_a| \ge 2$ and $x = e_j \cap V(P_1)$ and $y = e_a \cap V(P_1)$ are distinct vertices;
- (B5) There exists $a \in [i+1, j-1]$ such that $|e_j \cap e_a| \ge 2$ and $|(e_a \cup e_j) \cap V(P_1)| = 1$;
- (B6) There exists $a \in [i+1, j-1]$ such that $|e_j \cap e_a| \ge 2$ and $(e_a \cup e_j) \cap V(P_1) = \emptyset$;
- (B7) For all $a \in [0, j-1]$ we have $|e_j \cap e_a| \leq 1$, and $|e_j \cap V(P_1)| = 1$;
- (B8) For all $a \in [0, j-1]$ we have $|e_j \cap e_a| \leq 1$, and $e_j \cap V(P_1) = \emptyset$.

For each case, it suffices to find a subhypergraph of J' which is a bicycle. Throughout we will make use of the following fact:

(B9) For all $a \in [j-1] \setminus \{i\}$, e_a is good in E_1 .

Case 1: (B3) holds. Clearly e_0, \ldots, e_i, e_j forms an overlapping pair/loose cycle with handle.

Case 2: (B4) holds. We have precisely one of the following:

- There exists $d \in [0, i]$ such that $x, y \in e_d$; then e_j, e_a, e_d forms an overlapping pair with handle.
- There does not exist d ∈ [0, i] such that x, y ∈ e_d; without loss of generality suppose that x ∈ e₀ \ e₁. If P₁ is an overlapping pair, then e₀, e₁, e_a, e_j forms a double overlapping pair. (Note that |e_a ∩ e_j| ≤ k − 2 since otherwise e_j would be k-bad in the edge order E₁.) If P₁ is a loose cycle, then let d ∈ [i] be the smallest integer such that y ∈ e_d. If d = i, then e_a, e_j, e₀, e_i forms a spoiled cycle; otherwise e_a, e_j, e₀, ..., e_d forms a spoiled cycle.

Case 3: (B5) holds. Let $P_2 := e_a, e_j$ and note that $|V(P_2) \cap V(P_1)| = 1$. It follows that $e_0, \ldots, e_i, e_a, e_j$ forms an overlapping pair/loose cycle to overlapping pair.

Case 4: (B6) holds. Again let $P_2 := e_a, e_j$ and note that we have $V(P_2) \cap V(P_1) = \emptyset$. So consider a minimal path $P_3 := e_{f_1}, \ldots, e_{f_u}$ in $\{e_{i+1}, \ldots, e_{j-1}\} \setminus e_a$ from $X_1 := V(P_1)$ to $X_2 := V(P_2)$. By (B9) P_3 is a loose path, moreover by definition of P_3 , we have $|V(P_1) \cap V(P_3)| = 1$ and $|V(P_2) \cap V(P_3)| = 1$. Additionally if $u \ge 2$, then $e_{f_2} \cap V(P_1) = \emptyset$ and $e_{f_{u-1}} \cap V(P_2) = \emptyset$. Thus P_1 , P_3 and P_2 together form an overlapping pair/loose cycle to overlapping pair.

Case 5: (B7) holds. Since e_j has at least one old vertex which is not in e_0, \ldots, e_i , we may consider a minimal path $P_2 := e_{f_1}, \ldots, e_{f_u}$ in $\{e_{i+1}, \ldots, e_{j-1}\}$ from $X_1 := V(P_1)$ to $X_2 := e_j \setminus V(P_1)$. First note by (B9) that P_2 is a loose path. Now we have precisely one of the following:

- We have (e_{f1} ∩ e_j ∩ V(P₁)) ≠ Ø; then P₃ := e_{f1},..., e_{fu}, e_j forms a loose cycle. Now since |V(P₁) ∩ V(P₃)| = 1, P₁ and P₃ together form an overlapping pair/loose cycle to loose cycle.
- There exists $d \in [0, i]$ such that $x = e_j \cap e_d$ and $y = e_{f_1} \cap e_d$ are distinct vertices; then $P_3 := e_d, e_{f_1}, \dots, e_{f_u}, e_j$ forms a loose cycle. If P_1 is an overlapping pair, then let

 $d' \in \{0, 1\}$ be such that $d' \neq d$. Then P_3 together with $e_{d'}$ forms a loose cycle with handle. If P_1 is a loose cycle, then let P_4 be the loose path $e_{f_1}, \ldots, e_{f_u}, e_j$ and define z := j if u = 1 and $z := f_2$ otherwise. Observe that P_1 and P_4 are vertex-disjoint except for $x = (e_j \setminus e_{f_u}) \cap e_d$ and $y = (e_{f_1} \setminus e_z) \cap e_d$, and thus together form a double loose cycle.

• We have $x = e_j \cap V(P_1)$ and $y = e_{f_1} \cap V(P_1)$ are not together in any edge of P_1 ; without loss of generality suppose that $x \in e_0 \setminus e_1$. If P_1 is an overlapping pair, then $e_0, e_1, e_{f_1}, \ldots, e_{f_u}, e_j$ forms a spoiled cycle. If P_1 is a loose cycle, then define P_4, z as in the previous bullet point. Then observe that P_1 and P_4 are vertex-disjoint except for $x = (e_j \setminus e_{f_u}) \cap e_0$ and $y = (e_{f_1} \setminus e_z) \cap e_d$ for some $d \in [i]$, and thus as before, form a double loose cycle.

Case 6: (B8) holds. Label the $q \ge 2$ old vertices of e_j as x_1, \ldots, x_q and consider a minimal path $P_2 := e_{f_1}, \ldots, e_{f_u}$ in $\{e_{i+1}, \ldots, e_{j-1}\}$ from $X_1 := V(P_1)$ to $X_2 := \{x_1, \ldots, x_q\}$. Let $x_a := e_{f_u} \cap e_j$ and consider a minimal path $P_3 := e_{f_{u+1}}, \ldots, e_{f_v}$ in $\{e_{i+1}, \ldots, e_{j-1}\} \setminus E(P_2)$ from $X_3 := X_2 \setminus x_a$ to $X_4 := V(P_1) \cup V(P_2)$. By (B9) both P_2 and P_3 are loose paths. Now we have precisely one of the following:

We have e_{fv} ∩ V(P₁) = Ø; let d ∈ [u] be the largest integer such that e_{fd} ∩ e_{fv} ≠
Ø. Then P₄ := e_{fd},..., e_{fu}, e_j, e_{fu+1},..., e_{fv} forms a loose cycle. If d = 1, then since |V(P₁) ∩ V(P₄)| = 1, we have that P₁ and P₄ together form an overlapping pair/loose cycle to loose cycle. Otherwise let P₅ := e_{f1},..., e_{fd-1}. Then we have |V(P₁) ∩ V(P₅)| = 1, |V(P₄) ∩ V(P₅)| = 1 and V(P₁) ∩ V(P₄) = Ø. Additionally, if d ≥ 3, then we see that e_{f2} ∩ V(P₁) = Ø and e_{fd-2} ∩ V(P₄) = Ø. Thus P₁, P₄, P₅ form an overlapping pair/loose cycle to loose cycle to loose cycle to loose cycle.

• We have $e_{f_v} \cap V(P_1) \neq \emptyset$; the properties of the hypergraph

$$e_0, \ldots, e_i, e_{f_1}, \ldots, e_{f_u}, e_j, e_{f_{u+1}}, \ldots, e_{f_v}$$

are identical to that the hypergraph $e_0, \ldots, e_i, e_{f_1}, \ldots, e_{f_u}, e_j$ found in Case 5 (up to the labelling of the edges), so a similar case study yields a bicycle.

For the remainer of the proof, we shall call a valid edge order which contains at least two bad edges a *bad edge order*. If a hypergraph J does not contain a bicycle, then by Claim 6.12, the existence of a bad edge order of E(J') where J' is a subhypergraph of J is a contradiction. In the claims which follow we will always assume that J does not contain a bicycle, and hence whenever some assumed condition of a case within a case analysis leads to the discovery of a bad edge order, we can immediately stop and move onto the next case.

Claim 6.13. Let S be a hypergraph with s edges, which is an overlapping pair, a loose cycle, a (k, s/2, 2)-star with $s \ge 4$, a (k, s, a)-link with $s \ge 3$ and $a \in [\lfloor k/(s-1) \rfloor]$, or a Pasch configuration. Suppose J is a connected hypergraph which does not contain a bicycle, and does contain S. Then there exists an allowed edge order $E_1 := e_0, \ldots, e_t$ of E(J) such that

- (i) e_0, \ldots, e_i forms an overlapping pair or loose cycle, e_0, \ldots, e_{s-1} forms S, and every edge e_j for $j \in [s, t]$ is either good or k-bad;
- (ii) For all $j \in [s+1,t]$, if e_j is k-bad, then either e_{j-1} is also k-bad, or there exists a vertex $x \in e_j$ which is new in e_{j-1} .

Proof. For (i), by Claim 6.11(ii) we can assume that the edge order starts with the overlapping pair or loose cycle, followed by the rest of the edges of S, and that e_i is bad.

If there exists another bad edge e_j for some $j \in [s, t]$, then the edge order E_2 , found by deleting from e_0, \ldots, e_j all k-bad edges, is bad. Thus for all $j \in [s, t]$, e_j must either be good or k-bad.

For (ii) suppose that $E_1 = e_0, \ldots, e_t$ does not satisfy the property stated in (ii). Then we have the following:

(C4) There exists $j \in [s+1,t]$ such that e_{j-1} is good and e_j is k-bad, and all vertices in e_j appeared in the edge order before e_{j-1} .

Now consider the edge order $E_2 := e_0, \ldots, e_{j-2}, e_j, e_{j-1}, e_{j+1}, \ldots, e_t$. In this order, e_j is still k-bad and e_{j-1} is still good; moreover E_2 still starts with the overlapping pair or loose cycle. Hence by continuously performing swaps whenever such a pair e_{j-1} and e_j exists (satisfying (C4)), we eventually reach an edge order E_p where no such pair exists. Thus in the final edge order E_p the property stated in (ii) holds.

Claim 6.14. Let S be a hypergraph with s edges, which is an overlapping pair, a loose cycle, a (k, s/2, 2)-star with $s \ge 4$, a (k, s, a)-link with $s \ge 3$ and $a \in [\lfloor k/(s-1) \rfloor]$, or a Pasch configuration. Suppose J is a connected hypergraph which does not contain a bicycle, and does contain S. Finally suppose $E_1 := e_0, \ldots, e_t$ is the allowed edge order of E(J) guaranteed by Claim 6.13, which starts with the edges of S, in particular with the overlapping pair or loose cycle $P_1 := e_0, \ldots, e_i$. Suppose that E_1 contains at least one k-bad edge amongst the edges e_s, \ldots, e_t . Then we have precisely one of the following:

- (i) We have that e_0, e_1, e_{j-1}, e_j forms a (k, 2, 2)-star for some $j \in [s + 1, t]$. Moreover either S is an overlapping pair, or S is a (k, s/2, 2)-star with $s \ge 4$ and $e_0, \ldots, e_{s-1}, e_{j-1}, e_j$ forms a (k, s/2 + 1, 2)-star.
- (ii) We have that e_0, e_1, e_s forms a (k, 3, a)-link, where $a = |e_0 \setminus e_1|$. Moreover either S is an overlapping pair, or S is a (k, s, a)-link with $s \ge 3$ and e_0, \ldots, e_s forms a (k, s + 1, a)-link.

(iii) We have that S is a loose cycle with three edges, k = 3, and e_0, e_1, e_2, e_3 forms a Pasch configuration.

Proof. Suppose that e_j is the first k-bad edge amongst the edges e_s, \ldots, e_t (so if j > s then e_s, \ldots, e_{j-1} are good). Let $e_j := \{x_1, \ldots, x_k\}$ and let e_{f_1}, \ldots, e_{f_k} be the respective edges in which each x_i is new in E_1 , noting that without loss of generality we have $f_1 \leq \cdots \leq f_k$.

We have precisely one of the following:

- (D1) We have $j \in [s+1, t];$
- (D2) We have j = s and $P_1 = e_0, e_1$ is an overlapping pair;
- (D3) We have j = s and $P_1 = e_0, \ldots, e_i$ is a loose cycle.

We will go through each of these cases in turn and show that Claims 6.14(i), (ii) and (iii) hold respectively.

Case 1: (D1) holds. Without loss of generality, we have precisely one of the following:

- (D4) We have $2 \leq |e_j \cap V(P_1)| \leq k-1$;
- (D5) We have $|e_j \cap V(P_1)| \leq 1$ and for all $a \in [i+1, j-1]$ we have $|e_j \cap e_a| \leq 1$;
- (D6) There exists $a \in [i+1, j-1]$ such that $|e_j \cap e_a| \ge 2$ and $|(e_j \cup e_a) \cap V(P_1)| \le 1$;
- (D7) There exists $a \in [i + 1, j 1]$ such that $|e_j \cap e_a| \ge 2$ and $x = e_j \cap V(P_1)$ and $y = e_a \cap V(P_1)$ are distinct vertices.

Case 1a: (D4) holds. We have that e_0, \ldots, e_i, e_a forms an overlapping pair/loose cycle with handle, a contradiction to J not containing a bicycle.

Case 1b: (D5) holds. Since we have $|V(P_1) \cap e_j| \leq 1$ and for all $a \in [i+1, j-1]$ we have $|e_j \cap e_a| \leq 1$, we obtain

- (D8) $f_2 > i;$
- (D9) $f_1 < \cdots < f_k$.

Then consider the edge orders $E_2 := e_0, \ldots, e_{f_2}, e_j$ and E_3 , which is formed by deleting from E_2 each edge which is k-bad. We will show that E_3 is a bad edge order. First note that clearly E_3 is a valid edge order since all k-bad edges were deleted. By (D8), E_2 and hence also E_3 both start with the edges of P_1 , so have at least one bad edge. Further e_j is bad in E_2 since precisely two of the vertices in e_j are old in E_2 , namely x_1 and x_2 , which appear in e_{f_1} and e_{f_2} respectively. Since e_{f_1} and e_{f_2} both contain a new vertex, they are not k-bad, and so are both contained in E_3 . Thus e_j is also bad in E_3 , and so E_3 is indeed a bad edge order.

Case 1c: (D6) holds. Let $P_2 := e_a, e_j$ and note that if $|V(P_1) \cap V(P_2)| = 1$, then $V(P_1)$ and $V(P_2)$ together form an overlapping pair/loose cycle to overlapping pair, a contradiction to J not containing a bicycle. Otherwise we have $V(P_1) \cap V(P_2) = \emptyset$, so consider a minimal path $P_3 := e_{g_1}, \ldots, e_{g_u}$ in $\{e_d : d \in [i+1, j-1], d \neq a, e_d \text{ is good in } E_1\}$ from $X_1 := V(P_1)$ to $X_2 := V(P_2)$. By the choice of where the edges in P_3 are selected from, P_3 is a loose path. Additionally we have $|V(P_1) \cap V(P_3)| = 1$, $|V(P_2) \cap V(P_3)| = 1$, and if $u \ge 2$, then $V(P_1) \cap e_{g_2} = \emptyset$ and $V(P_2) \cap e_{g_{u-1}} = \emptyset$. Thus P_1, P_2, P_3 form an overlapping pair/loose cycle to overlapping pair, a contradiction to J not containing a bicycle.

Case 1d: (D7) holds. First suppose that P_1 is a loose cycle and without loss of generality that $x \in e_0 \setminus e_i$. If $y \in e_0$, then let $E_2 := e_a, e_j, e_0$. If $y \in e_i \setminus e_0$, then let $E_2 := e_a, e_j, e_0, e_i$. Otherwise let $E_2 := e_a, e_j, e_0, \ldots, e_i$. For each case e_j and the last edge are both bad edges in E_2 . Further it is easy to see that in each case, E_2 is valid, and thus E_2 is bad.

Now suppose that P_1 is an overlapping pair. If x and y are both in e_d for d = 0or d = 1, then e_j, e_a, e_d is a bad edge order. So suppose without loss of generality that $x \in e_1 \setminus e_0$, and $y \in e_0 \setminus e_1$. If $|e_j \cap e_a| \leq k-2$, then e_0, e_1, e_j, e_a forms a double overlapping pair. Similarly if $|e_0 \cap e_1| \leq k-2$, then e_j, e_a, e_0, e_1 forms a double overlapping pair. Both would contradict J not containing a bicycle, and thus we have $|e_j \cap e_a| = |e_0 \cap e_1| = k-1$. Hence we also have $x = e_1 \setminus e_0 = e_j \setminus e_a$ and $y = e_0 \setminus e_1 = e_a \setminus e_j$ and so in particular, we have that e_0, e_1, e_j, e_a forms a (k, 2, 2)-star.

The conclusion of our case analysis is that there exists $a \in [2, j - 1]$ such that e_0, e_1, e_a, e_j forms a (k, 2, 2)-star and that P_1 is an overlapping pair where $x = e_1 \setminus e_0$ and $y = e_0 \setminus e_1$.

Since E_1 is the edge order obtained from Claim 6.13, we have e_j comes immediately after the edge for which the last of the vertices of e_j are new, or following another k-bad edge. Therefore, since e_{j-1} is not k-bad and $e_j \subseteq (e_0 \cup e_1 \cup e_a)$, we conclude that a = j-1.

Note that S cannot be a loose cycle or Pasch configuration, since P_1 is an overlapping pair. If S is a (k, s, d)-link for some $s \ge 3$, then we have $e_2 \cap (e_{j-1} \cup e_j) = \{x, y\}$ and so the edge order e_{j-1}, e_j, e_2 is bad. If S is an overlapping pair, then there is nothing further to prove. Finally if S is a (k, s/2, 2)-star, then clearly $e_0, \ldots, e_{s-1}, e_{j-1}, e_j$ forms a (k, s/2 + 1, 2)-star (with central vertices x and y).

Case 2: (D2) holds. First note that S cannot be a loose cycle or Pasch configuration, since P_1 is an overlapping pair. We have precisely one of the following:

- (D10) We have $|e_s \cap (e_0 \cup e_1)| = k;$
- (D11) We have $2 \leq |e_s \cap (e_0 \cup e_1)| \leq k 1;$
- (D12) We have $|e_s \cap (e_0 \cup e_1)| \leq 1$.

Case 2a: (D10) holds. First suppose that $|e_s \cap e_1| \leq 1$. Since e_0, e_1 is an overlapping

pair, we have

$$k = |e_s| = |e_s \cap e_1| + |e_s \cap (e_0 \setminus e_1)| \leqslant k - 1,$$

a contradiction, and hence we must have $|e_s \cap e_1| \ge 2$. Similarly $|e_s \cap e_0| \ge 2$. Thus any permutation of the edges e_0, e_1, e_s must have that the second edge is bad and the third edge is bad or k-bad. In order to not obtain a bad edge oder, we must have that the third edge is k-bad in all of these permutations. Thus by definition e_0, e_1, e_s forms a $(k, 3, |e_0 \setminus e_1|)$ -link. If S is an overlapping pair, then there is nothing further to prove. If S is a (k, s/2, 2)-star for some $s \ge 4$, then without loss of generality we have $x = e_0 \setminus e_1 = e_2 \setminus e_3$ and $y = e_1 \setminus e_0 = e_3 \setminus e_2$. But then since $x, y \in e_s$, the edge order e_2, e_3, e_s is bad. Finally suppose that S is a (k, s, a)-link. Then for all $d, d' \in [0, s - 1]$ with d < d', we have that $e_d, e_{d'}$ forms an overlapping pair, and $|e_d \setminus e'_d| = |e_0 \setminus e_1| = a$. By repeating the argument above for the permutations of e_0, e_1, e_s , we see that any permutation of $e_d, e_{d'}, e_s$ must have that the second edge is bad and the third is k-bad. Thus e_0, \ldots, e_s forms a $(k, s + 1, |e_0 \setminus e_1|)$ -link.

Case 2b: (D11) holds. We have that e_0, e_1, e_s forms an overlapping pair with handle, a contradiction to J not containing a bicycle.

Case 2c: (D12) holds. First note that since e_s contains vertices outside of $e_0 \cup e_1$, we have that S cannot be an overlapping pair or a $(k, s, |e_0 \setminus e_1|)$ -link, and thus S must be a (k, s/2, 2)-star. If there exists i, j such that $|e_i \cap e_j| = k - 1$ and $|e_s \cap (e_i \cup e_j)| \ge 2$, then repeat the argument from Case 2a or 2b with e_0 and e_1 replaced by e_i and e_j . For the remaining case we have that for all i, j such that $|e_i \cap e_j| = k - 1$, we have $|e_s \cap (e_i \cup e_j)| \le 1$ and in particular, the central vertices of the two stars are not in e_s . Now without loss of generality, suppose e_2 and e_3 is an overlapping pair, $|e_s \cap (e_0 \cap e_1)| = 1$ and $|e_s \cap (e_2 \cap e_3)| = 1$. Then the edge order e_0, e_1, e_2, e_s is a bad edge order.

Case 3: (D3) holds. Since S contains a loose cycle, S cannot be an overlapping

pair, (k, s/2, 2)-star or a (k, s, a)-link, and hence S is a loose cycle or Pasch configuration. If S is the latter, then we have j = 4, and by the symmetry of the Pasch configuration, we have without loss of generality that $|e_0 \cap e_4| = 2$ and $|e_1 \cap (e_0 \cup e_4)| = 2$. Hence e_0, e_4, e_1 is a bad edge order. Thus S must be a loose cycle and j = i + 1. Without loss of generality we have precisely one of the following:

• $|e_0 \cap e_{i+1}| \ge 2$ and $(e_i \setminus e_0) \cap e_{i+1} = \emptyset$; let $E_2 := e_{i+1}, e_0, \dots, e_i$. For all $a \in [i]$, we have $|(e_a \setminus e_{a-1}) \cap e_{i+1}| \le k-2$, and thus

$$|e_a \cap (e_{a-1} \cup e_{i+1})| \leq |e_{a-1} \cap e_a| + |(e_a \setminus e_{a-1}) \cap e_{i+1}| \leq k - 1.$$

Thus E_2 is a valid edge order. Further e_0 is bad in E_2 . Since e_i completes the cycle e_0, \ldots, e_i , it is also bad in E_2 , and thus E_2 is a bad edge order.

• $|e_0 \cap e_{i+1}| \ge 2$ and $(e_i \setminus e_0) \cap e_{i+1} \neq \emptyset$; let $E_2 := e_{i+1}, e_0, e_i$. Here we have

$$2 \leqslant |e_i \cap (e_{i+1} \cup e_0)| \leqslant k - 1$$

and so E_2 is a bad edge order.

- For all $a \in [0, i]$, we have $|e_{i+1} \cap e_a| \leq 1$ and $(e_i \setminus (e_0 \cup e_{i-1})) \cap e_{i+1} = \emptyset$; We have that (D9) holds, and thus $E_2 := e_0, \ldots, e_{f_2}, e_{i+1}, e_{f_2+1}, \ldots, e_i$ is a bad edge order.
- For all $a \in [0, i]$, we have $|e_{i+1} \cap e_a| \leq 1$, $|(e_a \setminus \{e_d : d \in [0, i], d \neq a\}) \cap e_{i+1}| = 1$ and $k \geq 4$; Then $E_2 := e_0, e_1, e_{i+1}, e_2$ is a bad edge order.
- For all $a \in [0, i]$, we have $|e_{i+1} \cap e_a| \leq 1$, $|(e_a \setminus \{e_d : d \in [0, i], d \neq a\}) \cap e_{i+1}| = 1$ and k = 3. Then we have i = 2 and e_0, e_1, e_2, e_3 forms a Pasch configuration.

Only the last case does not produce a bad edge order, and thus we have that e_0, e_1, e_2, e_3 forms a Pasch configuration, as required.

We are now ready to prove our two deterministic lemmas.

Proof of Lemma 6.9. If there exists a simple edge order of E(H'), then we have (i), so suppose that any edge order of E(H') contains at least one bad or k-bad edge. Now by Claim 6.11 we may assume that $E_1 := e_0, \ldots, e_t$ is an allowed edge order of E(H') which starts with an overlapping pair or loose cycle $P_1 := e_0, \ldots, e_i$. Then using Claim 6.13 applied with $S := P_1$, we have that every edge e_{i+1}, \ldots, e_t is either k-bad or good. If all of these edges are good, then we have (ii) or (iii), so are done. So assume that there is at least one k-bad edge. Then using Claim 6.14 applied with $S := P_1$, we have that H' contains a (k, 2, 2)-star, a $(k, 3, |e_0 \setminus e_1|)$ -link or a Pasch configuration. We can now repeatedly use Claims 6.13 and 6.14 as follows:

- (a) Let S be the (k, p, 2)-star, a (k, p + 1, |e₀ \ e₁|)-link or a Pasch configuration found previously (where p ≥ 2). By Claim 6.13, there exists an allowed edge order of E(H') which starts with all of the edges of S. If there are no further k-bad edges, then we have (iv), (v) or (vi). If there are, then move to step (b).
- (b) By Claim 6.14, either S was a (k, p, 2)-star and H' contains a (k, p + 1, 2)-star, or S was a (k, p + 1, |e₀ \ e₁|)-link and H' contains a (k, p + 2, |e₀ \ e₁|)-link. Now return to step (a).

Since H' is a finite hypergraph, this process must eventually stop, and hence we have (iv), (v) or (vi).

For the proof of Lemma 6.10, we simply find an explicit strategy for Breaker to win the game played on H'.

Proof of Lemma 6.10. By Lemma 6.9, H' may contain a subhypergraph S for which there exists an edge order which starts with all of the edges of S, and all subsequent edges are good: The subhypergraph S if it exists must be

Case 1: A (k, u, 2)-star for some integer $u \ge 2$;

Case 2: A (k, u, a)-link for some $u, a \in \mathbb{N}$ with $u \ge 3$ and $a \le \lfloor k/(u-1) \rfloor$;

Case 3: A Pasch configuration;

Case 4: A loose cycle;

Case 5: An overlapping pair.

Let the edges of S be e_0, \ldots, e_{s-1} , and the rest of the good edges e_s, \ldots, e_t . (If S does not exist, then set s = 0.)

Breaker uses the following strategy.

• If Maker selects a vertex in S, then Breaker does the following, corresponding to the cases above for what S could be.

Case 1: There are 2u edges; suppose without loss of generality that they are labelled so that e_i and e_{i+u} have intersection k-1 for each $i \in [0, u-1]$. Suppose Maker has selected a vertex in $e_j \cap e_{j+u}$ for $j \in [0, u-1]$. Then Breaker if he can, also selects such a vertex. Otherwise he selects an arbitrary vertex.

Case 2: Breaker selects an arbitrary vertex in S if he can. Otherwise he selects an arbitrary vertex.

Case 3: If Maker has two out of three vertices from one of the edges of S, Breaker chooses the final vertex from this edge. Otherwise Breaker chooses an arbitrary vertex in S if he can. If he cannot then he chooses an arbitrary vertex.

Case 4: Assume without loss of generality that the edges are ordered cyclically e_0, \ldots, e_{s-1} . If Maker has selected an element from $e_i \setminus e_j$ where $i \in [s-1]$ and j = i - 1, or i = 0 and j = s - 1, then Breaker if he can, also selects such a vertex. Otherwise he selects an arbitrary vertex.

Case 5: These two edges are e_0 and e_1 . If Maker has selected an element from $e_0 \cap e_1$, then Breaker if he can, also selects such a vertex. Otherwise he selects an arbitrary vertex.

• If Maker selects any other vertex, let i be such that e_i is the edge in which this vertex is new. If Breaker can, he also selects a vertex which is new in e_i . Otherwise he selects an arbitrary vertex.

We must now show that at the end of the game, Maker has failed to claim every vertex of any edge in H'. First observe that Maker has failed to claim any of the edges e_s, \ldots, e_t . Indeed, let $i \in [s, t]$. In the edge order, e_i is good, and so there exists at least $k - 1 \ge 2$ vertices x_i and y_i which are new in e_i (they do not appear in any edge e_j for j < i). By part two of the strategy above, Maker cannot claim all of the vertices of e_i since as soon as she tries to claim one of the at least two new vertices in e_i , Breaker will claim another new vertex in e_i . Thus if Maker has won the game, she must have claimed an edge from S. However, we will now run through each case, corresponding to the cases for S in Breaker's strategy above, showing that Maker has not claimed such an edge.

Case 1: Let $j \in [0, u - 1]$ and suppose Maker is trying to claim e_j or e_{j+u} . There are $k - 1 \ge 2$ vertices in $e_j \cap e_{j+u}$ and so Maker cannot claim all of the vertices of e_j or e_{j+u} since as soon as she tries to claim one of the vertices which lie in $e_j \cap e_{j+u}$. Breaker will claim another vertex in $e_j \cap e_{j+u}$. Since all edges of S are of this form, Maker cannot claim any edge of S.

Case 2: Note that a (k, u, a)-link has at most 2k - 2 vertices. Hence by Breaker always claiming any vertex in S whenever Maker does, he ensures that Maker can claim at most $\lceil (2k-2)/2 \rceil = k-1$ of the vertices in S, therefore does not have enough to claim a full edge of S.

Case 3: Breaker always tries to claims a vertex in S if Maker does, hence Maker claims at most three of the six vertices in S. Note that any pair of vertices in S lie together in at most one edge. Hence by Breaker selecting the third vertex of an edge if Maker has selected the first two, Maker is never able to claim all three vertices of an edge of S.

Case 4: Suppose Maker is trying to claim e_i for some $i \in [0, s - 1]$. Let j = i - 1 if $i \ge 1$, and let j = s - 1 if i = 0. There are $k - 1 \ge 2$ vertices in $e_i \setminus e_j$ and so Maker cannot claim all of the vertices of e_i since as soon as she tries to claim one of the vertices in $e_i \setminus e_j$, Breaker will claim another vertex in $e_i \setminus e_j$. Since e_i was arbitrary, Maker cannot claim any edge of S.

Case 5: There are at least two vertices in $e_0 \cap e_1$ and so Maker cannot claim all of the vertices in e_0 or e_1 since as soon as she tries to claim one of the vertices in $e_0 \cap e_1$, Breaker will claim another vertex in $e_0 \cap e_1$.

(Q3) A probabilistic lemma.

Lemma 6.15. Suppose B is a strictly balanced full rank $\ell \times k$ matrix and suppose ℓ divides k-1. Then there exists a positive constant c such that if $p < cn^{-1/m(A)}$, then with high probability H does not contain a bicycle.

Proof. Let R_{b_t} be a random variable counting the number of bicycles which are in H. By Proposition 1.1, it suffices to show that the expectation of R_{b_t} converges to zero as n tends to infinity.

We let R_{b_1}, \ldots, R_{b_8} respectively count the number of hypergraphs $J := e_{f_1}, \ldots, e_{f_u}$ in H with $u \leq \log n$, for which J corresponds to

- (i) a spoiled cycle;
- (ii) a double overlapping pair;
- (iii) a double loose cycle;
- (iv) an overlapping pair with handle;
- (v) a loose cycle with handle;
- (vi) a loose cycle to loose cycle;
(vii) an overlapping pair to overlapping pair;

(viii) an overlapping pair to loose cycle.

Note that each of these hypergraphs contain a loose path of length u - 2; hence let R_{b_9} count the number of loose paths e_{f_1}, \ldots, e_{f_u} in H with $u \ge (\log n) - 1$. Then we have $R_{b_t} \le \sum_{i=1}^{9} R_{b_i}$ and hence it suffices to show $\mathbb{E}(R_{b_i}) = o(1)$ for each i. The cases for i = 1, 5, 9 were covered by Rödl and Ruciński's proof, however we will repeat them here for clarity.

Suppose that $J := e_{f_1}, \ldots, e_{f_u}$ is the valid edge order corresponding to one of the nine cases listed above given by the definitions earlier. When calculating an upper bound on the expected number of copies of some hypergraph J in H, we need to first bound the number of ways to draw J (i.e. bound the number of non-isomorphic hypergraphs which J could be - e.g. for a spoiled cycle e_{f_1}, \ldots, e_{f_u} we need to choose the size of the intersection $e_{f_1} \cap e_{f_2}$, and also the number of edges u). Second, we should consider J as being drawn, and bound the number of ways to pick elements from $[n]_p$ to represent each vertex of J. Thus we are interested in bounding the number of ways of drawing each Jand also the number of ways of choosing representatives from $[n]_p$ for each vertex of J.

Each hypergraph J which we wish to count can be written as a union of at most three hypergraphs P_1, P_2, P_3 , for which each of these are one of an overlapping pair, loose cycle, or loose path. Further, if P_2 and P_3 exist, we have $|V(P_1) \cap V(P_2)| \leq k - 1$, $|V(P_2) \cap V(P_3)| \leq k - 1$ and $|V(P_1) \cap V(P_3)| = \emptyset$. Thus for each $i \in [2]$ we have at most $(|V(P_i)| \cdot |V(P_{i+1})|)^{k-1}$ choices for how to make P_i and P_{i+1} intersect. There is only one way to draw a loose cycle or loose path of given length, and at most k - 2 ways to draw an overlapping pair. Further for each J in (i)–(viii), we have $|V(J)| \leq k \log n$. Thus the total number of ways of drawing each J in (i)–(viii) is at most polylogarithmic in n.

Recall that B is a strictly balanced matrix of dimension $\ell \times k$, and hence for every

 $W \subseteq [k]$ for which $2 \leq |W| < k$ we have

$$\frac{|W| - 1}{|W| - 1 + \operatorname{rank}(B_{\overline{W}}) - \ell} < \frac{k - 1}{k - 1 - \ell}.$$
(6.3.1)

Additionally we have $m(B) = \frac{k-1}{k-1-\ell}$. Now let $p < cn^{-1/m(B)} = cn^{-(k-\ell-1)/(k-1)}$ where we choose c to be a constant satisfying

$$c < 1/(ke^2).$$
 (6.3.2)

Given $i \in [u]$ and $J = e_{f_1}, \ldots, e_{f_u}$ has been drawn, we wish to bound the expected number of ways of picking e_{f_i} to be an edge with q old elements. Such an edge represents a solution x to Bx = b', where q of the x_i have already been chosen. Let these indices be W; we are now attempting to solve $B_{\overline{W}}x' = b''$ for some vector b'' of k - q elements. Note also we must choose one of the q! possible assignments of the q indices in W to the q old elements. Thus the expected number of ways, Y, of picking the k - q new vertices for e_i , satisfies

$$Y \leqslant \sum_{\substack{W \subseteq [k] \\ |W| = q}} q! n^{k-q-\operatorname{rank}(B_{\overline{W}})} p^{k-q}.$$

We wish to bound this conditional expectation Y. By rearranging the inequality given by (6.3.1), we have (if $|W| \ge 2$)

$$\ell(k - |W|) - (k - 1)\operatorname{rank}(B_{\overline{W}}) < 0.$$

In fact since all quantities above are integers and ℓ divides k-1, we must have

$$\ell(k - |W|) - (k - 1)\operatorname{rank}(B_{\overline{W}}) \leqslant -\ell.$$
(6.3.3)

Thus we have

$$n^{k-|W|-\operatorname{rank}(B_{\overline{W}})}p^{k-|W|} \leqslant \begin{cases} c^k n^{\frac{\ell}{k-1}} & \text{if } |W| = 0; \\ c^{k-1} & \text{if } |W| = 1; \\ cn^{\frac{-\ell}{k-1}} & \text{if } 2 \leqslant |W| \leqslant k-1. \end{cases}$$
(6.3.4)

(Note that here we used Proposition 5.18(i) which states that $\operatorname{rank}(B_{\overline{W}}) = \ell$ if |W| = 1.)

For each hypergraph J in (i)–(viii), there is always precisely one initial edge, u - 3 good edges, and two bad edges. Thus the number of choices we have for picking which element of $[n]_p$ to use for each vertex in J has expectation which is at most:

$$n^{k-\ell}p^k \left(\sum_{\substack{W \subseteq [k] \\ |W|=1}} n^{k-|W|-\operatorname{rank}(B_{\overline{W}})}p^{k-|W|}\right)^{u-3} \left(\sum_{\substack{W \subseteq [k] \\ 2 \leqslant |W| \leqslant k-1}} |W|! n^{k-|W|-\operatorname{rank}(B_{\overline{W}})}p^{k-|W|}\right)^2$$

$$\overset{(6.3.4)}{\leqslant} (k!)^2 \cdot (kc)^u \cdot n^{-\ell/(k-1)}.$$

We conclude that for each $i \in [1, 8]$ we have $\mathbb{E}(R_{b_i}) < O(n^{-\ell/(k-1)} \cdot \operatorname{polylog}(n)) = o(1)$.

Finally, note that a loose path with at most n vertices clearly has at most n edges. Further using Proposition 5.18(iv), we have $k \ge \ell + 2$. Thus we have

$$\mathbb{E}(R_{b_9}) \leqslant O\left(\sum_{\substack{u \ge (\log n) - 1}}^n n^{k-\ell} p^k \prod_{i=1}^{u-1} \left(\sum_{\substack{W \subseteq [k] \\ |W| = 1}}^n n^{k-|W|-\operatorname{rank}(B_{\overline{W}})} p^{k-|W|}\right)\right)$$

$$\stackrel{(6.3.4)}{\leqslant} O\left(n^{\frac{\ell}{k-1}} \sum_{\substack{u \ge (\log n) - 1}}^n (kc)^u\right) \stackrel{(6.3.2)}{=} o(1),$$

as required.

Putting the parts together. To reiterate the main points of the proof of Theo-

rem 6.3(ii), we finish by showing that it follows easily from the lemmas in each of the parts (Q1)–(Q3).

Proof (summary) of Theorem 6.3(ii). Let A be a fixed integer-valued matrix of dimension $\ell' \times k'$ and b a fixed integer-valued vector of dimension ℓ' , such that the system of linear equations \mathcal{L} is irredundant, and A is irredundant and satisfies (*). In order to prove w.h.p. Breaker wins the (A, b)-game on $[n]_p$, by Proposition 6.8, it suffices to show w.h.p. Breaker wins the (B, b')-game on $[n]_p$, where (B, b') is the associated pair of (A, b). We rephrase the problem to a game on the hypergraph $H := H([n]_p, B, b')$. We then show that if a component H' of H does not contain a bicycle, then it satisfies certain conditions stated in Lemma 6.9. Breaker wins the game played on such a component by Lemma 6.10. Supposing that B is an $\ell \times k$ matrix where ℓ divides k - 1, then by Lemma 6.15, w.h.p. H (and therefore each component of H) does not contain a bicycle. Finally since Breaker can win the game on each component of H, he wins the game on H, and thus wins the (A, b)-game on $[n]_p$ w.h.p., as required.

6.4 Concluding remarks and Proof of Theorem 6.2

6.4.1 Improvements on Breaker's strategy

The strange fact that our proof of Theorem 6.3(ii) works when B(A) is an $\ell \times k$ matrix such that ℓ divides k - 1 follows precisely via inequality given by (6.3.3). We found an equivalence between bicycles and valid edge orders with two bad edges in Claim 6.12. Suppose we extended our language to *p*-cycles (corresponding to valid edge orders with p bad edges), and were able to find a winning strategy for Breaker playing the game on any component of H that does not contain a *p*-cycle. We could then obtain a proof for matrices A for which the associated matrix B of dimension $\ell \times k$ satisfies $\ell = p - 1$ (without the need for any divisibility conditions). However given the number of cases that arose from considering bicycles, it would seem unfeasible to attempt this.

Lemma 6.9 gives a precise description of hypergraphs with at most one bad edge and a fixed number of k-bad edges. What can be said of hypergraphs with at most p bad edges (for fixed p) and a fixed number of k-bad edges? Also note that our definition of a valid edge order (where every edge after the first one is either good or bad, i.e. there are zero k-bad edges) is a hypergraph generalisation of a tree. This is since trees have precisely this property in the graph case; a 2-bad edge here is an edge which completes a cycle. Thus it would be interesting to obtain a more detailed description of hypergraphs with a valid edge order.

Observe the following connection of this with Rödl and Ruciński's proof of Theorem 5.8. It is very easy to 2-colour a hypergraph with a valid edge order so that it has no monochromatic edges; simply go through the edges in order, colouring the (at least one) new vertex of an edge e_i the colour which was not assigned to one of the old vertices of e_i . Thus if the hypergraph associated to $[n]_p$ has a valid edge order, then $[n]_p$ can be 2-coloured so that there are no monochromatic solutions to Ax = 0.

6.4.2 Matrices which do not satisfy (*)

Matrices A which are irredundant and do not satisfy (*) traditionally have not received as much attention. For such a matrix, \mathbb{N} is not (\mathcal{L}^d, r) -Rado for any $r \ge 2$ since we can 2-colour \mathbb{N} and avoid any monochromatic solutions to Ax = 0 (see Section 5.3.1). Also note that m(A) is ill-defined in this case. Further in the bias version of the \mathcal{L} -game, recall that Theorem 6.1(ii) states that Breaker wins the (1:2) \mathcal{L} -game on [n]. All of these facts follow easily via use of the row of the matrix which (under Gaussian elimination) has at most two non-zero entries. We show through some examples that the threshold for the random \mathcal{L} -game is at least slightly less trivial.

Theorem 6.16. Let A be a fixed integer-valued matrix of dimension $\ell \times k$ and b a fixed

integer-valued vector. Given the system of linear equations \mathcal{L} is irredundant and A is irredundant and does not satisfy (*), we have the following:

- (i) If additionally $A = \begin{pmatrix} \alpha & -\beta \end{pmatrix}$ where α, β are non-equal positive integers, then Maker wins the \mathcal{L} -game on $[n]_p$ w.h.p. if $p \gg n^{-1/3}$.
- (ii) Breaker wins the \mathcal{L} -game on $[n]_p$ w.h.p. if $p \ll n^{-1/3}$.
- (iii) If additionally $A = \begin{pmatrix} \alpha & -\beta & 0 \\ 0 & \alpha & -\beta \end{pmatrix}$ where α, β are non-equal positive integers, then Breaker wins the (A, 0)-game on [n] (i.e. the non-biased non-random game; thus with probability equal to one, Breaker wins the (A, 0)-game on $[n]_p$ for any 0).

Proof. For the \mathcal{L} -game in (i), since \mathcal{L} is irredundant, we have that there exists a solution to $\alpha x_1 - \beta x_2 = b$ in \mathbb{N} with $x_1 \neq x_2$. Thus we have that $t := \gcd(\alpha, \beta)$ must divide b: hence we may assume without loss of generality that $\gcd(\alpha, \beta) = 1$.

Maker wins the \mathcal{L} -game in (i) if there exists a distinct triple $\{(\alpha x - b)/\beta, x, (\beta x + b)/\alpha\} \subseteq [n]_p$, since she can have the first pick and choose x. Then she can complete a solution by picking whichever of $(\alpha x - b)/\beta$ and $(\beta x + b)/\alpha$ remains unchosen after Breaker's turn.

Claim 6.17. There exists a fixed $z \in [0, \alpha\beta - 1]$ (depending on α, β, b) such that whenever $x \equiv z \mod \alpha\beta$, the triple $\{(\alpha x - b)/\beta, x, (\beta x + b)/\alpha\}$ is contained in the integers.

Proof. Let $x \in \mathbb{Z}$. Note that we have $(\beta x + b)/\alpha \in \mathbb{Z}$ whenever $\beta x \equiv -b \mod \alpha$. Since $gcd(\alpha, \beta) = 1$, there exists $y \in [0, \alpha - 1]$ such that whenever $x \equiv y \mod \alpha$, we have $(\beta x + b)/\alpha \in \mathbb{Z}$. Similarly there exists $y' \in [0, \beta - 1]$ such that whenever $x \equiv y' \mod \beta$, we have $(\alpha x - b)/\beta \in \mathbb{Z}$. Combining these two facts, the Chinese remainder theorem implies that there exists $z \in [0, \alpha\beta - 1]$ such that whenever $x \equiv z \mod \alpha\beta$ we have $(\beta x + b)/\alpha \in \mathbb{Z}$ and $(\alpha x - b)/\beta \in \mathbb{Z}$. Call triples which satisfy the property in Claim 6.17 good. From the claim, for sufficiently large n we deduce that [n] contains $n/(2\alpha^2\beta^2)$ good triples. Further, each $x \in [n]$ is in at most three good triples. Thus, there is a collection X of at least $n/(6\alpha^2\beta^2)$ good triples in [n] that are all pairwise disjoint. The expected number of triples in X in $[n]_p$ is $\Theta(np^3)$. Hence if $p \gg n^{-1/3}$ then by Proposition 1.2(ii) w.h.p. there exists x such that $\{(\alpha x - b)/\beta, x, (\beta x + b)/\alpha\} \subseteq [n]_p$, so Maker wins as required.

If $p \ll n^{-1/3}$, then the expected number of triples is o(1), so via Proposition 1.1 w.h.p. there are no triples of this form at all. For the game in (ii), since A is irredundant but does not satisfy (*), under Gaussian elimination there exists one row of A which consists of $\alpha, -\beta$ (where α and β are non-equal positive integers) and zeroes, and thus any solution to Ax = b contains the positive integers $(\beta z + c)/\alpha$ and $(\beta((\beta z + c)/\alpha) + c)/\alpha$ for some rational number z and fixed integer c = c(A, b). Since w.h.p. there are no triples (replacing b with c and x with z), whenever we have the pair $(\beta z + c)/\alpha, (\beta((\beta z + c)/\alpha) + c)/\alpha \in [n]_p$, then $z, (\beta((\beta((\beta z + c)/\alpha) + c)/\alpha) + c)/\alpha \notin [n]_p$. Thus Breaker can devise a pairing strategy to win the game.

For (iii), as in (i) we may assume without loss of generality that $gcd(\alpha, \beta) = 1$. Then every solution in [n] is a triple of the form $\{\alpha^2 x, \alpha\beta x, \beta^2 x\}$ for some $x \in \mathbb{N}$. Every element of [n] which could be in a solution is of the form $\alpha^i \beta^j y$ with $y \in \mathbb{N}$, where α and β do not divide y, and i, j are non-negative integers where at most one of i or j is zero. Using these facts, Breaker has a strategy to win the game. Indeed, Breaker can create a pairing strategy as follows: pair $\alpha^i \beta^j y$ with $\alpha^{i+1} \beta^{j-1} y$ whenever i is even and $j \ge 1$. Then observe that for any triple $\{\alpha^2 x, \alpha\beta x, \beta^2 x\}$ with $x \in \mathbb{N}$, the middle element is paired with one of the two end elements, so Maker cannot obtain a triple. \Box

6.4.3 Proof of Theorem 6.2

We conclude the chapter by finishing the proof of Theorem 6.2.

Theorem 6.2. Let A be a fixed non-zero-integer-valued matrix of dimension $1 \times k$ and b a fixed integer (i.e. Ax = b corresponds to a single linear equation $a_1x_1 + \cdots + a_kx_k = b$ with the a_i non-zero integers).

- (i) If the system of linear equations *L* is irredundant and A is irredundant and satisfies
 (*), then the *L*-game on [n]_p has a threshold probability of Θ(n^{-k-2}/_{k-1});
- (ii) If the system of linear equations L is irredundant and A is irredundant and does not satisfy (*), then the L-game on [n]_p is Maker's win if p ≫ n^{-1/3} and Breaker's win if p ≪ n^{-1/3};
- (iii) If the system of linear equations \mathcal{L} is irredundant and A is not irredundant, then
 - (a) the \mathcal{L} -game on $[n]_p$ is Breaker's win w.h.p. for any p = o(1) if the coefficients a_i are all positive or all negative;
 - (b) the \mathcal{L} -game on $[n]_p$ is Maker's win if $p \gg n^{-1/3}$ and Breaker's win if $p \ll n^{-1/3}$ otherwise;
- (iv) If the system of linear equations L is not irredundant, then the L-game on [n] is
 (trivially) Breaker's win.

Proof.

- (i) As discussed in the introduction, this follows immediately from Theorem 6.3.
- (ii) We have $k \ge 3$ if and only if A satisfies (*). Hence if A does not satisfy (*) and is a linear equation, we must have k = 2. So write $A = \begin{pmatrix} \alpha & \beta \end{pmatrix}$, where $\alpha, \beta \in \mathbb{Z}$. Note that since A is irredundant there exist $x_1, x_2 \in \mathbb{N}$ such that $\alpha x_1 + \beta x_2 = 0$ where $x_1 \ne x_2$. Thus we must have $\alpha > 0$ and $\beta < 0$ or vice versa, and $\alpha \ne -\beta$. Thus the result follows from Theorem 6.16(i) and (ii).

- (iii) Note that any linear equation $a_1x_1 + \ldots a_kx_k = 0$ with $k \ge 3$ clearly has a k-distinct solution in N if there exists at least one positive a_i and at least one negative a_j , for some $i, j \in [k]$. The same holds for k = 2 unless if $a_1 = -a_2$. Thus, since A is not irredundant, we have one of the following:
 - (a) the a_i are all positive integers or all negative integers;
 - (b) we have k = 2 and $a_1 = -a_2$.

For (a), we may assume without loss of generality that a_1, \ldots, a_k and therefore b are positive integers. For such a game there are a finite number of k-distinct solutions in \mathbb{N} , all of which are contained in [b]. Thus for any p = o(1), w.h.p. there are no solutions in $[n]_p$ by Proposition 1.1, so the game is Breaker's win. For (b), the existence of any triple $\{x - b/a_1, x, x + b/a_1\}$ leads to a win for Maker, meanwhile if no triples exist then Breaker can win by a pairing strategy. Since the number of such triples in $[n]_p$ is of order np^3 , the result follows by a similar argument to that given for Theorems 6.16(i) and (ii).

(iv) The \mathcal{L} -game is trivially Breaker's win, since there are no winning sets in \mathbb{N} .

LIST OF REFERENCES

- H.L. Abbott and E.T.H. Wang, Sum-free sets of integers, Proc. Amer. Math. Soc., 67, (1977), 11–16.
- [2] N. Alon, Independent sets in regular graphs and sum-free subsets of finite groups, *Israel J. Math.*, 73, (1991), 247–256.
- [3] N. Alon, R.A. Duke, H. Lefmann, V. Rödl and R. Yuster, The algorithmic aspects of the regularity lemma, J. Algorithms, 16, (1994), 80–109.
- [4] J. Balogh, B. Csaba and W. Samotij, Local resilience of almost spanning trees in random graphs, *Random Structures & Algorithms*, 38, (2011), 121–139.
- [5] J. Balogh, H. Liu, S. Petříčková and M. Sharifzadeh, The typical structure of maximal triangle-free graphs, *Forum Math. Sigma*, 3, (2015), E20.
- [6] J. Balogh, H. Liu, M. Sharifzadeh and A. Treglown, The number of maximal sum-free subsets of integers, Proc. Amer. Math. Soc., 143, (2015), 4713–4721.
- [7] J. Balogh, H. Liu, M. Sharifzadeh and A. Treglown, Sharp bound on the number of maximal sum-free subsets of integers, J. Eur. Math. Soc., 20, (2018), 1885–1911.
- [8] J. Balogh, R. Morris and W. Samotij, Independent sets in hypergraphs, J. Amer. Math. Soc., 28, (2015), 669–709.
- [9] J. Balogh, R. Morris and W. Samotij, The method of hypergraph containers, submitted.
- [10] J. Balogh and S. Petříčková, The number of the maximal triangle-free graphs, Bull. London Math. Soc., 46, (2014), 1003–1006.
- [11] A. Baltz, P. Hegarty, J. Knape, U. Larsson and T. Schoen, The structure of maximum subsets of $\{1, \ldots, n\}$ with no solutions to a+b=kc, *Electron. J. Combin.*, 12, (2005), R19.
- [12] J. Beck, Combinatorial Games: Tic-Tac-Toe Theory. *Cambridge University Press* (2008).

- [13] J. Beck, Van der Waerden and Ramsey type games, Combinatorica, 1, (1981), 103– 116.
- [14] M. Bednarska and T. Łuczak, Biased positional games for which random strategies are nearly optimal, *Combinatorica*, 20, (2000), 477–488.
- [15] F. Behrend, On sets of integers which contain no three terms in arithmetic progression, Proc. Nat. Acad. Sci., 32, (1946), 331–332.
- [16] A. Bialostocki, Y. Caro, and Y. Roditty, On zero sum Turan numbers, Ars Combin., 29, (1990), 117–127.
- [17] Y. Bilu, Sum-free sets and related sets, Combinatorica, 18, (1998), 449–459.
- [18] T.F. Bloom, A quantitative improvement for Roth's theorem on arithmetic progressions, J. London Math. Soc., 93, (2016), 643–663.
- [19] B. Bollobás and A. Thomason, Threshold functions, *Combinatorica*, 7, (1987), 35–38.
- [20] J. Böttcher, Y. Kohayakawa and A. Taraz, Almost spanning subgraphs of random graphs after adversarial edge removal, *Combin. Probab. Comput.*, 22, (2013), 639– 683.
- [21] S. A. Burr, P. Erdős, L. Lovász, On graphs of Ramsey type, Ars Combinatoria, 1, (1976), 167–190.
- [22] N.J. Calkin and J.M. Thomason, Counting generalized sum-free sets, J. Number Theory, 68, (1996), 151–159.
- [23] P. Cameron and P. Erdős, On the number of sets of integers with various properties, in Number Theory (R.A. Mollin, ed.), 61–79, Walter de Gruyter, Berlin, 1990.
- [24] P. Cameron and P. Erdős, Notes on sum-free and related sets, Combin. Probab. Comput., 8, (1999), 95–107.
- [25] A. Cauchy, Recherches sur les nombres, Jour. Ecole polytechn., 9, (1813), 99–116.
- [26] V. Chvátal, personal communication to the authors of [21].
- [27] D. Conlon, Combinatorial theorems relative to a random set, Proceedings of the International Congress of Mathematicians 2014, Vol. 4, 303–328.
- [28] D. Conlon and W.T. Gowers, Combinatorial theorems in sparse random sets, Ann. Math., 84, (2016), 367–454.
- [29] D. de Caen, Extension of a theorem of Moon and Moser on complete subgraphs, Ars Combin., 16, (1983), 5–10.

- [30] J. Deshouillers, G. Freiman, V. Sós and M. Temkin, On the structure of sum-free sets II, Astérisque, 258, (1999), 149–161.
- [31] K. Dilcher and L. Lucht, On finite pattern-free sets of integers, Acta Arith., 121, (2006), 313–325.
- [32] M. Elkin, An improved construction of progression-free sets, Israel J. Math., 184, (2011), 93–128.
- [33] P. Erdős, A problem on independent r-tuples, Ann. Univ. Sci. Budapest. Eötvös Sect. Math., 8, (1965), 93–95.
- [34] P. Erdős, P. Frankl and V. Rödl, The asymptotic number of graphs not containing a fixed subgraph and a problem for hypergraphs having no exponent, *Graphs Combin.*, 2, (1986), 113–121.
- [35] P. Erdős and J. L. Selfridge, On a combinatorial game, J. Comb. Theory Ser. A, 14, (1973), 298–301.
- [36] P. Erdős and M. Simonovits, Supersaturated graphs and hypergraphs, Combinatorica, 3, (1983), 181–192.
- [37] P. Erdős and A. H. Stone, On the structure of linear graphs, Bulletin American Math. Soc., 52, (1946), 1087–1091.
- [38] P. Erdős and P. Turán, On a problem of Sidon in additive number theory, and on some related problems, J. London Math. Soc., 16, (1941), 212–215.
- [39] P. Erdős and P. Turán, On some sequences of integers, J. London Math. Soc., 11, (1936), 261–264.
- [40] V. Falgas-Ravry, K. O'Connell and A. Uzzell, Multicolour containers, extremal entropy and counting, *Random Structures & Algorithms*, (2018), doi:10.1002/rsa.20777.
- [41] P. Frankl, R.L. Graham and V. Rödl, Quantitative theorems for regular systems of equations, J. Combin. Theory A, 47, (1988), 246–261.
- [42] E. Friedgut, V. Rödl and M. Schacht, Ramsey properties of random discrete structures, Random Structures & Algorithms, 37, (2010), 407–436.
- [43] Z. Füredi, Extremal hypergraphs and combinatorial geometry, Proceedings of the International Congress of Mathematicians, Zurich, Switzerland 1994. 1343–1352, Birkhäuser, Basel, 1995.
- [44] W.T. Gowers, Quasirandomness, counting and regularity for 3-uniform hypergraphs, Combin. Probab. Comput., 15, (2006), 143–184.

- [45] W.T. Gowers, Hypergraph regularity and the multidimensional Szemerédi theorem, Ann. of Math., 166, (2007), 897–946.
- [46] R. Graham, V. Rödl, and A. Ruciński, On Schur properties of random subsets of integers, J. Number Theory, 61, (1996), 388–408.
- [47] B. Green, The Cameron-Erdős conjecture, Bull. London Math. Soc., 36, (2004), 769– 778.
- [48] B. Green, A Szemerédi-type regularity lemma in abelian groups, with applications, Geom. Funct. Anal., 15, (2005), 340–376.
- [49] B. Green and T. Tao, The primes contain arbitrarily long arithmetic progressions, Annals of Math., 167, (2008), 481–547.
- [50] B. Green and J. Wolf, A note on Elkin's improvement of Behrend's construction, Additive number theory: Festschrift in honor of the sixtieth birthday of Melvyn B. Nathanson, pages 141–144. Springer-Verlag, 1st edition, 2010.
- [51] L. Gugelmann, R. Nenadov, Y. Person, A. Steger, N. Skorić and H. Thomas, Symmetric and asymmetric Ramsey properties in random hypergraphs, *Forum Math. Sigma*, 5, (2017), E28.
- [52] D. Hefetz, M. Krivelevich, M. Stojaković and T. Szabó, Positional Games, Oberwolfach Seminars, vol. 44, Birkhäuser (2014).
- [53] P. Hegarty, Extremal subsets of {1,...,n} avoiding solutions to linear equations in three variables, *Electron. J. Combin.*, 14, (2007), R74.
- [54] R. Hancock, The Maker-Breaker Rado game on a random set of integers, to appear in *SIAM J. Discrete Math.*
- [55] R. Hancock, K. Staden and A. Treglown, Independent sets in hypergraphs and Ramsey properties of graphs and the integers, submitted.
- [56] R. Hancock and A. Treglown, On solution-free sets of integers, European J. Combin., 66, (2017), 110–128.
- [57] R. Hancock and A. Treglown, On solution-free sets of integers II, Acta Arith., 180, (2017), 15–33.
- [58] P.E. Haxell, Y. Kohayakawa and T. Łuczak, Turán's extremal problem in random graphs: forbidding even cycles, J. Combin. Theory B, 64, (1995), 273–287.
- [59] P.E. Haxell, Y. Kohayakawa and T. Łuczak, Turán's extremal problem in random graphs: forbidding odd cycles, *Combinatorica*, 16, (1996), 107–122.

- [60] N. Hindman and I. Leader, Nonconstant monochromatic solutions to systems of linear equations, *Topics in Discrete Math. Springer, Berlin* (2006), 145–154.
- [61] M.C. Hu, A note on sum-free sets of integers, Proc. Amer. Math. Soc., 80, (1980), 711–712.
- [62] M. Hujter and Z. Tuza, The number of maximal independent sets in triangle-free graphs, SIAM J. Discrete Math., 6, (1993), 284–288.
- [63] Y. Ishigami, The number of hypergraphs and colored hypergraphs with hereditary properties, preprint.
- [64] S. Janson, T. Łuczak and A. Ruciński, *Random Graphs*, Wiley, 2000.
- [65] S. Janson and A. Ruciński, Upper tails for counting objects in randomly induced subhypergraphs and rooted random graphs, Ark. Mat., 49, (2011), 79–96.
- [66] G. Katona, T. Nemetz and M. Simonovits, On a problem of Turán in the theory of graphs, *Mat. Lapok*, 15, (1964), 228–238.
- [67] P. Keevash, Hypergraph Turán problems, Surveys in Combinatorics, Cambridge University Press, 2011, 83–140.
- [68] D.J. Kleitman and K.J. Winston, The asymptotic number of lattices, Ann. Discrete Math., 6, (1980), 243–249.
- [69] D.J. Kleitman and K.J. Winston, On the number of graphs without 4-cycles, *Discrete Math.*, 41, (1982), 167–172.
- [70] Y. Kohayakawa and B. Kreuter, Threshold functions for asymmetric Ramsey properties involving cycles, *Random Structures & Algorithms*, 11, (1997), 245–276.
- [71] Y. Kohayakawa, S. Lee, V. Rödl, and W. Samotij, The number of Sidon sets and the maximum size of Sidon sets contained in a sparse random set of integers, *Random Structures & Algorithms*, 46, (2015), 1–25.
- [72] Y. Kohayakawa, T. Łuczak and V. Rödl, On K_4 -free subgraphs of random graphs, Combinatorica, 17, (1997), 173–213.
- [73] P. Kolaitis, H.J. Prömel and B. Rothschild, K_{l+1} -free graphs: asymptotic structure and a 0 1 law, Trans. Amer. Math. Soc., 303, (1987), 637–671.
- [74] D. Král', O. Serra and L. Vena, A removal lemma for systems of linear equations over finite fields, *Israel J. Math.*, 187, (2012), 193–207
- [75] C. Kusch, J. Rué, C. Spiegel and T. Szabó, On the optimality of the uniform random strategy, submitted.

- [76] B.M. Landman and A. Robertson, Ramsey theory on the integers, Student Mathematical Library 24. Providence, RI: American Mathematical Society (AMS), 2004.
- [77] C. Lee and B. Sudakov, Dirac's theorem for random graphs, *Random Structures & Algorithms*, 41, (2012), 293–305.
- [78] S. Lin, On Ramsey numbers and K_r-coloring of graphs, J. Combin. Theory, 12, (1972), 82–92.
- [79] T. Luczak and T. Schoen, On the number of maximal sum-free sets, Proc. Amer. Math. Soc., 129, (2001), 2205–2207.
- [80] W. Mantel, Problem 28, solution by H. Gouwentak, W. Mantel, J. Teixeira de Mattes, F. Schuh and W.A. Wythoff, *Wiskundige Opgaven*, 10, (1907), 60–61.
- [81] M. Marciniszyn, J. Skokan, R. Spöhel and A. Steger, Asymmetric Ramsey properties of random graphs involving cliques, *Random Structures & Algorithms*, 34, (2009), 419–453.
- [82] J.W. Moon and L. Moser, On cliques in graphs, *Israel J. Math.*, 3, (1965), 23–28.
- [83] F. Mousset, R. Nenadov and W. Samotij, Towards the Kohayakawa-Kreuter conjecture on asymmetric Ramsey properties, submitted.
- [84] B. Nagle, V. Rödl and M. Schacht, The counting lemma for regular k-uniform hypergraphs, *Random Structures Algorithms*, 28, (2006), 113–179.
- [85] R. Nenadov and A. Steger, A Short Proof of the Random Ramsey Theorem, Combin. Probab. Comput., 25, (2016), 130–144.
- [86] R. Nenadov, A. Steger and M. Stojaković, On the threshold for the Maker-Breaker H-game, Random Structures & Algorithms, 49, (2016), 558–578.
- [87] H. Prömel and A. Steger, The asymptotic number of graphs not containing a fixed color-critical subgraph, *Combinatorica*, 12, (1992), 463–473.
- [88] R. Rado, Studien zur Kombinatorik, Math. Z., 36, (1933), 425–480.
- [89] F. P. Ramsey, On a problem in formal logic, Proc. London Math. Soc., 30, (1930), 264–286.
- [90] V. Rödl and A. Ruciński, Lower bounds on probability thresholds for Ramsey properties, in Combinatorics, Paul Erdős is Eighty, Vol. 1, 317–346, Bolyai Soc. Math. Studies, János Bolyai Math. Soc., Budapest, 1993.
- [91] V. Rödl and A. Ruciński, Random graphs with monochromatic triangles in every edge coloring, *Random Structures & Algorithms*, 5, (1994), 253–270.

- [92] V. Rödl and A. Ruciński, Threshold functions for Ramsey properties, J. Amer. Math. Soc., 8, (1995), 917–942.
- [93] V. Rödl and A. Ruciński, Rado partition theorem for random subsets of integers, Proc. London Math. Soc., 74, (1997), 481–502.
- [94] V. Rödl and A. Ruciński, Ramsey properties of random hypergraphs, J. Combin. Theory Ser. A, 81, (1998), 1–33.
- [95] V. Rödl, A. Ruciński, and M. Schacht, Ramsey properties of random k-partite, kuniform hypergraphs, SIAM J. Discrete Math., 21, (2007) 442–460.
- [96] V. Rödl and M. Schacht, Extremal results in random graphs, in Erdős centennial, Vol. 25, 535–583, Bolyai Soc. Math. Studies, János Bolyai Math. Soc., Budapest, 2013.
- [97] V. Rödl and J. Skokan, Regularity lemma for k-uniform hypergraphs, Random Structures & Algorithms, 25, (2004), 1–42.
- [98] K.F. Roth, On certain sets of integers, J. London Math. Soc., 28, (1953), 104–109.
- [99] J. Rué, C. Spiegel, and A. Zumalacárregui, Threshold functions and poisson convergence for systems of equations in random sets, A. Math. Z., 288, (2018), 333–360.
- [100] I.Z. Ruzsa, Solving a linear equation in a set of integers I, Acta Arith., 65, (1993), 259–282.
- [101] I.Z. Ruzsa, Solving a linear equation in a set of integers II, Acta Arith., 72, (1995), 385–397.
- [102] I.Z. Ruzsa and E. Szemerédi, Triple systems with no six points carrying three triangles, Combinatorics (Keszthely, 1976), Coll. Math. Soc. J. Bolyai 18, Volume II, 939–945.
- [103] A.A. Sapozhenko, On the number of independent sets in extenders, *Diskret. Mat.*, 13, (2001), 56–62.
- [104] A.A. Sapozhenko, The Cameron-Erdős conjecture, (Russian) Dokl. Akad. Nauk., 393, (2003), 749–752.
- [105] A.A. Sapozhenko, Systems of containers and enumeration problems, Stochastic Algorithms: Foundations and Applica- tions: Third International Symposium, SAGA 2005 (Moscow, Russia), Lecture Notes in Computer Science, vol. 3777, 2005, pp. 1–13.
- [106] D. Saxton and A. Thomason, Hypergraph containers, *Invent. Math.*, 201, (2015), 925–992.

- [107] D. Saxton and A. Thomason, Online containers for hypergraphs, with applications to linear equations, J. Comb. Theory Ser. B, 121, (2016), 248–283.
- [108] M. Schacht, Extremal results for random discrete structures, Ann. Math., 184, (2016), 331–363.
- [109] T. Schoen, The number of (2,3)-sum-free subsets of $\{1,\ldots,n\}$, Acta Arith., 98, (2001), 155–163.
- [110] I. Schur, Über die Kongruenz $x^m + y^m \equiv z^m \pmod{p}$, Jber. Deutsch. Math. Verein., 25, (1916), 114–117.
- [111] M. Simonovits, A method for solving extremal problems in graph theory, stability problems, in 1968 Theory of Graphs (Proc. Colloq., Tihany, 1966) 279–319, Academic Press, New York.
- [112] C. Spiegel, A note on sparse supersaturation and extremal results for linear homogeneous systems, *Electron. J. Combin.*, 24, (2017), #P3.38.
- [113] M. Stojaković and T. Szabó, Positional games on random graphs, Random Structures & Algorithms, 26, (2005), 204–223.
- [114] B. Sudakov, Robustness of graph properties, Surveys in Combinatorics, Cambridge University Press, 2017, 372–408.
- [115] B. Sudakov and V.H. Vu, Local resilience of graphs, Random Structures & Algorithms, 33, (2008), 409–433.
- [116] J.J. Sylvester, Mathematical questions with their solutions, *Educational Times*, 41, (1884), 21.
- [117] E. Szemerédi, On sets of integers containing no k elements in arithmetic progression, Acta Arith., 27, (1975), 299–345.
- [118] C. Terry, Structure and enumeration theorems for hereditary properties in finite relational languages, Ann. Pure Appl. Logic, 169, (2018), 413–449.
- [119] T. Tran, On the structure of large sum-free sets of integers, Israel J. Math., 228, (2018), 249–292.
- [120] B.L. van der Waerden, Beweis einer Baudetschen Vermutung, Nieuw Arch. Wiskunde, 15, (1927), 212–216.
- [121] G. Wolfovitz, Bounds on the number of maximal sum-free sets, European J. Combin., 30, (2009), 1718–1723.